# UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D. C. 20549 FORM 8-K

### CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (date of earliest event reported): April 10, 2014

H&R BLOCK, INC. (Exact name of registrant as specified in charter)

MISSOURI

1-6089

(Commission File Number)

44-0607856 (I.R.S. Employer Identification No.)

(State or other jurisdiction of incorporation or organization)

One H&R Block Way, Kansas City, MO 64105

(Address of Principal Executive Offices) (Zip Code)

(816) 854-3000

(Registrant's telephone number, including area code)

**Not Applicable** 

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

#### Item 1.01 Entry into a Material Definitive Agreement.

H&R Block, Inc. (the "Company") has made significant progress on its previously announced plan to cease being a savings and loan holding company ("SLHC") while continuing to offer financial products to its clients.

#### Sale to Bofl Federal Bank

On April 10, 2014, the Company's subsidiaries, H&R Block Bank, a federal savings bank ("HRB Bank"), and Block Financial LLC, a Delaware limited liability company and the sole shareholder of HRB Bank ("Block Financial"), entered into a definitive Purchase and Assumption Agreement (the "P&A Agreement") with Bofl Federal Bank, a federal savings bank ("Bofl"). Pursuant to the P&A Agreement, HRB Bank will sell assets and assign liabilities, including all of HRB Bank's deposit liabilities, to Bofl (the "P&A Transaction"), subject to various closing conditions, including the receipt of certain required approvals, entry into certain additional agreements, and the fulfillment of various other customary conditions.

In connection with the additional agreements being entered into upon the closing of the P&A Transaction, Bofl will offer H&R Block-branded financial products distributed by the Company to the Company's clients. Emerald Financial Services, LLC ("EFS"), an indirect subsidiary of the Company, will provide servicing and administrative support for such financial products. See discussion below under "Continued Offering of Financial Products" and "Announced Transactions are Subject to Numerous Conditions and the Closing May be Delayed or May Not Occur at All."

#### **Divestiture Transaction**

The P&A Transaction is part of a three-step transaction pursuant to which the Company plans to divest HRB Bank (the "Divestiture Transaction"), including: (i) the conversion of HRB Bank from a federal savings bank to a national bank; (ii) the sale of certain HRB Bank assets and liabilities, including all deposit liabilities, to BofI in the P&A Transaction; and (iii) the merger of HRB Bank with and into Block Financial.

The Company, H&R Block Group, Inc. and Block Financial (collectively, "our Holding Companies") are SLHCs because they control HRB Bank. As previously announced, we have evaluated alternative means of ceasing to be an SLHC. By consummating the Divestiture Transaction, our Holding Companies will cease to be SLHCs and will no longer be subject to regulation by the Board of Governors of the Federal Reserve System (the "Federal Reserve") as an SLHC or to the regulatory capital requirements applicable to SLHCs.

#### Step One: The Conversion

HRB Bank is in the process of finalizing and filing an application with the Office of the Comptroller of the Currency (the "OCC") for approval to convert to a national banking association (the "Conversion"). In connection with the Conversion, our Holding Companies will request from the Federal Reserve a waiver of the Bank Holding Company Application requirements under Sections 3(a)(1) and 4 of the Bank Holding Company Act of 1956 and file an application with the Federal Reserve to deregister as an SLHC.

#### Step Two: The P&A Transaction

Bofl is in the process of finalizing and filing a Bank Merger Act application with the OCC, and HRB Bank is in the process of filing a Change in Assets application with the OCC, seeking approval of the P&A Transaction. Upon receipt of regulatory approvals, the parties will consummate the P&A Transaction immediately following the Conversion. HRB Bank will sell assets and assign liabilities, including all of HRB Bank's deposit liabilities, to Bofl.

#### Step Three: The Affiliate Merger

Block Financial and HRB Bank will file an application with the OCC to merge HRB Bank with and into Block Financial, with Block Financial as the surviving entity (the "Affiliate Merger"), pursuant to the terms and conditions of an Agreement and Plan of Merger by and between HRB Bank and Block Financial. HRB Bank is converting to a national bank because there is clear regulatory authority for a national bank to merge with and into a non-bank affiliate, whereas the regulatory authority for a federal savings bank to do so is unclear. Upon completion of the Affiliate Merger, HRB Bank will cease to exist as a separate legal entity, will surrender its charter to the OCC, and the FDIC will terminate the deposit insurance of HRB Bank. The Affiliate Merger will occur immediately following the closing of the P&A Transaction.

#### **Closing of the Divestiture Transaction**

Subject to fulfillment of all closing conditions, the closing of steps one, two and three of the Divestiture Transaction will all happen on the same day, one immediately after the other. The parties are in the process of finalizing and filing regulatory applications for approval of steps one, two and three of the Divestiture Transaction.

Subject to fulfillment of all closing conditions, including receipt of all required regulatory approvals, we continue to expect that the closing will occur in time for next tax season. See discussion below under "Announced Transactions are Subject to Numerous Conditions and the Closing May be Delayed or May Not Occur at All."

#### Formation of Emerald Financial Services and HRB Mortgage Holdings

As previously disclosed in our Form 8-K filed on July 11, 2013, on July 1, 2013, HRB Bank formed two Delaware limited liability companies as wholly-owned operating subsidiaries. EFS was formed to market and distribute HRB Bank products and services, as well as to serve as the program manager for such products and services. After the closing of the Divestiture Transaction, EFS will market and distribute Bofl's H&R Block-branded financial products and services. HRB Mortgage Holdings, LLC ("Mortgage Holdings") was formed to hold and collect HRB Bank's existing portfolio of residential mortgage loans and other real estate owned ("OREO") properties. HRB Bank has assigned all of its residential mortgage loans and OREO properties to Mortgage Holdings. Mortgage Holdings will receive payments on such mortgage loans and liquidate such OREO properties. HRB Bank will retain the marketable investment securities in its investment portfolio.

Upon completion of the Affiliate Merger, Block Financial will succeed, by operation of law, to all of the assets and liabilities of HRB Bank that are not sold or assigned to Bofl. At that time, EFS and Mortgage Holdings will become direct, wholly-owned subsidiaries of Block Financial.

#### **Continued Offering of Financial Products**

Although we are divesting the assets and surrendering the charter of HRB Bank, we plan to continue to enhance our business by delivering financial products and services to our clients. EFS will be the H&R Block entity that markets and services Bofl's H&R Block-branded financial products. EFS will provide certain servicing and administrative support to Bofl with respect to such financial products.

In connection with the execution of the P&A Agreement, EFS and Block Financial plan to enter into certain other agreements with Bofl concurrent with the closing of the Divestiture Transaction, including without limitation a Program Management Agreement, including related product schedules (the "PMA Agreement"), and a related Emerald Advance Receivables Participation Agreement (the "RPA Agreement").

#### The PMA Agreement

The PMA Agreement will be entered into by EFS and BofI and will be an umbrella agreement for EFS's offering and servicing of BofI's financial products through H&R Block's distribution channels. The PMA Agreement will become effective as soon as the P&A Transaction is completed and will expire June 30, 2021. EFS has the right to terminate the PMA Agreement under certain circumstances, including a one-time right to terminate the PMA Agreement on June 30, 2017. Upon termination of the PMA Agreement or any of the related product schedules, EFS has the right to purchase or arrange the purchase of all of the affected accounts related to its ongoing product offerings.

The PMA Agreement covers the Company's distribution channels in the United States, Puerto Rico, Guam

and certain identified U.S. military bases outside the United States. Under the PMA Agreement and its related product schedules, BofI will act as the bank for H&R Block-branded financial services products: Emerald Prepaid MasterCard®, Refund Transfers and Emerald Advance® lines of credit through our retail and digital channels. In general, EFS will provide certain servicing and administrative support to BofI. As fully set forth in the PMA, EFS is obligated to provide broad indemnification to BofI, subject to limited exceptions. BofI's H&R Block-branded financial products will be available to H&R Block franchisees in accordance with distribution agreements executed by each franchisee, EFS and BofI.

EFS's performance of its obligations under the PMA Agreement is guaranteed by the Company pursuant to a separate Guaranty Agreement between the Company and Bofl.

#### The RPA Agreement

The RPA Agreement will be entered into by BofI, the Company, EFS, and HRB Participant I, LLC ("Participant"), an indirect subsidiary of the Company. The Company is a party to the RPA Agreement solely for the purpose of guarantying Participant's obligations under the RPA Agreement. Participant is a newly-formed, indirect subsidiary of the Company that will hold participation interests generated under the RPA Agreement. The RPA Agreement will become effective as soon as the P&A Transaction is completed and will expire June 30, 2021. Pursuant to the RPA Agreement, Participant will periodically purchase ninety percent participation interests in the receivables of BofI from the Emerald Advance product offered through the Company's retail channels. EFS will administer and service the accounts and participated receivables on behalf of BofI and Participant, according to the PMA Agreement.

#### Announced Transactions are Subject to Numerous Conditions and the Closing May be Delayed or May Not Occur at All.

The obligations of the parties to complete the P&A Transaction and the other transactions discussed herein are subject to the fulfillment of numerous conditions, including, but not limited to: (i) receipt of all regulatory approvals necessary to consummate the entire Divestiture Transaction; (ii) receipt of required third party consents; and (iii) the execution and delivery of the PMA Agreement and the RPA Agreement with Bofl. We cannot be certain when or if the conditions to and other components of the P&A Transaction will be satisfied, or whether the P&A Transaction will be completed. In addition, there may be changes to the terms and conditions of the P&A Agreement, PMA Agreement, RPA Agreement and other contemplated agreements as part of the regulatory approval process.

#### Exhibits

Copies of the P&A Agreement, the PMA Agreement, the RPA Agreement and the Guaranty Agreement are attached hereto as Exhibits 10.1, 10.2, 10.3 and 10.4, respectively. The summary description of the P&A Agreement, PMA Agreement, RPA Agreement and Guaranty Agreement does not purport to be complete and is qualified in its entirety by reference to the P&A Agreement, PMA Agreement, RPA Agreement, RP

On April 10, 2014, the Company issued a press release announcing the P&A Agreement. The full text of the press release is attached hereto as Exhibit 99.1.

Exhibit 99.1 is being furnished but shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934 (the "Exchange Act") or otherwise subject to the liabilities of that section, nor shall it be deemed incorporated by reference in any filing under the Securities Act, or the Exchange Act, except as expressly set forth in such filing.

### Item 9.01. Financial Statements and Exhibits.

## (d) Exhibits

Exhibit Number Description

- 10.1 Purchase and Assumption Agreement
- 10.2 Program Management Agreement
- 10.3 Receivables Participation Agreement
- 10.4 Guaranty Agreement
- 99.1 Press Release

# SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

# H&R BLOCK, INC.

Date: April 10, 2014

By: /s/ Scott W. Andreasen Scott W. Andreasen Vice President and Secretary

# EXHIBIT INDEX

Exhibit 10.1	Purchase and Assumption Agreement
Exhibit 10.2	Program Management Agreement
Exhibit 10.3	Receivables Participation Agreement
Exhibit 10.4	Guaranty Agreement
Exhibit 99.1	Press Release

# PURCHASE AND ASSUMPTION AGREEMENT

among

BofI Federal Bank, a federal savings bank,

and

H&R Block Bank, a federal savings bank,

and

Block Financial LLC, a Delaware limited liability company,

Dated as of April 10, 2014.

## **TABLE OF CONTENTS**

# PURCHASE AND ASSUMPTION AGREEMENT

## **ARTICLE I DEFINITIONS** 1

1.01. Certain Definitions 1

#### ARTICLE II PURCHASE OF ASSETS AND ASSUMPTION OF LIABILITIES 9

- 2.01. Purchase of Assets 9
- 2.02. Excluded Assets 10
- 2.03. Assumed Liabilities 11
- 2.04. Excluded Liabilities 11
- 2.05. Nonassignable Contracts and Rights 12

#### ARTICLE III CLOSING AND TRANSFER AMOUNT 12

- 3.01. Closing 12
- 3.02. Transfer Amount 13
- 3.03. Post-Closing Transfer Calculation 14
- 3.04. Final Settlement 14
- 3.05. Purchase Price Allocation 14
- 3.06. Prorations 15
- 3.07. Custodial Accounts 15

### ARTICLE IV REPRESENTATIONS AND WARRANTIES OF HRB BANK AND BLOCK FINANCIAL 15

- 4.01. Organization and Authority 15
- 4.02. Execution and Delivery 16
- 4.03. Capitalization of HRB Bank 17
- 4.04. Compliance with Laws, Permits and Instruments 17
- 4.05. Financial Statements 17
- 4.06. Undisclosed Liabilities 18
- 4.07. Litigation 18
- 4.08. Consents and Approvals 18
- 4.09. Ownership of Assets 18
- 4.10. Absence of Certain Changes or Events 18
- 4.11. Contracts and Commitments 19
- 4.12. Taxes and Tax Returns 19
- 4.13. No Material Adverse Change 20
- 4.14. Evidences of Indebtedness 20
- 4.15. Regulatory Compliance 20
- 4.16. Books and Records 20
- 4.17. Financial Products and Services 21
- 4.18. Interest Rate Risk Management Instruments 21
- 4.19. Participations 21
- 4.20. Finder's Fee 21

- 4.21. Custodial Accounts 21
- 4.22. Trust Accounts 21
- 4.23. Deposits 21
- 4.24. Customer Information Security 21
- 4.25. Prior Purchase and Assumption Agreement 21
- 4.26. Available Funds 22
- 4.27. Disclosure 22

# ARTICLE V REPRESENTATIONS AND WARRANTIES OF BOFI 22

- 5.01. Organization and Authority 22
- 5.02. Execution and Delivery 22
- 5.03. Compliance with Laws, Permits and Instruments 23
- 5.04. Consents and Approvals 23
- 5.05. Financial Statements 23
- 5.06. Undisclosed Liabilities 23
- 5.07. Litigation 24
- 5.08. Regulatory Compliance 24
- 5.09. Finder's Fee 24
- 5.10. Association Rights 24
- 5.11. Nevada Branch 24
- 5.12. Disclosure 24

## **ARTICLE VI COVENANTS** 24

- 6.01. Conduct of the Business in the Ordinary Course 24
- 6.02. Required Acts 24
- 6.03. Prohibited Acts 25
- 6.04. Regulatory Approvals 26
- 6.05. Access to Information and Properties 27
- 6.06. Agreements and Consents 27
- 6.07. Fees and Expenses 27
- 6.08. Employees 27
- 6.09. Notification 28
- 6.10. BINs 28
- 6.11. Exclusive Dealing 28
- 6.12. Nevada Branch Opening 28
- 6.13. Execution and Delivery of Program Contracts 28
- 6.14. Further Assurances 29

# ARTICLE VII POST-CLOSING 29

- 7.01. Affiliate Merger 29
- 7.02. Filings; Power of Attorney 30
- 7.03. Overdrafts True-Ups 30

## **ARTICLE VIII CONDITIONS 30**

8.01. Conditions to Obligations of HRB Bank and Block Financial 30

### ARTICLE IX TERMINATION, AMENDMENT, AND WAIVER 33

- 9.01. Termination 33
- 9.02. Effect of Termination 34

## **ARTICLE X INDEMNIFICATION 34**

- 10.01. HRB Bank's Indemnity 34
- 10.02. BofI Indemnity 35
- 10.03. Indemnity Procedure 36
- 10.04. Limitations on Indemnity 37
- 10.05. Exclusive Remedy 38

# ARTICLE XI TRANSITIONAL AND POST-CLOSING MATTERS 38

- 11.01. Notification to Customers 38
- 11.02. Payment of Instruments 39
- 11.03. Statements 39
- 11.04. Access to Records 39
- 11.05. Information Reporting 39
- 11.06. Transition 39

## ARTICLE XII GENERAL PROVISIONS 40

- 12.01. Amendment 40
- 12.02. Waiver 40
- 12.03. Survival of Representations and Warranties 40
- 12.04. Binding Effect; Assignment 40
- 12.05. Severability 40
- 12.06. Headings 40
- 12.07. Entire Agreement 40
- 12.08. Counterparts 41
- 12.09. Notices 41
- 12.10. Governing Law; Jurisdiction; Waiver of Jury Trial 42
- 12.11. Third Party Beneficiaries 43
- 12.12. Specific Performance 43
- 12.13. Mutual Drafting 43
- 12.14. Interpretive Provisions 43

# Signature Page S-1

## List of Schedules

- Schedule 3.02 Sample Transfer Amount Calculation
- Schedule 3.03 Prior Engagements
- Schedule 3.06 Items of Income and Expenses to be Prorated
- Schedule 6.01 Transactions Outside the Ordinary Course of Business
- Schedule 8.01(h) Required Third Party Consents

## List of Exhibits

Closing Ancillary Agreements:

- Exhibit A Assignment and Assumption of Deposit Liabilities Agreement
- Exhibit B Assignment and Assumption of Assumed Contracts Agreement
- Exhibit C Bill of Sale
- Exhibit D Assignment and Assumption of Loans Agreement
- Exhibit E Assignment, Transfer and Appointment of Successor Custodian for Custodial Accounts
- Exhibit F Limited Power of Attorney

## Program Agreements:

- Exhibit G Program Management Agreement
- Exhibit H Guaranty Agreement
- Exhibit I Emerald Advance Receivables Participation Agreement
- Exhibit J Company Financial Products Distribution Agreement
- Exhibit K Franchisee Financial Products Distribution Agreement
- Exhibit L Trademark Licensing Agreement
- Exhibit M Trademark Licensing Agreement
- Exhibit N Collection Services Agreement
- Exhibit O -Amended and Restated Legacy Emerald Advance Participation Agreement
- Exhibit P Amended and Restated Legacy Emerald Unsecured Credit Card Participation Agreement

### PURCHASE AND ASSUMPTION AGREEMENT

THIS PURCHASE AND ASSUMPTION AGREEMENT (this "<u>Agreement</u>"), dated as of April 10, 2014, is made by and among BofI Federal Bank, a federal savings bank ("<u>Boff</u>"), H&R Block Bank, a federal savings bank ("<u>HRB Bank</u>"), and Block Financial LLC, a Delaware limited liability company and the sole shareholder of HRB Bank ("<u>Block Financial</u>").

## RECITALS

A. BofI desires to purchase and assume, and HRB Bank desires to sell and transfer, certain assets and liabilities, including all deposit liabilities, of HRB Bank as further set forth in this Agreement (the "<u>P&A Transaction</u>").

B. Immediately prior to the Closing, HRB Bank will convert to a national banking association (the "<u>Conversion</u>") and immediately following the closing of the P&A Transaction, HRB Bank will merge with and into Block Financial (the "<u>Affiliate</u> <u>Merger</u>"). The Conversion, the P&A Transaction and the Affiliate Merger will occur in immediate succession on the Closing Date. C. Because the Conversion will occur prior to the Closing, HRB Bank will be a national banking association at the time

of the Closing.

D. Contemporaneous with the execution and delivery of this Agreement, the parties hereto and certain of their Affiliates are executing and delivering the Confidentiality and Common Interest Agreement, dated as of April 10, 2014.

#### AGREEMENT

ACCORDINGLY, in consideration of the foregoing, the mutual covenants and agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereby agree as follows:

### ARTICLE I DEFINITIONS

**1.01.** Certain Definitions. Unless the context otherwise requires, capitalized terms used in this Agreement shall have the meanings set forth below.

"<u>Accounting Firm</u>" has the meaning set forth in Section 3.03(b).

"<u>Accrued Interest</u>" means (a) with respect to the Deposits, the interest that has been accrued but not paid on or credited to the Deposits, and (b) with respect to the Loans, the interest, fees, costs and other charges that have accrued on or been charged to the Loans but not paid by the borrower, or if applicable, a guarantor or surety therefor, or otherwise collected by offset or recourse to collateral for the applicable Loan.

"ACH" means the Automated Clearing House system.

"<u>Affiliate</u>" means any Person that, directly or indirectly, through one or more intermediaries, (a) owns or controls another Person, (b) is owned or controlled by another Person, or (c) is under

common control or ownership with another Person, and "ownership" means the direct or indirect beneficial ownership of more than 20% of the equity securities of a Person, or, in the case of a Person that is an entity but not a corporation, more than 20% of the other voting and/or equity interest. For purposes of this definition, "control" (and its derivatives) of a given Person means the power or authority, whether exercised or not, to direct the business, management and policies of such Person, directly or indirectly, whether through the ownership of voting securities, by contract or otherwise, which power or authority shall conclusively be presumed to exist upon possession of beneficial ownership or power to direct the vote of more than 20% of the votes entitled to be cast at meetings of the members or stockholders of such Person or power to control composition of a majority of the members of the board of directors or other governing body of such Person.

"Affiliate Merger" has the meaning set forth in the Recitals.

"<u>Ancillary Agreements</u>" means the merger agreement entered into in connection with the Affiliate Merger, the Program Management Agreement, the Receivables Participation Agreement and the Closing Ancillary Agreements.

"<u>Approving Authorities</u>" means, collectively or individually, as applicable, the OCC, the Board of Governors of the Federal Reserve System and any other Governmental Entity whose approval is necessary for BofI, HRB Bank and Block Financial to consummate the transactions contemplated hereby.

"<u>Assumed Contracts</u>" has the meaning set forth in Section 4.11(b).

"Assumed Liabilities" has the meaning set forth in Section 2.03.

"<u>ATM</u>" means the automated teller machine owned by HRB Bank, which is located at One H&R Block Way, Kansas City, MO 64105.

"Bank Office" means the sole location of HRB Bank located at One H&R Block Way, Kansas City, MO 64105.

"Beneficial Rights" has the meaning set forth in Section 2.05.

"BINs" means the existing bank identification numbers for the Emerald Cards, HRB Gift Cards and HRB Incentive Cards.

"Block Financial" has the meaning set forth in the opening paragraph.

"<u>Block Financial Capital Contribution</u>" means a capital contribution by Block Financial to HRB Bank of an amount of cash that when combined with HRB Bank's cash on hand is sufficient to pay the Estimated Transfer Amount at the Closing.

"BofI" has the meaning set forth in the opening paragraph.

"BofI Disclosure Schedule" has the meaning set forth in Article V.

"BofI Financial Statements" has the meaning set forth in Section 5.05.

"BofI Holding" means BofI Holding, Inc., a Delaware corporation.

"BofI Indemnified Parties" has the meaning set forth in Section 10.01.

"BofI's Knowledge" means the actual and imputed knowledge of Gregory Garrabrants or Andrew J. Micheletti.

"Brokered Deposits" means all Deposits that have been placed through a broker or other intermediary for which HRB Bank has incurred liability for payment of a commission or other remuneration to such broker or intermediary.

"Business Day" means any day except a Saturday, Sunday or other day on which banking institutions located in the State of Missouri or the State of California are authorized by Law to close.

"<u>Call Report Instructions</u>" means, as of any date, the then-applicable instructions to the Reports of Condition and Income as promulgated by the Federal Financial Institutions Examination Council.

"<u>Claim Notice</u>" has the meaning set forth in Section 10.03(a).

"<u>Closing</u>" has the meaning set forth in Section 3.01.

"<u>Closing Ancillary Agreements</u>" means the Assignment and Assumption of Deposit Liabilities Agreement, the Assignment and Assumption of Assumed Contracts Agreement, the Bill of Sale, the Assignment and Assumption of Loans Agreement, the Assignment, Transfer and Appointment of Successor Custodian for Custodial Accounts, and the Limited Power of Attorney.

"Closing Date" has the meaning set forth in Section 3.01.

"Code" means the Internal Revenue Code of 1986, as amended.

"<u>Commitments</u>" means the unfunded portion of a line of credit or other unfunded commitment to make a Consumer Loan (or additional advances with respect to a Consumer Loan) as reflected on the books and records of HRB Bank, that was legally binding on HRB Bank immediately prior to the Closing.

"<u>Consent</u>" shall mean any consent, approval, authorization, clearance, exemption, waiver, or similar affirmation by any Person pursuant to any Contract, Law, Order or Permit.

"<u>Constituent Documents</u>" means collectively or individually, as the case may be, the charter, articles of incorporation or certificate of incorporation and/or the bylaws (and/or any equivalent governing documents) of a Person.

"<u>Consumer Loans</u>" means all Loans made primarily for personal, family or household purposes (excluding, however, all Residential Mortgage Loans), including but not limited to all Emerald Advances for the current tax season, all Emerald Advances for any prior tax season, all Unsecured Credit Card Accounts, and all Unsecured Credit Card Receivables (in each case regardless of whether a participation interest in such Emerald Advances, Unsecured Credit Card Accounts or Unsecured Credit Card Receivables has been sold to a third party).

"<u>Contract</u>" shall mean any contract, agreement, arrangement, authorization, commitment, indenture, instrument, license, lease, obligation, plan, practice, restriction, understanding, or undertaking of any kind or character, or other document to which any Person is a party or that is binding on any Person or its assets or business.

"Conversion" has the meaning set forth in the Recitals.

"Custodial Account" has the meaning set forth in Section 3.07.

"Damages" has the meaning set forth in Section 10.01.

"<u>Deposits</u>" mean all deposit liabilities of HRB Bank, which are defined as deposits in the FDIA, including in each case collected and uncollected Deposits plus Accrued Interest.

"Divestiture Transaction" means the Conversion, the P&A Transaction and the Affiliate Merger, collectively.

"<u>EFS</u>" means Emerald Financial Services, LLC, a Delaware limited liability company.

"Emerald Advance" means each H&R Block Emerald Advance line of credit account established pursuant to an agreement between HRB Bank and any individual.

"Emerald Cards" means the prepaid cards issued under the brand name "Emerald Card" by HRB Bank.

"Estimated Transfer Amount" has the meaning set forth in Section 3.02(d).

"Excluded Assets" has the meaning set forth in Section 2.02.

"Excluded Liabilities" has the meaning set forth in Section 2.04.

"FDIA" means the Federal Deposit Insurance Act, as amended.

"Federal Reserve Borrowings" means all borrowings of HRB Bank from the Federal Reserve Bank of Kansas City.

"FHLB Advances" means all borrowings of HRB Bank from the Federal Home Loan Bank of Des Moines.

"Final Settlement Date" has the meaning set forth in Section 3.04.

"Final Transfer Amount" has the meaning set forth in Section 3.03(a).

"Final Transfer Payment" has the meaning set forth in Section 3.04.

"<u>Fundamental Representations</u>" means (i) with respect to BofI, the representations and warranties set forth in Section 4.01 (Organization and Authority), Section 4.02 (Execution and Delivery), Section 4.04 (Compliance with Laws, Permits and Instruments), Section 4.08 (Consents and Approvals), Section 4.09 (Ownership of Assets), and Section 4.20 (Finder's Fee) and (ii) with respect to HRB Bank and Block Financial, the representations and warranties set forth in Section 5.01 (Organization and Authority), Section 5.02 (Execution and Delivery), Section 5.03 (Compliance with Laws, Permits and Instruments), Section 5.04 (Consents and Approvals), Section 5.09 (Finder's Fee), Section 5.10 (Association Rights) and Section 5.11 (Nevada Branch).

"GAAP" means generally accepted accounting principles as in effect in the United States.

"<u>Governmental Entity</u>" means any United States, foreign, federal, state or local court, tribunal, judicial body, arbitral body, administrative agency or commission, or other governmental instrumentality.

"H&R Block" means, collectively, HRB and its subsidiaries.

"HRB" means H&R Block, Inc., a Missouri corporation.

"<u>HRB Bank</u>" has the meaning set forth in the opening paragraph. Following consummation of the Affiliate Merger, references to HRB Bank in this Agreement shall be deemed to refer to Block Financial, as successor to HRB Bank pursuant to the Affiliate Merger.

"HRB Bank Common Stock" has the meaning set forth in Section 4.03.

"HRB Bank Disclosure Schedule" has the meaning set forth in Article IV.

"HRB Bank Indemnifying Parties" means HRB Bank and Block Financial.

"HRB Bank Indemnified Parties" has the meaning set forth in Section 10.02.

"HRB Bank's Knowledge" means the actual and imputed knowledge of Jeffrey Brown, Greg Macfarlane, Greg Quarles, Alana Neale or Jim Koger.

"HRB Financial Statements" has the meaning set forth in Section 4.05.

"HRB Gift Cards" means the prepaid cards issued by HRB Bank and identified as "gift cards."

"HRB Incentive Cards" means the prepaid cards issued by HRB Bank and identified as "incentive cards."

"HRB Prepaid Cards" means the Emerald Cards, HRB Gift Cards and HRB Incentive Cards.

"HRB Prepaid Card Balances" means the balances on HRB Prepaid Cards.

"Indemnified Party" has the meaning set forth in Section 10.03.

"Indemnifying Party" has the meaning set forth in Section 10.03(a).

"Investment Securities" means all of the readily marketable investment securities in HRB Bank's investment portfolio.

"<u>Law</u>" means any federal or state law, statute, regulation, rule, or reporting or licensing requirement applicable to a Person or its assets, liabilities or business, including those promulgated, interpreted, or enforced by any Governmental Entity.

"<u>Liability</u>" means any direct or indirect indebtedness, guaranty, endorsement, claim, loss, damage, deficiency, cost, expense, obligation or responsibility, fixed or unfixed, contingent or non-contingent, known or unknown, asserted or unasserted, liquidated or unliquidated, secured or unsecured.

"<u>Lien</u>" means any lien, pledge, claim, charge, security interest, mortgage, easement, covenant, restriction, or encumbrance of any nature whatsoever, but excluding participation interests in the Transferred Loans.

"<u>Litigation</u>" means any complaint, action, suit, proceeding, review, arbitration or other alternative dispute resolution procedure, demand, claim, investigation or inquiry of any kind or nature, whether civil, criminal or administrative.

"Loans" means all of the following owed to or held by HRB Bank:

(a) All loans, funded portions of lines of credit or credit plans (whether revolving or not and whether commercial or consumer), including loans secured by Deposits, extensions of credit pursuant to a credit card plan or any other extensions of credit;

(b) All liens, rights (including rights of setoff), remedies, powers, privileges, demands, claims, priorities, equities and benefits owned or held by, or accruing or to accrue to or for the benefit of, the holder of the obligations or instruments referred to in clause (a) above, including

those arising under or based upon all Records evidencing or securing a Loan (including any note, mortgage, deed of trust, security agreement, guarantee, other collateral documents, attorney opinions, all other tangible or electronic items relating to the Loans and all amendments, modifications, extensions or renewals of any of the foregoing), casualty insurance, gap/warranty and credit life insurance policies and binders, standby letters of credit, mortgagee title insurance policies and binders, payment bonds and performance bonds at any time and from time to time existing with respect to any of the obligations or instruments referred to in clause (a) above; and

(c) All amendments, modifications, renewals, extensions, refinancings, and refundings of or for any of the foregoing, and any financing statements, participation agreements, subordination agreements, and intercreditor agreements related to any of the foregoing.

"MasterCard" means MasterCard Incorporated, a Delaware corporation.

"<u>Material Adverse Change</u>" with respect to a party shall mean an event, change, or occurrence which, individually or together with any other event, change, or occurrence, has or would reasonably be expected to have a material adverse effect on (a) the business, regulatory status, assets, financial position or prospects of such party, or results of operations of such party, (b) the ability of such party to perform its obligations under this Agreement or to consummate the transactions contemplated by this Agreement or (c) in the case of HRB Bank and Block Financial, the Transferred Assets or Assumed Liabilities, <u>provided</u>, <u>however</u>, that none of the following will be taken into account in determining whether there has been a Material Adverse Change, unless such event, change, or occurrence, individually or together with any other event, change, or occurrence, has or would reasonably be expected to have a disproportionate impact on the business, regulatory status, assets, financial position or prospects of such party: (i) adverse changes in GAAP or regulatory accounting requirements; (ii) adverse changes in Laws of general applicability to companies in the U.S. banking, financial services or tax preparation industries; (iii) adverse changes in global or national political conditions or general economic or market conditions (including changes in prevailing interest rates, and currency exchange rates) affecting other companies in the U.S. banking industry; or (iv) any outbreak or escalation of hostilities, declared or undeclared acts of war or terrorism.

"Materially Burdensome Regulatory Condition" has the meaning set forth in Section 6.04(e).

"Mortgage Holdings, TLC, a Delaware limited liability company.

"Nevada Branch" has the meaning set forth in Section 5.11.

"Non-Assigned Assets" has the meaning set forth in Section 2.05.

"<u>Nondisclosure Agreement</u>" means the Amended and Restated Letter Agreement, dated as of October 21, 2013, by and among Block Financial, HRB Tax Group, Inc., and BofI Holding, Inc.

"<u>Notice Period</u>" has the meaning set forth in Section 10.03(a).

"<u>OCC</u>" means the Office of the Comptroller of the Currency.

"<u>Operating Cash</u>" means all teller working cash, petty cash, vault cash and any other cash of HRB Bank located at the Bank Office or in the ATM, as of the close of business on the Closing Date.

"<u>Order</u>" shall mean any order, injunction, judgment, decision, award, decree, ruling, writ, or administrative decision of any Governmental Entity.

"<u>OREO</u>" means all real property acquired by HRB Bank through foreclosure or satisfaction of judgments or indebtedness, together with all improvements, fixtures and appurtenances located thereon and owned by HRB Bank as of the Closing Date.

"<u>Other Correspondent Accounts</u>" means all deposit accounts of HRB Bank located at any other bank or financial institution, other than the UMB Correspondent Accounts.

"Overdrafts" means negative balances on Deposit accounts and HRB Prepaid Cards.

"<u>P&A Transaction</u>" has the meaning set forth in the Recitals.

"Pending Litigation" has the meaning set forth in Section 4.07.

"<u>Permit</u>" shall mean any permit, approval, authorization, certificate, easement, franchise, license, or right given by any Governmental Entity to any Person, or that is or may be binding upon or inure to the benefit of any Person, or its assets or business.

"<u>Person</u>" means an individual, corporation, partnership, limited liability company, unincorporated association, trust, joint venture or other organization or entity or Governmental Entity and shall include any successor (by merger or otherwise) of such entity.

"<u>Personal Property</u>" means the ATM and all other equipment, furniture, appliances, fixtures, furnishings, computer equipment (hardware and software), materials, supplies, signs, structures, improvements and other items of tangible personal property owned by HRB Bank and located at the Bank Office.

"Post-Closing Transfer Calculation" has the meaning set forth in Section 3.03(a).

"Program Agreements" has the meaning set forth in Section 6.13.

"<u>Program Management Agreement</u>" means the Program Management Agreement, including the product schedules attached thereto, to be entered into by Bofl, Block Financial and EFS concurrent with the Closing, pursuant to which EFS will provide certain services to Bofl and will market and distribute Bofl products and services.

"Pro Rata Adjustment" has the meaning set forth in Section 3.06.

"<u>Real Property</u>" means the real property leased by HRB Bank in which the Bank Office is located, together with all improvements, improvements in progress, fixtures and appurtenances located thereon or related thereto.

"<u>Receivables Participation Agreement</u>" means the Emerald Advance Receivables Participation Agreement to be entered into by BofI, HRB and a subsidiary of Block Financial concurrent with the Closing, pursuant to which, from time to time, BofI will sell and such subsidiary will purchase participation interests in certain Emerald Advance Loans originated by BofI.

"<u>Records</u>" means all books, records and files relating primarily to the Transferred Assets and the Assumed Liabilities, in every format, in the possession or control of HRB Bank or any of its Affiliates, agents or service providers.

"<u>Required Regulatory Approval</u>" means a regulatory authorization, consent, order or approval from an Approving Authority required to complete any part of the Divestiture Transaction, including but not limited to (a) with respect to HRB Bank, approval of (i) the OCC, as the regulator of national banks, to complete (A) the Conversion, (B) the P&A Transaction and (C) the Affiliate Merger and (ii) the OCC, as the regulator of federal savings banks, to complete the Conversion; (b) with respect to Block Financial, H&R Block Group, Inc., and HRB, approval of the Board of Governors of the Federal Reserve System (i) of a waiver of the application and filing requirements under section 3(a)(1) and the notification requirements under Section 4 of the Bank Holding Company Act of 1956 and (ii) to deregister as savings and loan holding companies under the Home Owners Loan Act; and (c) with respect to BofI, approval of the OCC to complete the P&A Transaction, as appropriate.

"<u>Residential Mortgage Loans</u>" means each first or second lien residential mortgage Loan including all of HRB Bank's right, title and interest in and to such Loan, including the servicing rights and all escrows related to such Loan.

"<u>Retained Contracts</u>" means all Contracts of HRB Bank that are not set forth in Section 4.11(b) of the HRB Bank Disclosure Schedule.

"Secured Credit Card Account" means each open-end credit account that is accessed by a credit card established pursuant to a cardholder agreement between HRB Bank and any individual, that is secured by a Deposit.

"Secured Credit Card Receivables" means any amounts shown on HRB Bank's records as amounts payable by obligors on any Secured Credit Card Account.

"<u>Tax</u>" or "<u>Taxes</u>" means any federal, state, local, or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental (including taxes under Code §59A), customs, duties, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, escheat, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated, or other tax of any kind whatsoever, including any interest, penalty, or addition thereto, whether disputed or not and including any obligation to indemnify or otherwise assume or succeed to the Tax liability of any other Person.

"<u>Tax Return</u>" means any return, declaration, report, claim for refund, or information return or statement relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"Third Party Claim" has the meaning set forth in Section 10.03(a).

"Threshold Amount" has the meaning set forth in Section 10.04(a).

"Time Deposits" means Deposits that are time deposits or certificates of deposit.

"<u>Transfer Amount</u>" has the meaning set forth in Section 3.02(b).

"Transferred Assets" has the meaning set forth in Section 2.01.

"<u>Transferred Loans</u>" means all Consumer Loans, regardless of whether any such Consumer Loan has been written off on the books of HRB Bank or participated, in whole or in part, to a third party, including all Accrued Interest thereon.

"<u>UMB Correspondent Accounts</u>" means the three deposit accounts of HRB Bank with UMB Bank, n.a. which are HRB Bank's primary operating and clearing accounts.

"<u>Unsecured Credit Card Account</u>" means each unsecured open-end credit account that is accessed by a credit card established pursuant to a cardholder agreement between HRB Bank and any individual.

"<u>Unsecured Credit Card Receivables</u>" means all amounts shown on HRB Bank's records as amounts payable by obligors on any Unsecured Credit Card Account.

## ARTICLE II PURCHASE OF ASSETS AND ASSUMPTION OF LIABILITIES

**2.01. Purchase of Assets**. Subject to the terms, conditions, representations and warranties set forth in this Agreement, at the Closing, HRB Bank will sell, assign, transfer, grant, bargain, deliver and convey to BofI, and BofI will purchase and otherwise acquire from HRB Bank, free and clear of any Lien, HRB Bank's entire right, title and interest in and to the following assets (the "<u>Transferred Assets</u>"):

(a) the Transferred Loans, and all Accrued Interest on the Transferred Loans;

- (b) the Overdrafts;
- (c) the Assumed Contracts;
- (d) the Records;

(e) the Contracts between HRB Bank and each depositor governing the terms and conditions of the Deposits (including the Brokered Deposits), the HRB Prepaid Cards and the Transferred Loans;

(f) all ABA routing numbers, BINs, and customer account numbers, transit routing numbers, and any other account numbers used or held for use in connection with the Transferred Assets and/or the Assumed Liabilities; and

(g) all security and/or other deposits and prepaid expenses exclusively relating to the Transferred Assets.

No later than three (3) Business Days prior to the Closing Date, HRB Bank shall complete and provide to BofI (i) an electronic schedule, in form and substance reasonably acceptable to BofI, setting forth such information as BofI may reasonably request with respect to the Transferred Loans as they existed as of the cut-off date for such schedule which shall be not earlier than the fifth (5 <sup>th</sup>) Business Day prior to the Closing Date and (ii) an electronic schedule in form and substance reasonably acceptable to BofI setting forth such information as BofI may reasonably request with respect to the Overdrafts as they existed as of the cut-off date for such schedule which shall be not earlier than the fifth (5 <sup>th</sup>) Business Day prior to the Closing Date. The parties acknowledge and agree that the schedules to be delivered as provided in the preceding sentence are preliminary and

will not account for changes in the Transferred Loans or Overdrafts between the cut-off dates of such schedules and the Closing Date. No later than five (5) Business Days after the Closing Date, HRB Bank shall deliver final electronic versions of the aforementioned schedules containing information regarding the Transferred Loans and the Overdrafts as of the close of business on the Closing Date.

2.02. Excluded Assets. Notwithstanding anything to the contrary in Section 2.01, all of the assets of HRB Bank that are not Transferred Assets, including the following assets, will be retained by HRB Bank and are excluded from the Transferred Assets (the "<u>Excluded Assets</u>"):

(a)	all Investment Securities;
(b)	all Residential Mortgage Loans;
(c)	all OREO;
(d)	all commercial and industrial Loans;
(e)	the Retained Contracts;
(f)	HRB Bank's membership interest in EFS;
(g)	HRB Bank's membership interest in Mortgage Holdings;
(h)	HRB Bank's equity interest in any other subsidiary or other entity;
(i) HRB Bank's stock in the Federal Home Loan Bank of Des Moines (and any right to the proceeds of the redemption of such stock);	
(j)	all claims, demands or causes of action for refunds of taxes and governmental charges related to HRB Bank's
(k) HRB Bank's corporate seal, minute books, charter, corporate stock record books, tax records, organizational documents and such other books and records as pertain to the organization, taxation or capitalization of HRB Bank;	
(1)	any claims, demands or causes of action or rights associated with the Excluded Assets or the Excluded
(m)	any intellectual property owned by, or licensed to, HRB Bank;
(n)	any intercompany receivables from an Affiliate of HRB Bank;
(0)	the Secured Credit Card Accounts;
(p)	the Secured Credit Card Receivables;
(q)	the Other Correspondent Accounts;
(r)	the Operating Cash;
(s)	the Personal Property;
(t)	the Real Property; and
(u)	the UMB Correspondent Accounts.
	(b) (c) (d) (e) (f) (g) (h) f such stock (j) (k) nd such other (l) (m) (n) (o) (p) (q) (r) (s) (t)

**2.03.** Assumed Liabilities. Upon the terms and subject to the conditions set forth in this Agreement, on the Closing Date, BofI will assume and agree to pay and perform, as and when due, only those Liabilities of HRB Bank listed below in this Section 2.03 (the "Assumed Liabilities"):

- (a) all Deposits;
- (b) all HRB Prepaid Card Balances;

(c) the payment and performance of obligations and other Liabilities (other than Excluded Liabilities) arising after the Closing Date relating to the Transferred Assets or the Assumed Contracts (other than (i) obligations or Liabilities arising out of or relating to a breach or non-performance of the terms thereof prior to the Closing and (ii) obligations or Liabilities payable after Closing but subject to proration as provided in Section 3.06, which obligations or Liabilities will be paid by BofI for HRB Bank's account after Closing); and

(d) the obligations under the Commitments (other than obligations arising out of or relating to a breach or nonperformance thereof prior to the Closing).

No later than three (3) Business Days prior to the Closing Date, HRB Bank shall complete and provide to Bofl (i) an electronic schedule, in form and substance reasonably acceptable to Bofl, setting forth such information as Bofl may reasonably request with respect to the Deposits as they existed as of the cut-off date for such schedule which shall be not earlier than the fifth (5 <sup>th</sup>) Business Day prior to the Closing Date, (ii) an electronic schedule, in form and substance reasonably acceptable to Bofl, setting forth such information as Bofl may reasonably request with respect to the HRB Prepaid Card Balances as they existed as of the cut-off date for such schedule which shall be not earlier than the fifth (5 <sup>th</sup>) Business Day prior to the Closing Date, and (iii) an electronic schedule, in form and substance reasonably acceptable to Bofl, setting forth such information as Bofl may reasonably acceptable to Bofl, setting forth such information as Bofl may reasonably acceptable to Bofl, setting forth such information as Bofl may reasonably acceptable to Bofl, setting forth such information as Bofl may reasonably acceptable to Bofl, setting forth such information as Bofl may reasonably request with respect to the Closing Date, and (iii) an electronic schedule, in form and substance reasonably acceptable to Bofl, setting forth such information as Bofl may reasonably request with respect to the Closing Date. The parties acknowledge and agree that the schedule which shall be not earlier than the fifth (5 <sup>th</sup>) Business Day prior to the Closing Date. The parties acknowledge and agree that the schedules to be delivered as provided in the preceding sentence are preliminary and will not account for changes in the Deposits, HRB Prepaid Card Balances and the Commitments between the cut-off dates of such schedules and the Closing Date. No later than five (5) Business Days after the Closing Date, HRB Bank shall deliver final electronic versions of the aforementioned schedules containing information regarding the Deposits,

**2.04.** Excluded Liabilities. Except as and to the limited extent specifically set forth in Section 2.03, BofI is not assuming any Liabilities of HRB Bank. HRB Bank is, and following the Closing HRB Bank will continue to be, responsible for all Liabilities of HRB Bank other than the Assumed Liabilities (collectively, the "<u>Excluded Liabilities</u>"). Without limiting the generality of the foregoing, the Excluded Liabilities include the following:

(a) any Liability arising out of or in connection with the business of HRB Bank or any transactions or series of transactions, any facts or series of facts existing, or any events or series of events to the extent they occurred on or prior to the Closing Date (other than the Liabilities expressly assumed by BofI pursuant to Section 2.03);

(b) any Liability with respect to employment or consulting agreements, pension, profit-sharing, welfare or benefit plans, or amounts owing for commissions or compensation,

termination, severance or other payments to present or former employees, officers, managers or members of HRB Bank and/or to the spouse, dependents, and beneficiaries of such individuals, regardless of whether any such person is employed by HRB Bank or BofI following the Closing;

(c) any Liability of HRB Bank for any Taxes of any kind or nature, or any interest or penalties thereon, including any of HRB Bank's Tax obligations arising out of the transactions contemplated hereunder;

(d) all FHLB Advances (if any) and Federal Reserve Borrowings (if any);

(e) any Liability under or relating to the Retained Contracts;

(f) any Liability arising from or relating to the Excluded Assets;

(g) any Liability for commissions or other payments due to brokers on any Brokered Deposits incurred or accrued prior to the Closing Date;

(h) any Liabilities arising from or related to the Pending Litigation, which are expressly retained by HRB Bank and expressly excluded from the Assumed Liabilities to be assumed by BofI; and

(i) any fee payable by HRB Bank pursuant to its engagement letter with either of Goldman, Sachs. & Co. or First Annapolis Consulting, Inc. or any other broker, investment bank, consultant or other advisor engaged by HRB Bank, Block Financial or any Affiliate thereof and any other costs, expenses or other Liabilities incurred by HRB Bank, Block Financial or any Affiliate thereof in connection with the Divestiture Transactions.

**2.05.** Nonassignable Contracts and Rights . Notwithstanding anything to the contrary in this Agreement, no Contracts, Permits, properties, rights or other assets (collectively, the "<u>Non-Assigned Assets</u>") of HRB Bank shall be deemed sold, transferred or assigned to BofI pursuant to this Agreement if the attempted sale, transfer or assignment thereof to BofI without the Consent of any other Person would be ineffective or would constitute a default or breach of Contract or a violation of any Law or Order, and such Consent is not obtained at or prior to the Closing. In such case, to the extent possible, (a) the beneficial interest in or to such Non-Assigned Assets (collectively, the "<u>Beneficial Rights</u>") shall in any event pass at the Closing to BofI under this Agreement; and (b) pending such Consent, BofI shall discharge, as agent for HRB Bank, the obligations of HRB Bank under such Beneficial Rights to the extent such obligations are Assumed Liabilities, and HRB Bank shall act as BofI's agent in the receipt of any benefits, rights or interest received from the Beneficial Rights. With respect to each Non-Assigned Asset, HRB Bank and Block Financial will comply with the covenants contained in Section 6.13(b).

## ARTICLE III CLOSING AND TRANSFER AMOUNT

**3.01.** Closing. Unless this Agreement shall have been terminated in accordance with Article IX, the closing of the transactions contemplated by this Agreement (the "<u>Closing</u>") shall take place on a date mutually agreed upon by HRB Bank and BofI, and absent mutual agreement then on the last Business Day of a calendar month following the satisfaction or waiver by the party entitled to the benefit thereof of each of the conditions to Closing set forth in Article VIII hereof (the "<u>Closing Date</u>"); provided that, unless otherwise mutually agreed upon by HRB Bank and BofI,

the Closing Date shall not occur prior to May 31, 2014, and if the satisfaction or waiver by the party entitled to the benefit thereof of each of the conditions to Closing set forth in Article VIII hereof occurs prior to May 1, 2014, then the Closing Date shall be May 31, 2014. The P&A Transaction, and all calculations made in connection therewith, shall be effective as of the close of HRB Bank's banking business on the Closing Date. The Closing will take place, subject to the satisfaction or waiver of all conditions set forth in Article VIII, at the offices of Stinson Leonard Street LLP, 1201 Walnut Street, Suite 2900, Kansas City, Missouri 64106.

### 3.02. Transfer Amount.

(a) In connection with the sale by HRB Bank to BofI of the Transferred Assets and the assumption by BofI of the Assumed Liabilities as provided for herein, at the Closing, if:

- i. the Estimated Transfer Amount is negative, HRB Bank will transfer to BofI by wire transfer of immediately available funds to an account designated by BofI an amount equal to the Estimated Transfer Amount; or
- ii. the Estimated Transfer Amount is positive, BofI will transfer to HRB Bank by wire transfer of immediately available funds to an account designated by HRB Bank an amount equal to the Estimated Transfer Amount.
- (b) For purposes of this Agreement, the <u>"Transfer Amount"</u> will be equal to:
  - i. the aggregate book value of the Transferred Loans, plus all Accrued Interest on the Transferred Loans;
  - ii. *plus,* the aggregate book value of the Overdrafts;
  - iii. *minus*, the sum of (A) aggregate par value of the Deposits (without a premium), *plus* (B) all accrued and unpaid interest on the Deposits;
  - iv. plus or minus (as applicable) the Pro Rata Adjustment.

(c) If the Estimated Transfer Amount is a negative number then, prior to the Closing Date, Block Financial will make the Block Financial Capital Contribution; *provided, however*, that the failure of Block Financial to timely make the Block Financial Capital Contribution shall not relieve Block Financial or HRB Bank of their respective obligations under this Agreement.

(d) Three (3) Business Days prior to the Closing Date, HRB Bank shall deliver to BofI a written calculation of the Transfer Amount setting forth in reasonable detail HRB Bank's good faith best estimate of the Transfer Amount calculated as of the close of business on the date that is five (5) Business Days prior to the Closing Date, determined in accordance with GAAP (the "<u>Estimated Transfer Amount</u>") and shall make available the work papers, schedules and other supporting data used by HRB Bank to calculate and prepare the Estimated Transfer Amount. The Estimated Transfer Amount will be subject to the approval of BofI, which approval shall not be unreasonably delayed, conditioned or withheld.

(e) A pro forma hypothetical calculation of the Transfer Amount is included in Schedule 3.02.

#### 3.03. Post-Closing Transfer Calculation.

(a) Not later than the twentieth (20<sup>th</sup>) Business Day after the Closing Date, HRB shall deliver to BofI a written calculation (the "<u>Post-Closing Transfer Calculation</u>") of the Transfer Amount setting forth in reasonable detail the Transfer Amount calculated as of the close of business on the Closing Date, determined in accordance with GAAP (the "<u>Final Transfer Amount</u>") and shall make available the work papers, schedules and other supporting data used by HRB Bank to calculate and prepare the Post-Closing Transfer Calculation to enable BofI to verify the accuracy of the Final Transfer Amount.

(b) If HRB Bank and BofI are unable to agree upon the Post-Closing Transfer Calculation within ten (10) Business Days after HRB delivers the Post-Closing Transfer Calculation, the disputed items or amounts concerning the Post-Closing Transfer Calculation shall be determined by PricewaterhouseCoopers LLP (the "Accounting Firm") within fourteen (14) Business Days following the submission of such disputed items or amounts to the Accounting Firm. In making such calculation, the Accounting Firm may only consider those items or amounts in the Post-Closing Transfer Calculation as to which HRB Bank and BofI have disagreed, and the Accounting Firm shall act as an expert and not an arbitrator. The fees of the Accounting Firm incurred in determining the Post-Closing Transfer Calculation will be paid equally by HRB Bank and BofI. Except as set forth on <u>Schedule 3.03</u>, neither HRB Bank, Block Financial or their Affiliates (on the one side), or BofI, BofI Holding or their Affiliates (on the other side) have engaged the Accounting Firm for services in the last two years.

**3.04. Final Settlement**. On the Business Day immediately following the day on which the Post-Closing Transfer Calculation shall have been finally determined pursuant to the terms of Section 3.03 (the "<u>Final Settlement Date</u>"), if the Final Transfer Amount is greater than the Estimated Transfer Amount, then BofI shall pay the difference to HRB Bank. If the Final Transfer Amount is less than the Estimated Transfer Amount, then HRB Bank shall pay the difference to BofI. In each case such payment shall be made within three (3) Business Days after the Final Settlement Date by wire transfer in immediately available funds to an account designated in writing by HRB Bank to BofI or to an account designated in writing by BofI to HRB Bank, as applicable. Any payment pursuant to this Section 3.04 (the "<u>Final Transfer Payment</u>") shall include interest on such amount for the number of days from and including the Closing Date to but excluding the Final Settlement Date at a rate equal to the effective Federal Funds rate as published by the Board of Governors of the Federal Reserve System on the Final Settlement Date. If the Final Settlement Date is not a Business Day, the Federal Funds Rate shall be the rate applicable to federal funds transactions on the immediately preceding day for which such rate is reported.

**3.05.** Purchase Price Allocation. The allocation of the consideration paid by BofI is intended to comply with the allocation method required by Section 1060 of the Code. The parties will each report the federal, state and local and other tax consequences of the P&A Transaction (including the filing of Internal Revenue Service Form 8594) in a manner consistent with such allocation and take no position in any tax filing, return, proceeding, audit or otherwise which is inconsistent with such allocation, unless required to do so by law as interpreted or modified subsequent to the Closing Date. Within thirty (30) days following the Closing Date, BofI shall provide to HRB Bank a schedule showing BofI's proposed allocation of the consideration payable with respect to the Transferred Assets. BofI, on the one hand, and HRB Bank on the other, shall

agree upon such allocation within sixty (60) days after the Closing Date; provided that if BofI and HRB Bank are unable to agree upon an allocation schedule within such sixty (60) day period, the allocation schedule shall be prepared by the Accounting Firm, who shall be mutually engaged by HRB Bank and BofI for such purpose. The fees of the Accounting Firm incurred in preparing the allocation schedule shall be paid equally by BofI and HRB Bank.

**3.06. Prorations**. The parties intend that HRB Bank will own the Transferred Assets and be responsible for the Assumed Liabilities for its own account until the close of business on the Closing Date, and that Bofl will own the Transferred Assets and be responsible for the Assumed Liabilities for its own account after the close of business on the Closing Date. Thus, except as otherwise specifically provided in this Agreement, items of expense and income directly attributable to the Transferred Assets and Assumed Liabilities, other than any general overhead expenses of HRB Bank, will be prorated as of the close of business on the Closing Date, whether or not such adjustment would normally be made as of such time, including any payments due or made on any Assumed Contracts. The aggregate net amount of such proration shall result in an adjustment (the "<u>Pro-Rata Adjustment</u>") in the calculation of the Estimated Transfer Amount and the Final Transfer Amount as provided for in Section 3.02 and Section 3.04, respectively. A schedule setting forth the items of income and expense expected to be prorated with respect to the Transferred Assets and Assumed Liabilities is attached as Schedule 3.06.

**3.07.** Custodial Accounts. At the Closing, HRB Bank will resign as custodian with respect to any individual retirement account included as part of the Assumed Liabilities as to which HRB Bank is custodian (the "<u>Custodial Account</u>"). At the Closing, HRB Bank will designate and appoint BofI as successor custodian under each such Custodial Account. BofI covenants and agrees that it will, following its designation or appointment as successor custodian under the Custodial Accounts, perform, fulfill, and discharge each of the obligations required to be performed by the custodian with respect to such Custodial Accounts pursuant to Law, or pursuant to the governing documents establishing such Custodial Account.

## ARTICLE IV REPRESENTATIONS AND WARRANTIES OF HRB BANK AND BLOCK FINANCIAL

As a material inducement to BofI to enter into and perform its obligations under this Agreement, except as disclosed in the Disclosure Schedule (the "<u>HRB Bank Disclosure Schedule</u>") delivered by HRB Bank and Block Financial to BofI before the execution of this Agreement, HRB Bank and Block Financial, jointly and severally, hereby represent and warrant to BofI that the statements contained in this Article IV are true and correct as of the date hereof (unless they are made as of a specific date, in which case they shall be true and correct as of such specific date) and will be true and correct as of the Closing (unless they are made as of a specific date, in which case they shall be true and correct as of such specific date);

#### 4.01. Organization and Authority.

(a) HRB Bank is a federal savings bank, duly organized and validly existing under the Laws of the United States. HRB Bank has all requisite corporate power and authority to carry on its business as now being conducted, to own, lease and operate its properties and assets as now owned, leased or operated and to enter into and to carry on the business and activities now conducted by it. True and complete copies of all HRB Bank Constituent Documents, as amended to date, have been made available to BofI. HRB Bank is an insured bank as defined in the FDIA. HRB Bank does not own or control any Affiliate or subsidiary other than EFS and Mortgage Holdings. HRB Bank is the sole Member of EFS and Mortgage Holdings. Other than the Member Interests in EFS and Mortgage Holdings, HRB Bank has no equity interest, direct or indirect, in any other corporation or in any partnership, joint venture or other business enterprise or entity.

(b) Block Financial is a limited liability company, duly organized and validly existing and in good standing under the laws of the State of Delaware. Block Financial has all requisite limited liability company power and authority to carry on its business as now being conducted, to own, lease and operate its properties and assets as now owned, leased or operated and to enter into and to carry on the business and activities now conducted by it, and is duly qualified to do business and is in good standing in each jurisdiction in which the nature of its business or the ownership, leasing or operation of its properties and assets makes such qualification necessary, except where the failure to be so qualified and in good standing would not reasonably be expected to create a Material Adverse Change.

#### 4.02. Execution and Delivery.

(a) HRB Bank has the corporate power and authority to execute, deliver and perform this Agreement, the Ancillary Agreements to which it is a party, and any documents, agreements or instruments to be executed by HRB Bank pursuant to this Agreement or any Ancillary Agreement, and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and the Ancillary Agreements to which it is a party by HRB Bank and the consummation by HRB Bank of the transactions contemplated hereby have been duly authorized by the Board of Directors of HRB Bank and, other than the approval of the Divestiture Transaction by the sole shareholder of HRB Bank (which approval will be obtained promptly following the execution of this Agreement), no other corporate proceedings are necessary to authorize this Agreement and the Divestiture Transaction. This Agreement has been, and the Ancillary Agreements to which it is a party and the other agreements and documents contemplated hereby and thereby, have been or at Closing will be, duly executed by HRB Bank, and each constitutes or will, assuming the due execution thereof by each of the other parties thereto, constitute the legal, valid and binding obligation of HRB Bank, enforceable in accordance with its respective terms and conditions, except as enforceability may be limited by bankruptcy, insolvency, fraudulent conveyance, reorganization, receivership, moratorium or similar Laws affecting creditors' rights and remedies and by general principles of equity, including principles of commercial reasonableness, good faith and fair dealing, regardless of whether enforcement is sought in a proceeding at law or in equity.

(b) Block Financial has the limited liability company power and authority to execute, deliver and perform this Agreement, the Ancillary Agreements to which it is a party, and any documents, agreements or instruments to be executed by Block Financial pursuant to this Agreement or any Ancillary Agreement to which it is a party, and to consummate the transactions contemplated hereby and thereby. Block Financial has taken all limited liability company action necessary (and no further action or proceeding on the part of Block Financial or its members is necessary) to authorize the execution, delivery and performance of this Agreement, the Ancillary Agreements to which it is a party and any documents, agreements or instruments to be executed by Block Financial pursuant to this Agreement, the Ancillary Agreements to which it is a party, and to consummate the transactions contemplated hereby and thereby. This Agreement has been, and the Ancillary Agreements to which it is a party and the other agreements and documents contemplated hereby and thereby, have been or at Closing will be, duly executed by Block Financial and each constitutes or will, assuming the due execution thereof by each of the other parties thereto, constitute the legal, valid and binding obligation of Block Financial, enforceable in accordance with its respective terms and conditions, except as enforceability may be limited by bankruptcy, insolvency, fraudulent conveyance, reorganization, receivership, moratorium or similar laws affecting creditors' rights and remedies generally and by general principles of equity, including principles of commercial reasonableness, good faith and fair dealing, regardless of whether enforcement is sought in a proceeding at law or in equity.

**4.03. Capitalization of HRB Bank**. The entire authorized capital stock of HRB Bank consists of 10,000 shares of common stock, par value \$1.00 per share (the "<u>HRB Bank Common Stock</u>"). As of the date hereof, 100 shares of HRB Bank Common Stock are issued and outstanding and all 100 of such shares are owned, legally and beneficially, by Block Financial. All of the shares of HRB Bank Common Stock have been duly authorized, validly issued, and are fully paid and nonassessable, and have not and will not have been issued in violation of the preemptive rights of any Person.

#### 4.04. Compliance with Laws, Permits and Instruments.

(a) HRB Bank has performed and abided by all obligations required to be performed by it to the date hereof, and has complied with, and is in compliance with, and is not in default (and with the giving of notice or the passage of time will not be in default) under, or in violation of, (i) any provision of the HRB Bank Constituent Documents, (ii) any provision of any Contract applicable to the Transferred Assets or Assumed Liabilities, except where nonperformance, noncompliance, default or violation would not reasonably be expected to result in a Material Adverse Change, (ii) any Law, Order or Permit applicable to HRB Bank, except where nonperformance, noncompliance, default or violation would not reasonably be expected to result in a Material Adverse Change, or (iv) any Law, Order or Permit applicable specifically to the Transferred Assets or Assumed Liabilities, in any material respect.

(b) The execution, delivery and (provided the Required Regulatory Approvals are obtained) performance of this Agreement and the other agreements contemplated hereby, and the completion of the transactions contemplated hereby and thereby will not conflict with, result in any violation or breach of or result in a default, under (i) the HRB Bank Constituent Documents (ii) any provision of any Contract, applicable to the Transferred Assets or Assumed Liabilities, except where such conflict, breach, default or violation would not reasonably be expected to result in a Material Adverse Change, or (iii) any Law, Order or Permit applicable to HRB Bank, the Transferred Assets or Assumed Liabilities, in any material respect.

**4.05. Financial Statements**. HRB Bank and Block Financial have made available to BofI true and complete copies of the financial statements of each of HRB Bank and Block Financial identified in Section 4.05 of the HRB Bank Disclosure Schedule (the "<u>HRB Financial Statements</u>"). The HRB Financial Statements (including the related notes) complied as to form, as of their respective dates, in all material respects with applicable accounting requirements, have been prepared in accordance with GAAP, fairly present the financial condition of each of HRB Bank and

Block Financial at the dates thereof and the results of operations for the periods then ended (subject, in the case of unaudited financial statements, to notes and normal year-end audit adjustments that were not material in amount or effect), and the accounting records underlying the HRB Financial Statements accurately and fairly reflect in all material respects the transactions of HRB Bank and Block Financial. The Assumed Liabilities have been reflected in the HRB Financial Statements and other material financial records (including those delivered or made available to any Governmental Entity) in accordance with GAAP or regulatory accounting principles, as applicable, and all applicable legal and regulatory requirements. Since January 1, 2009, HRB Bank's and Block Financial's external auditor was independent of HRB Bank and Block Financial and their management. As of the date hereof, HRB Bank's and Block Financial's external auditor has not resigned or been dismissed as a result of or in connection with any disagreements with HRB Bank or Block Financial on a matter of accounting principles or practices, financial statement disclosure or auditing scope or procedure.

**4.06.** Undisclosed Liabilities. Neither HRB Bank nor Block Financial has any material Liability that is not reflected in or disclosed in the HRB Financial Statements.

**4.07.** Litigation. Except for Litigation listed in Section 4.07 of the HRB Bank Disclosure Schedule (the "<u>Pending</u> <u>Litigation</u>"), there is no Litigation pending or, to HRB Bank's Knowledge, threatened against either of HRB Bank or Block Financial at law or in equity, or by or before any Governmental Entity (a) relating to the Transferred Assets or the Assumed Liabilities or (b) that challenges the validity of this Agreement or the transactions contemplated hereby, that challenges any actions taken or to be taken by HRB Bank or Block Financial pursuant hereto or thereto or that seeks to enjoin or otherwise restrain the transactions contemplated hereby or thereby. There is no material Litigation now pending in which HRB Bank or Block Financial is the plaintiff or claimant which relate to, or would reasonably be expected to materially affect BofI's ownership, operation or use in its business of, the Transferred Assets or the Assumed Liabilities.

**4.08. Consents and Approvals**. No Consent or Order of any Governmental Entity or other Person is required on the part of HRB Bank or Block Financial in connection with the execution, delivery or performance of this Agreement by HRB Bank or Block Financial, or any agreement to be executed by HRB Bank or Block Financial contemplated hereby, or the completion by HRB Bank or Block Financial of the transactions contemplated hereby or thereby, and to HRB Bank's Knowledge, no such Governmental Entity or other Person has indicated that it will withhold such Consent or Order.

**4.09. Ownership of Assets**. HRB Bank has good and transferable title to, or in the case of the Transferred Assets described in Sections 2.01(f), a valid license or right to use, the Transferred Assets, free and clear of all Liens.

**4.10.** Absence of Certain Changes or Events. Since April 30, 2013, HRB Bank has conducted its business only in the ordinary course and has not:

(a) Instituted, had instituted against it, settled or agreed to settle any Litigation relating to the Transferred Assets or Assumed Liabilities other than in the ordinary course of business;

(b) Made any, or acquiesced with any, material change in any accounting methods, principles or practices applicable to the Transferred Assets or Assumed Liabilities, except as required by applicable Law;

(c) Changed its interest rate or fee pricing policies or other material terms with respect to its Deposits, Loans or Commitments other than in the ordinary course of business;

(d) Waived any material right related to the Transferred Assets or the Assumed Liabilities;

(e) Amended, terminated, waived, assigned or modified the terms of any Loan, Deposit or Commitment except in the ordinary course of business; or

(f) Entered into any agreement or made any commitment whether in writing or otherwise to take any of the types of action described in subsections (a) through (e) above.

#### 4.11. Contracts and Commitments.

(a) HRB Bank is not a party to or bound by any Contract for the purchase or sale of any of the Transferred Assets or the assumption of the Assumed Liabilities, or for the grant of any preferential right to purchase any of the Transferred Assets or assume any of the Assumed Liabilities.

(b) Each of the Contracts identified in Section 4.11(b) of the HRB Bank Disclosure Schedule (the "<u>Assumed</u> <u>Contracts</u>") and each of the Commitments constitutes and, on the Closing Date, each of the Assumed Contracts and the Commitments will constitute, the legal, valid and binding obligation of HRB Bank, and each of the other parties thereto. Each of the Assumed Contracts and Commitments is, and, on the Closing Date, will be, in full force and effect (except to the extent that any Assumed Contract or Commitment expires in accordance with its terms). HRB Bank has fulfilled and performed in all material respects its obligations under the Assumed Contracts and Commitments is, or is alleged to be, in breach or default under any Assumed Contract or Commitment, nor does there exist any condition which with the passage of time or the giving of notice or both would result in a breach or default thereunder. HRB Bank has made available to BofI true and complete copies of each Assumed Contract and of the terms applicable to each Commitment.

(c) Section 4.11(c) of the HRB Bank Disclosure Schedule identifies each material Contract pursuant to which HRB Bank obtains services from a third party material to the administration or management of the Transferred Assets or the Assumed Liabilities.

#### 4.12. Taxes and Tax Returns

(a) HRB Bank has duly and timely filed all Tax Returns that it was required to file under applicable Law with the appropriate Federal, state, local or foreign governmental agencies. All such Tax Returns were correct and complete in all material respects and have been prepared in substantial compliance with all applicable Law. All Taxes due and owing by HRB Bank (whether or not shown on any Tax Return) have been paid. There are no Liens for Taxes (other than Taxes not yet due and payable) upon any of the assets of HRB Bank.

(b) HRB Bank has withheld and paid all Taxes required to have been withheld and paid in connection with any amounts paid or owing to any employee, independent contractor, creditor, depositor, shareholder, or other third party, and all information reporting (including Forms W-2, 1098 and 1099) required with respect thereto have been properly completed and timely filed. HRB Bank has timely and properly taken such actions in response to and in compliance with notices

from the Internal Revenue Service in respect of information reporting and backup and nonresident withholding as are required by Law, including the notation in its records of any B notices or C notices received with respect to any customers, shareholders, or payees.

(c) There is no dispute or claim concerning any Tax liability of HRB Bank either (i) claimed or raised by any authority in writing, or (ii) as to which any director or officer (or employee responsible for Tax matters) of HRB Bank has knowledge based upon personal contact with any agent of such authority. No claim has been made in writing by any Governmental Entity in a jurisdiction where HRB Bank does not file Tax Returns that HRB Bank is, or may be, subject to taxation by that jurisdiction.

(d) HRB Bank is not a party to or bound by any tax allocation or sharing agreement.

**4.13.** No Material Adverse Change. There has not been any Material Adverse Change with respect to HRB Bank or Block Financial since April 30, 2013, nor has any event or condition occurred that has resulted in, or would be reasonably likely to result in, a Material Adverse Change with respect to HRB Bank or Block Financial.

**4.14.** Evidences of Indebtedness. All of the Transferred Loans and Overdrafts are legal, valid and binding obligations of the respective obligors thereof, enforceable in accordance with their respective terms, and are not subject to any known or, to HRB Bank's Knowledge, any threatened, defenses, offsets or counterclaims that may be asserted against HRB Bank or any subsequent holder thereof; <u>provided</u>, <u>however</u> that the enforceability of the Loans may be affected by bankruptcy, reorganization, insolvency and similar laws of general application relating to or affecting the rights or remedies of creditors generally and by equitable principles that may be applied by a court in construing or enforcing such Loans. Any collateral securing a Transferred Loan is (i) the collateral described in the applicable security agreement, mortgage, pledge, collateral assignment or other security document and (ii) subject to a valid, enforceable and perfected Lien.

**4.15. Regulatory Compliance**. HRB Bank is not subject to any Order of any Governmental Entity other than Orders broadly applicable to similarly situated financial institutions. Other than routine regulatory examinations, there are no actions or proceedings pending or, to HRB Bank's Knowledge, threatened against HRB Bank or affecting or relating to in any material manner the Transferred Assets or the Assumed Liabilities by or before any Governmental Entity. HRB Bank has not received notice from any Governmental Entity indicating that it would oppose or not grant or issue its Consent, if required, with respect to the transactions contemplated by this Agreement.

**4.16. Books and Records**. The Records (a) have been accurately maintained in all material respects in the ordinary course of business, (b) are substantially complete and correct in all material respects, (c) have only entered transactions therein that represent bona fide transactions, (d) do not fail to reflect transactions involving the business of HRB Bank that properly should have been set forth therein and that have not been accurately so set forth, (e) comply with all applicable Laws and (f) are sufficient to allow BofI to manage the Transferred Assets and Assumed Liabilities in the same manner as they have been managed by HRB Bank and Block Financial.

**4.17. Financial Products and Services**. Each of the financial products and services that are a part of the Transferred Assets or Assumed Liabilities is, and at all times has been, offered and sold in compliance in all material respects with all applicable Laws.

**4.18.** Interest Rate Risk Management Instruments. HRB Bank has no interest rate swaps, caps, floors, option agreements or other interest rate risk management arrangements, entered into for the account of HRB Bank.

**4.19. Participations**. Each outstanding Emerald Advance and Unsecured Credit Card Receivable originated by HRB Bank has been, or will be prior to the Closing, fully participated to an H&R Block Affiliate in a participation arrangement that meets GAAP requirements for de-recognition of assets.

**4.20.** Finder's Fee. No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with this Agreement or any of the transactions contemplated hereby based upon arrangements made by or on behalf of HRB Bank, or any officer, member, director or employee of HRB Bank, or any Affiliate of HRB Bank.

**4.21.** Custodial Accounts. HRB Bank has no Custodial Accounts other than individual retirement accounts. HRB Bank has managed the Custodial Accounts included in the Transferred Assets at all times in compliance in all material respects with Law and the governing documents of such Custodial Accounts.

**4.22. Trust Accounts**. HRB Bank has no trust accounts, trust assets or any other accounts or relationships of a fiduciary nature.

**4.23. Deposits**. Subject to applicable Law, all of the Deposits (other than Time Deposits) may be re-priced by HRB Bank in the ordinary course of business. All of the Deposits (including the Records pertaining to such Deposits) have been established and are held in compliance in all material respects with (i) all applicable policies, practices and procedures of HRB Bank, and (ii) all applicable Laws, including anti-money laundering, anti-terrorism, or embargoed persons requirements. All of the Deposit accounts are insured to the maximum limit set by the FDIC and any premiums and assessments required to be paid in connection therewith have been fully paid, and no proceedings for the termination or revocation of such insurance are pending, or to HRB Bank's Knowledge threatened.

**4.24.** Customer Information Security. To HRB Bank's Knowledge no facts or circumstances exist that would cause HRB Bank to be deemed not to be in satisfactory compliance in any material respect with the applicable privacy of customer information requirements contained in any applicable federal and state privacy laws and regulations, including, without limitation, in Title V of the Gramm-Leach-Bliley Act of 1999 and regulations promulgated thereunder, as well as the provisions of any information security program adopted by the Bank pursuant to 12 C.F.R. Part 170.

# 4.25. Prior Purchase and Assumption Agreement.

(a) A true and complete copy, excluding the schedules, of the Purchase and Assumption Agreement, dated as of July 11, 2013 (the "Prior P&A Agreement"), among Republic Bank & Trust Company, a Kentucky bank and trust company, HRB Bank and Block Financial was

included as Exhibit 10.1 to H&R Block's form 8-k, filed with the Securities and Exchange Commission on July 11, 2013.

(b) HRB Bank and Block Financial have validly terminated the Prior P&A Agreement prior to entering into this Agreement.

**4.26.** Available Funds. Block Financial has, and as of the Closing Date will have, access to sufficient funds to make any required Block Financial Capital Contribution.

**4.27. Disclosure**. No representation or warranty by HRB Bank or Block Financial in this Agreement, nor any statement, certificate, schedule or exhibit hereto furnished or to be furnished by or on behalf of HRB Bank or Block Financial pursuant to this Agreement or in connection with transactions contemplated hereby, contains or will contain any untrue statement of material fact or omits or will omit a material fact necessary to make the statements contained therein not misleading.

### ARTICLE V REPRESENTATIONS AND WARRANTIES OF BOFI

As a material inducement to HRB Bank and Block Financial to enter into and perform its obligations under this Agreement, except as disclosed in the Disclosure Schedule (the "<u>Bofl Disclosure Schedule</u>") delivered by Bofl to HRB Bank and Block Financial before the execution of this Agreement, Bofl hereby represents and warrants to HRB Bank and Block Financial that the statements contained in this Article V are true and correct as of the date hereof (unless they are made as of a specific date, in which case they shall be true and correct as of such specific date) and will be true and correct as of the Closing (unless they are made as of a specific date, in which case they shall be true and correct as of such specific date):

**5.01.** Organization and Authority. BofI is a federal savings bank duly organized and validly existing under the Laws of the United States. BofI has all requisite corporate power and authority to carry on its business as now being conducted, to own, lease and operate its properties and assets as now owned, leased or operated and to enter into and to carry on the business and activities now conducted by it. True and complete copies of all Constituent Documents of BofI, as amended to date, have been made available to HRB Bank. BofI is an insured bank as defined in the FDIA.

**5.02.** Execution and Delivery. BofI has the corporate power and authority to execute, deliver and perform this Agreement, the Ancillary Agreements to which it is a party and any documents, agreements or instruments to be executed by BofI pursuant to this Agreement or any Ancillary Agreement, and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and the Ancillary Agreements to which it is a party by BofI and the consummation by BofI of the transactions contemplated hereby have been duly authorized by the Board of Directors of BofI, and no other corporate proceedings are necessary to authorize this Agreement and the P&A Transaction. This Agreement has been, and the Ancillary Agreements to which it is a party, and the other agreements and documents contemplated hereby and thereby, have been or at Closing will be, duly executed by BofI, and each constitutes or will, assuming the due execution thereof by each of the other parties thereto, constitute the legal, valid and binding obligation of BofI, enforceable in accordance with its respective terms and conditions, except as enforceability may be limited by bankruptcy, insolvency, fraudulent conveyance,

reorganization, receivership, moratorium or similar Laws affecting creditors' rights and remedies and by general principles of equity, including principles of commercial reasonableness, good faith and fair dealing, regardless of whether enforcement is sought in a proceeding at law or in equity.

## 5.03. Compliance with Laws, Permits and Instruments.

(a) BofI has performed and abided by all obligations required to be performed by it to the date hereof, and has complied with, and is in compliance with, and is not in default (and with the giving of notice or the passage of time will not be in default) under, or in violation of, (i) any provision of its Constituent Documents, (ii) any provision of any Contract, except where nonperformance, noncompliance, default or violation would not reasonably be expected to result in a Material Adverse Change, or (iii) any Law, Order or Permit applicable to BofI, except where nonperformance, noncompliance, default or violation where nonperformance, noncompliance, default or violation would not reasonably be expected to result in a Material Adverse Change.

(b) The execution, delivery and (provided the Required Regulatory Approvals are obtained) performance of this Agreement and the other agreements contemplated hereby, and the completion of the transactions contemplated hereby and thereby will not conflict in any material respect with, result in any violation or breach of or result in a material default, under (i) BofI's Constituent Documents (ii) any provision of any Contract, applicable to BofI's activities, except where nonperformance, noncompliance, default or violation would not reasonably be expected to result in a Material Adverse Change, or (iii) any Law, Order or Permit applicable to BofI.

**5.04.** Consents and Approvals. No Consent or Order of any Governmental Entity or other Person is required on the part of BofI in connection with the execution, delivery or performance of this Agreement by BofI, or any agreement to be executed by BofI contemplated hereby, or the completion by BofI of the transactions contemplated hereby or thereby, and to BofI's Knowledge, no such Governmental Entity or other Person has indicated that it will withhold such Consent or Order.

**5.05. Financial Statements**. BofI has made available to HRB Bank true and complete copies of the BofI Holding financial statements identified in Section 5.05 of the BofI Disclosure Schedule (the "<u>BofI Financial Statements</u>"). The BofI Financial Statements (including the related notice) complied as to form, as of their respective dates, in all material respects with applicable accounting requirements, have been prepared in accordance with GAAP, fairly present the financial condition of BofI at the dates thereof and the results of operations for the periods then ended (subject, in the case of unaudited financial statements, to notes and normal year-end audit adjustments that were not material in amount or effect), and the accounting records underlying the BofI Financial Statements accurately and fairly reflect in all material respects the transactions of BofI. Since January 1, 2009, BofI's and BofI Holding's external auditor was independent of BofI and BofI Holdings and their management. As of the date hereof, BofI's and BofI Holding's external auditor has not resigned or been dismissed as a result of or in connection with any disagreements with BofI or BofI Holdings on a matter of accounting principles or practices, financial statement disclosure or auditing scope or procedure.

**5.06.** Undisclosed Liabilities. BofI does not have any material Liability that is not reflected in or disclosed in the BofI Financial Statements.

**5.07.** Litigation. There is no Litigation pending or, to BofI's Knowledge, threatened against BofI at law or in equity, or by or before any Governmental Entity (a) relating to the Transferred Assets or the Assumed Liabilities or (b) that challenges the validity of this Agreement or the transactions contemplated hereby, that challenges any actions taken or to be taken by BofI pursuant hereto or thereto or that seeks to enjoin or otherwise restrain the transactions contemplated hereby or thereby.

**5.08. Regulatory Compliance**. BofI is not subject to any Order of any Governmental Entity other than Orders broadly applicable to similarly situated financial institutions. Other than routine regulatory examinations, there are no actions or proceedings pending or, to BofI's Knowledge, threatened against BofI by or before any Governmental Entity. BofI has not received notice from any Governmental Entity indicating that it would oppose or not grant or issue its Consent, if required, with respect to the transactions contemplated by this Agreement.

**5.09.** Finder's Fee. No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with this Agreement or any of the transactions contemplated hereby based upon arrangements made by or on behalf of BofI, or any officer, member, director or employee of BofI, or any Affiliate of BofI.

**5.10.** Association Rights. BofI is a member in good standing of MasterCard and has full authority under applicable MasterCard operating regulations to issue Emerald Cards and to use and display MasterCard trademarks.

**5.11.** Nevada Branch. BofI has filed all regulatory notices and applications required under 12 C.F.R. § 145.93 or otherwise to open and operate a new full service branch in Reno, Nevada (the "<u>Nevada Branch</u>").

**5.12. Disclosure**. No representation or warranty by BofI in this Agreement, nor any statement, certificate, schedule or exhibit hereto furnished or to be furnished by or on behalf of BofI pursuant to this Agreement or in connection with transactions contemplated hereby, contains or will contain any untrue statement of material fact or omits or will omit a material fact necessary to make the statements contained therein not misleading.

### ARTICLE VI COVENANTS

**6.01.** Conduct of the Business in the Ordinary Course. Except as otherwise expressly contemplated under this Agreement or disclosed in <u>Schedule 6.01</u>, between the date of this Agreement and the Closing, HRB Bank will conduct its business only in the ordinary course of business consistent with past custom and practice with respect to the Transferred Assets and Assumed Liabilities, and shall incur no Assumed Liabilities other than in the ordinary course of business consistent with past custom and practice.

**6.02.** Required Acts. In furtherance of and without limiting Section 6.01, except as otherwise expressly contemplated under this Agreement, between the date of this Agreement and the Closing, HRB Bank will:

(a) Timely file all Tax Returns required to be filed by it and promptly pay all Taxes that become due and payable, except those being contested in good faith by appropriate proceedings, with adequate reserves established;

(b) Promptly classify and charge off Loans and make appropriate adjustments to loss reserves in accordance with the Call Report Instructions, applicable regulatory definitions and the Uniform Retail Credit Classification and Account Management Policy;

(c) Maintain its books of account in accordance with GAAP or regulatory accounting principles (as applicable);

(d) Promptly inform BofI of all material issues and developments arising in relation to the Transferred Assets and the Assumed Liabilities; and

(e) Promptly take all reasonable actions to wind down all Brokered Deposits.

# 6.03. Prohibited Acts.

(a) **HRB Bank**. In furtherance of and without limiting Section 6.01, except as specifically permitted under this Agreement, between the date of this Agreement and Closing, HRB Bank will not, without the prior written consent of BofI:

- i. Amend, terminate, waive, assign or modify any of the terms of any Assumed Contract or Commitment, except in the ordinary course of business and only to the extent not adverse to HRB Bank or Bofl;
- ii. Amend, terminate, waive, transfer, assign or modify any of the terms of any Consumer Loan or Deposit, except in the ordinary course of business and only to the extent not materially adverse to HRB Bank or BofI;
- iii. Amend, waive or modify HRB Bank's Consumer Loan and Deposit underwriting, credit servicing, collection and operating policies and procedures, except in response to changes in Applicable Laws and except in the ordinary course of business and only to the extent not materially adverse to HRB Bank or BofI;
- iv. Offer rates or other material terms on any category of Loans, Overdraft or Deposits in a manner inconsistent with HRB Bank's past practices;
- v. Sell, assign or transfer any Transferred Asset or Deposit to any other Person or, except as disclosed in the HRB Bank Disclosure Schedule, encumber or permit or suffer to exist any Lien on any Transferred Asset or Deposit;
- vi. Take any action that could reasonably be expected to complicate, hinder or delay any Required Regulatory Approval; or
- vii. Agree to do any of the foregoing.

(b) **BofI**. Except as specifically permitted under this Agreement, between the date of this Agreement and Closing, BofI will not, without the prior written consent of HRB Bank,

take any action that could reasonably be expected to complicate, hinder or delay any Required Regulatory Approval.

#### 6.04. Regulatory Approvals.

i.

(a) Within seven (7) Business Days following the date of this Agreement, each of BofI, HRB Bank and Block Financial will (i) file all applications necessary to obtain the Required Regulatory Approvals with the Approving Authorities and (ii) publish all required public notices required in connection with the applications filed to obtain the Required Regulatory Approvals.

(b) Each party will promptly furnish to the requesting party all information, data and documents required to be included in any regulatory application to be filed with the Approving Authorities; except to the extent that such information would be, or relates to information that would be, filed under a claim of confidentiality. Each party agrees to cooperate and join in with the other party in the preparation, execution and processing of all such applications. The party responsible for filing a regulatory application shall promptly deliver to the other party evidence of the filing of such application and a copy of the non-confidential portions of such application.

(c) The parties shall promptly advise each other upon receiving any communication from any Approving Authorities that would reasonably cause such party to believe that there is a likelihood that the Required Regulatory Approvals or any other consent or approval required hereunder will not be obtained or that the receipt of any such approval will be materially delayed.

(d) Each party will use its commercially reasonable best efforts to pursue and obtain the Required Regulatory Approvals and will promptly respond to all inquiries from the Approving Authorities. To the extent permitted by applicable law, each party shall promptly deliver to the other party a copy of each material Order and other correspondence received by such party from any Approving Authority or other Governmental Entity with respect to the Divestiture Transaction or any other transaction or agreement contemplated by this Agreement.

(e) Notwithstanding the foregoing or any other provision of this Agreement to the contrary, in no event shall any party hereto be obligated to, in connection with seeking Required Regulatory Approvals:

(A) agree to, or proffer to, divest or hold separate, or enter into any licensing (except those licensing agreements as may be required or contemplated by the Program Management Agreement) or similar arrangement with respect to, any material assets (whether tangible or intangible) or any material portion of any business of such party or its Affiliates, or (B) take any action, or commit to take any action, or agree to any condition or restriction (in each case including with respect to the terms of this Agreement or the Program Management Agreement), involving any party or any of such party's Affiliates or subsidiaries which, in the case of (A) or (B), such party determines, in its reasonable good faith judgment, is or is reasonably likely to be materially and unreasonably burdensome on such party's or its Affiliates' or subsidiaries' business following the Closing or would be reasonably likely to materially reduce the economic benefits of



the transactions contemplated by this Agreement or the Program Management Agreement to such party to such a degree that such party would not have entered into this Agreement had such condition or restriction been known to it at the date hereof (a "<u>Materially Burdensome Regulatory Condition</u>"); or

ii. litigate or contest any Litigation, whether judicial or administrative, (A) challenging or seeking to restrain or prohibit the consummation of the transactions contemplated by this Agreement, (B) seeking to prohibit or limit in any respect the ownership or operation by the party of any material asset (whether tangible or intangible) or any material portion of its business or to require the party to divest or hold separate, or enter into any licensing or similar arrangement with respect to, any asset (whether tangible) or any portion of its business as a result of the transactions contemplated by this Agreement, (C) seeking to impose limitations on the ability of BofI to acquire or hold, or exercise full rights of ownership of, any of the Transferred Assets or Assumed Liabilities, or (D) seeking to prohibit BofI from effectively controlling in any respect any of the Transferred Assets or Assumed Liabilities.

**6.05.** Access to Information and Properties. Between the date of this Agreement and the Closing Date, HRB Bank shall permit any officers, employees, representatives or agents of BofI, at BofI's cost and expense, access at all reasonable times to HRB Bank's assets and personnel, and disclose and make available to BofI and its officers, employees, representatives or agents, all books, papers and records relating to the Transferred Assets and the Assumed Liabilities as BofI may reasonably request. HRB Bank shall cause its personnel to provide reasonable assistance to BofI in BofI's investigation of matters relating to the Transferred Assets and the Assumed Liabilities; *provided* such assistance does not unreasonably interfere with such personnel's job duties. BofI will not exercise its rights under this section in a manner that materially disrupts the normal business operations of HRB Bank. Any information disclosed, or made available, between the parties hereto will at all times be subject to the Nondisclosure Agreement.

**6.06.** Agreements and Consents. Each of the parties hereto will use commercially reasonable efforts to take, or cause to be taken, all actions, and to do, or cause to be done, all things necessary, proper, or advisable under applicable Laws to consummate and make effective the transactions contemplated by this Agreement as expeditiously as possible. HRB Bank will act diligently and reasonably in attempting to obtain, before the Closing Date, the Consents, in form and substance reasonably satisfactory to BofI, from any party to any Contract required to be obtained to assign or transfer any such Contract to BofI or interests therein.

**6.07.** Fees and Expenses. Except as otherwise set forth in this Agreement, HRB Bank and Block Financial will pay their own, and BofI will pay its own, out-of-pocket expenses in connection with this Agreement, including appraisal, accounting, professional and legal fees, if any, whether or not the transactions contemplated by this Agreement are consummated.

**6.08.** Employees. Between the date of this Agreement and the Closing Date, BofI will not hire any employee of HRB Bank and HRB Bank will not hire any employee of BofI.

#### 6.09. Notification. Prior to the Closing Date:

(a) HRB Bank and Block Financial shall promptly provide BofI (after HRB Bank or Block Financial has notice thereof) with written notice of, and keep BofI advised as to, (a) any Material Adverse Change in the Transferred Assets or the Assumed Liabilities and (b) any pending or, to HRB Bank's Knowledge, threatened or anticipated Litigation that challenges any part of the Divestiture Transaction.

(b) BofI shall promptly provide HRB Bank and Block Financial (after BofI has notice thereof) with written notice of, and keep HRB Bank advised as to any pending or, to BofI's Knowledge, threatened or anticipated Litigation that challenges any part of the Divestiture6. Transaction.

**6.10. BINs**. The BINs shall be transferred to BofI on a mutually agreed date on or about the Closing Date. With respect to the BINs transfer, each party shall use commercially reasonable efforts to assist in the transfer of the BINs to BofI, including providing any notices to, or obtaining any consents from, MasterCard.

**6.11.** Exclusive Dealing. During the period beginning on the date of this Agreement and continuing through the earlier of the Closing Date or the termination of this Agreement, (a) neither HRB Bank nor Block Financial will, directly or indirectly, through any representative or otherwise, solicit or entertain offers from, negotiate with or accept, any proposal of any other person relating to any of the Transferred Assets, the P&A Transaction or the transactions contemplated by the Program Management Agreement and the Receivables Participation Agreement (or any similar transaction) or the assets, liabilities or business, in whole or in part, of HRB Bank (any such offer or proposal, a "Competing Proposal"), and (b) BofI will not negotiate or enter into any arrangement or transaction similar to those contemplated by the Program Management Agreement Agreement with any other tax return preparer. HRB Bank, Block Financial and their respective Affiliates shall immediately cease and cause to be terminated any existing discussions or negotiations with any Persons conducted heretofore with respect to any Competing Proposal.

6.12. Nevada Branch Opening. BofI shall complete the opening of its Nevada Branch by the Closing Date.

**6.13.** Execution and Delivery of Program Contracts. At the Closing on the Closing Date, BofI, HRB Bank and Block Financial will execute and deliver (and Block Financial will cause its Affiliates to execute and deliver) each of the following program agreements (collectively, the "Program Agreements"):

- (a) Program Management Agreement, substantially in the form attached hereto as <u>Exhibit G</u>;
- (b) Guaranty Agreement, substantially in the form attached hereto as <u>Exhibit H</u>;
- (c) Emerald Advance Receivables Participation Agreement, substantially in the form attached hereto as Exhibit I;
- (d) Company Financial Products Distribution Agreement, substantially in the form attached hereto as <u>Exhibit J</u>;



(e) Franchisee Financial Products Distribution Agreement, substantially in the form attached hereto as <u>Exhibit K</u>; provided, however, that the franchisees will not sign the Franchisee Financial Products Distribution Agreement on the Closing Date;

- (f) Trademark Licensing Agreement, substantially in the form attached hereto as <u>Exhibit L</u>;
- (g) Trademark Licensing Agreement, substantially in the form attached hereto as <u>Exhibit M</u>;
- (h) Collection Services Agreement, substantially in the form attached hereto as Exhibit N;

(i) Amended and Restated Legacy Emerald Advance Participation Agreement, substantially in the form attached hereto as Exhibit O; and

(j) Amended and Restated Legacy Emerald Unsecured Credit Card Participation Agreement, substantially in the form attached hereto as Exhibit P.

**6.14.** Further Assurances. From time to time after the Closing, each party, without further consideration, shall execute and deliver, or cause to be executed and delivered, such instruments and take such action as may be reasonably necessary or requested by the other party in order to carry out the transactions contemplated by this Agreement. Without limiting the foregoing, on and after the Closing Date, HRB Bank and Block Financial shall (a) give such further assistance to BofI and shall execute, acknowledge, and deliver all such instruments and take such further action as may be reasonably necessary and appropriate or reasonably requested by BofI effectively to transfer to, and vest in, BofI full, legal, and equitable title to the Transferred Assets and Assumed Liabilities and put BofI in possession of the Transferred Assets, (b) in the case of Contracts and other interests included in the Transferred Assets (including any Non-Assigned Assets) (i) which cannot be transferred or assigned effectively without the Consent of third parties which Consent has not been obtained prior to the Closing, exercise commercially reasonable efforts, and cooperate with BofI in exercising commercially reasonable efforts, to obtain such Consent promptly, and if any such Consent is unobtainable, to use its commercially reasonable efforts to secure to BofI the benefits thereof in some other manner or (ii) which are otherwise not transferable or assignable, to use its commercially reasonable efforts jointly with BofI to secure to BofI the benefits and burdens thereof in some other manner (including the exercise of the rights of HRB Bank thereunder), in the case of each of clause (i) and (ii) immediately above to the extent requested by, and at the option of, BofI; provided, however, that nothing herein shall relieve HRB Bank or Block Financial of their obligations under any other section of this Agreement, and (c) use reasonable efforts to assist BofI in the orderly transfer of the Transferred Assets and Deposits.

# ARTICLE VII POST-CLOSING

**7.01.** Affiliate Merger. Subject to regulatory approval, promptly following the Closing, HRB Bank will merge with and into Block Financial. Pursuant to the Affiliate Merger, Block Financial will succeed to all of the remaining assets and liabilities of HRB Bank. As a result of the Affiliate Merger, HRB Bank will surrender its bank charter and cease to exist as a separate legal entity.

**7.02.** Filings; Power of Attorney. BofI may file with Governmental Entities, on behalf and in the name of HRB Bank, all assignments related to the Loans and any financing statement amendments and any other documents covered by the Power of Attorney to be executed by HRB Bank on or after the Closing Date.

**7.03.** Overdrafts True-Ups. Block Financial agrees to indemnify BofI for the aggregate amount of all Overdrafts purchased by BofI that are not covered by subsequent Customer deposits or loads to HRB Prepaid Cards within sixty (60) days after the Closing Date. For a period of three (3) years after the Closing Date, BofI agrees to reimburse Block Financial, on a quarterly basis, for each Overdraft purchased by BofI with respect to which (i) Block Financial indemnified BofI pursuant to the preceding sentence and (ii) BofI thereafter obtains a recovery, either through a subsequent Customer deposit, load to an HRB Prepaid Card or otherwise, to the extent of such recovery.

# ARTICLE VIII CONDITIONS

**8.01.** Conditions to Obligations of HRB Bank and Block Financial. The obligation of HRB Bank and Block Financial to effect the P&A Transaction is subject to the fulfillment or HRB Bank's and Block Financial's written waiver at or prior to the Closing Date of all of the following additional conditions:

(a) **Regulatory** Approvals. All Required Regulatory Approvals have been obtained and remain in full force and effect and all applicable waiting periods have expired, and no such Regulatory Approval shall contain or shall have resulted in, or would reasonably be expected to result in, the imposition of any Materially Burdensome Regulatory Condition.

(b) **Representations and Warranties**. The representations and warranties of BofI set forth in this Agreement are true and correct in all material respects as of the date of this Agreement and as of the Closing Date (as though made on and as of the Closing Date except (i) to the extent such representations and warranties are by their express provisions made as of a specific date, and (ii) for the effect of changes or transactions permitted or contemplated by this Agreement).

(c) **Performance of Obligations**. BofI has performed all obligations, covenants and agreements required to be performed by it under this Agreement, in all material respects, prior to the Closing Date, including the delivery of the documents described in Section 8.01(j).

(d) *No Adverse Action*. No Law or Order shall have been promulgated, enacted, entered, enforced or deemed applicable to this Agreement or the transactions contemplated hereby by any Governmental Entity that would (i) make this Agreement or any other agreement contemplated hereby, or any part of the Divestiture Transaction, illegal, invalid or unenforceable, or (ii) impose material limits in the ability of any party to this Agreement to complete this Agreement or any other agreement contemplated hereby, or any part of the Divestiture Transaction.

(e) *Execution of Program Agreements*. BofI has executed and delivered the Program Agreements pursuant to Section 6.13.

(f) *No Material Adverse Change*. Since the date of this Agreement, there has been no Material Adverse Change with respect to BofI.

(g) *Estimated Transfer Amount*. BofI shall have tendered the Estimated Transfer Amount (if it is a positive number).

# (h) *Third Party Consents*. The parties shall have obtained the third party consents identified on <u>Schedule</u>

<u>8.01(h)</u>.

(i)

Nevada Branch. BofI shall have opened the Nevada Branch.

(j) *Other Deliverables*. BofI has delivered to HRB Bank and Block Financial all of the following agreements, certifications and other deliverables:

- i. an executed Assignment and Assumption of Deposit Liabilities Agreement, in substantially the form set forth in Exhibit A;
- ii. an executed Assignment and Assumption of Assumed Contracts Agreement, in substantially the form set forth in <u>Exhibit B;</u>
- iii. an executed Bill of Sale, in substantially the form of <u>Exhibit C</u>.
- iv. an executed Assignment and Assumption of Loans Agreement, in substantially the form set forth in Exhibit D;
- v. an executed Assignment, Transfer and Appointment of Successor Custodian for Custodial Accounts, in substantially the form set forth in Exhibit E;
- vi. a certified copy of the resolutions of the Board of Directors of BofI authorizing the execution of this Agreement and the consummation of the P&A Transaction;
- vii. a Certificate or Certificates signed by an authorized officer of BofI stating that all of the conditions set forth in Sections 8.01(b) and (c) have been satisfied or waived, as provided therein; and
- viii. such other documents, instruments, certificates and other agreements as HRB Bank or Block Financial may reasonably require to effect the transactions contemplated by this Agreement to be consummated as of the Closing.

**8.02.** Conditions to Obligations of BofI. The obligation of BofI to effect the P&A Transaction is subject to the fulfillment or BofI's written waiver at or prior to the Closing Date of all of the following additional conditions:

(a) **Regulatory Approvals**. All Required Regulatory Approvals have been obtained and remain in full force and effect and all applicable waiting periods have expired, and no such Regulatory Approval shall contain or shall have resulted in, or would reasonably be expected to result in, the imposition of any Materially Burdensome Regulatory Condition.

(b) **Representations and Warranties**. The representations and warranties of HRB Bank and Block Financial set forth in this Agreement are true and correct in all material respects as of the date of this Agreement and as of the Closing Date (as though made on and as of the Closing Date except (i) to the extent such representations and warranties are by their express

provisions made as of a specific date, and (ii) for the effect of changes or transactions permitted or contemplated by this Agreement).

(c) **Performance of Obligations**. Each of HRB Bank and Block Financial has performed all obligations, covenants and agreements required to be performed by it under this Agreement, in all material respects, prior to the Closing Date, including the delivery of the documents described in Section 8.02(i).

(d) *No Adverse Action*. No Law or Order shall have been promulgated, enacted, entered, enforced or deemed applicable to this Agreement, or the transactions contemplated hereby by any Governmental Entity that would (i) make this Agreement or any other agreement contemplated hereby, or the P&A Transaction, illegal, invalid or unenforceable, or (ii) impose material limits in the ability of any party to this Agreement to complete this Agreement or any other agreement contemplated hereby, or the P&A Transaction.

(e) *Execution of Program Agreements*. EFS and Block Financial (and their Affiliates) have executed and delivered the Program Agreements pursuant to Section 6.13.

(f) *No Material Adverse Change*. Since the date of this Agreement, there has been no Material Adverse Change with respect to HRB Bank or Block Financial.

(g) *Transfer Amount*. HRB Bank shall have tendered the Estimated Transfer Amount (if it is a negative number).

(h) *Third Party Consents*. The parties shall have obtained the third party consents identified on <u>Schedule</u> <u>8.01(h)</u>.

(i) *Other Deliverables*. Each of HRB Bank and Block Financial has delivered, or caused to be delivered, to BofI all of the following agreements, certifications and other deliverables:

- i. an executed Assignment and Assumption of Deposit Liabilities Agreement, in substantially the form set forth in Exhibit A;
- ii. an executed Assignment and Assumption of Assumed Contracts Agreement, in substantially the form set forth in <u>Exhibit B;</u>
- iii. an executed Bill of Sale, in substantially the form set forth in Exhibit C;
- iv. an executed Assignment and Assumption of Loans Agreement, in substantially the form set forth in Exhibit D;
- v. an executed Assignment, Transfer and Appointment of Successor Custodian for Custodial Accounts, in substantially the form set forth in Exhibit E;
- vi. an executed Limited Power of Attorney, in substantially the form set forth in Exhibit F;
- vii. a certified copy of the resolutions of the Board of Directors and Shareholder of HRB Bank authorizing the execution of this Agreement and the consummation of the P&A Transaction;

- viii. a Certificate or Certificates signed by an authorized officer of HRB Bank stating that all of the conditions set forth in Sections 8.02(b) and (c) have been satisfied or waived, as provided therein;
- ix. such other bills of sale, assignments, and other instruments and documents as counsel for BofI may reasonably require as necessary or desirable for transferring, assigning and conveying to BofI good, marketable and insurable title to the Transferred Assets;
- x. the Records; and
- xi. such other documents, instruments, certificates and other agreements as BofI may reasonably require to effect the transactions contemplated by this Agreement to be consummated as of the Closing.

#### ARTICLE IX TERMINATION, AMENDMENT, AND WAIVER

9.01. Termination. This Agreement may be terminated at any time prior to the Closing Date:

(a) By written consent of all parties hereto;

(b) By any party hereto at any time after September 30, 2014 if the Closing has not occurred on or prior to that date; provided, however, that the rights to terminate this Agreement under this Section 9.01(b) are not available to any party whose action or failure to act has been the principal cause of, or resulted in, the failure of the Closing to occur on or before such date and such action or failure to act constitutes a breach of this Agreement;

(c) By HRB Bank or Block Financial, if (i) any Approving Authority has (A) denied approval of any regulatory application filed by BofI in connection with the P&A Transaction; or (B) indicated to BofI that no final decision on a regulatory application filed by BofI can or will be made prior to September 30, 2014; or (ii) BofI has (A) withdrawn a regulatory application filed by BofI in connection with the P&A Transaction, or (B) failed to provide any additional or supplemental information that any Approving Authority has requested in connection with its review of a regulatory application filed by BofI within the shorter of thirty (30) days or the date requested by the Approving Authority;

(d) By BofI, if (i) any Approving Authority has (A) denied approval of any regulatory application filed by HRB Bank or Block Financial in connection with any part of the Divestiture Transaction; or (B) indicated to HRB Bank or Block Financial that no final decision on a regulatory application filed by HRB Bank or Block Financial can or will be made prior to September 30, 2014; or (ii) HRB Bank or Block Financial has (A) withdrawn a regulatory application filed by HRB Bank or Block Financial in connection with any part of the Divestiture Transaction, or (B) failed to provide any additional or supplemental information that any Approving Authority has requested in connection with its review of a regulatory application filed by HRB Bank or Block Financial within the shorter of thirty (30) days or the date requested by the Approving Authority;

(e) By any party hereto in the event of a material breach by the other of any representation, warranty, covenant or agreement contained in this Agreement, which breach is not cured within ten (10) days (or such longer period not exceeding twenty (20) days in the event such breach cannot reasonably be cured within fifteen (15) days and a cure is being pursued with reasonable diligence) after written notice thereof is given to the party committing such breach or waived by such other party(ies);

(f) By BofI, if there is a Material Adverse Change with respect to (i) HRB Bank or Block Financial that would adversely affect in any material respect HRB Bank's or Block Financial's ability to perform their obligations under this Agreement or the Ancillary Agreements or (ii) the Transferred Assets and Assumed Liabilities (taken as a whole) that would adversely affect BofI in any material respect; or

(g) By HRB Bank or Block Financial if there is a Material Adverse Change with respect to BofI that would adversely affect in any material respect BofI's ability to perform its obligations under this Agreement or the Ancillary Agreements.

**9.02.** Effect of Termination. In the event of termination of this Agreement as provided in Section 9.01 above, this Agreement becomes void and without further effect and there will be no liability on the part of any party hereto or the respective officers and directors of any party, except for Section 6.07 (regarding expenses and payment of fees) and except that no termination of this Agreement pursuant to Section 9.01(e) shall relieve the non-performing or defaulting party of any liability to any other party hereto arising from the material non-performance and/or breach prior to the date of such termination of any covenant, agreement, term, provision, representation or warranty required to be observed, performed, complied with and/or kept by such non-performing or defaulting party. For the avoidance of doubt, the Nondisclosure Agreement will survive any termination of this Agreement.

# ARTICLE X

#### **INDEMNIFICATION**

**10.01. HRB Bank's Indemnity**. Subject to the provisions of this Article X, if the Closing occurs, the HRB Bank Indemnifying Parties shall jointly and severally indemnify and hold BofI, and its directors, officers, employees, agents, representatives, successors and assigns (collectively, the "<u>BofI Indemnified Parties</u>"), harmless from and against any and all damages, losses, costs, obligations, claims, causes of action, demands, assessments, judgments, settlements or Liability (whether based on contract, tort, product liability, strict liability or otherwise and whether ultimately determined to be valid), including Taxes and regulatory fines, and all reasonable costs and expenses (including prejudgment interest and other interest, penalties and attorneys' and accountants' fees and disbursements) of defending any of the foregoing or of enforcing this Agreement (collectively, "<u>Damages</u>") asserted against, resulting to, imposed upon or incurred by any BofI Indemnified Parties after the Closing, directly or indirectly, arising out of, resulting from or in connection with:

(a) Any inaccuracy or breach of any representation or warranty (ignoring, for purposes of determining the existence of any such inaccuracy or breach and the amount of Damages with respect thereto, any qualification as to Material Adverse Change, "materiality," "knowledge," "HRB Bank's Knowledge" or similar qualifier set forth in such representation or warranty) or any

breach or failure (regardless of whether such breach or failure is deemed "material") to perform any covenant or agreement made or undertaken by HRB Bank in this Agreement or in any Closing Ancillary Agreement, certificate, schedule, exhibit or writing delivered by HRB Bank and/or Block Financial pursuant to this Agreement;

(b) Any of the Excluded Assets or Excluded Liabilities;

(c) Any Liability to a third party based upon any act or omission of HRB Bank or any of its Affiliates that relates to any of the Transferred Assets or Assumed Liabilities (other than the Assumed Liabilities themselves) or the business of HRB Bank conducted prior to the Closing and which occurred or failed to occur on or before the Closing Date; and

(d) Any Litigation incident, or relating, to any matter covered by subsections (a), (b) or (c) of this Section 10.01.

**10.02. BofI Indemnity**. Subject to the provisions of this Article X, if the Closing occurs, BofI shall indemnify and hold HRB Bank and Block Financial, and their directors, officers, employees, agents, representatives, successors and assigns (the "<u>HRB Bank</u> <u>Indemnified Parties</u>") harmless from and against any and all Damages asserted against, resulting to, imposed upon or incurred by any HRB Bank Indemnified Parties after the Closing, directly or indirectly, arising out of, resulting from or in connection with:

(a) Any inaccuracy or breach of any representation or warranty (ignoring, for purposes of determining the existence of any such inaccuracy or breach and the amount of Damages with respect thereto, any qualification as to Material Adverse Change, "materiality," "knowledge," "BofI's Knowledge" or similar qualifier set forth in such representation or warranty) or any breach or failure (regardless of whether such breach or failure is deemed "material") to perform any covenant or agreement made or undertaken by BofI in this Agreement or in any Closing Ancillary Agreement, certificate, schedule, exhibit or writing delivered by BofI pursuant to this Agreement;

(b) The Assumed Liabilities;

(c) Any Liability to a third party based upon any act or omission of BofI that relates to any of the Transferred Assets or Assumed Liabilities and which occurs or fails to occur after the Closing Date; and

(d) Any Litigation incident, or relating, to any matter covered by subsections (a), (b) or (c) of this Section 10.02;

provided, however, that BofI shall have no obligation to indemnify an HRB Bank Indemnified Party under Section 10.02 (b), (c) or (d) (but only as subparagraph (d) relates to subparagraph (b) or (c)) to the extent that BofI has a right to indemnification from EFS under the Program Management Agreement based upon the same facts or circumstances.

If an HRB Bank Indemnified Party makes a claim against BofI for indemnification under Section 10.02(b), (c) or (d) (but only as subparagraph (d) relates to subparagraphs (b) or (c)) of this Agreement, and BofI reasonably believes that should such HRB Indemnified Party succeed in such claim, then BofI would itself have a claim for indemnification from EFS under Section 16.1 of the Program Management Agreement, then BofI shall so notify such HRB Indemnified Party, and such related claims for indemnification under this Agreement and under the Program Management

Agreement (collectively, "Related Claims") shall only be litigated, adjudicated, decided and/or resolved, as the case may be, in a single proceeding before a single forum.

Each Party hereby agrees to take all reasonable steps to ensure strict compliance with the terms of the immediately preceding paragraph, including: (i) jointly filing such motions or pleadings as may be required for a consolidation of all Related Claims in a single proceeding; (ii) in the case of HRB Bank Indemnified Parties, causing EFS to also agree that Related Claims shall be, and take all steps required for Related Claims to be, determined in a single proceeding as contemplated hereunder; and (iii) refraining from executing upon or in any manner attempting to collect any award with respect to any claim for indemnification hereunder (other than such actions as may be required to preserve or perfect such award) until a final and unappealable ruling has been made with respect to all Related Claims. With respect to any proceeding in which a HRB Indemnified Party claims indemnification from BofI and BofI, in turn, claims indemnification from EFS, any recovery finally awarded to a HRB Indemnified Party in connection with such claim shall be offset against any recovery finally awarded to BofI in connection with its claim against EFS, and only the difference, if any, shall then be paid to the appropriate Party.

**10.03. Indemnity Procedure**. All claims for indemnification by a party seeking to be indemnified (an "<u>Indemnified Party</u>") under this Article X shall be asserted and resolved as follows:

In the event that any claim or demand for which the HRB Bank Indemnifying Parties or BofI, as applicable (each, an "Indemnifying Party") may be liable to an Indemnified Party hereunder is asserted against or sought to be collected from such Indemnified Party by a third party, including a Governmental Entity (a "<u>Third Party Claim</u>"), such Indemnified Party shall with reasonable promptness give notice (the "<u>Claim Notice</u>") to the Indemnifying Party of such claim or demand, specifying the nature of and specific basis for such claim or demand and the amount or the estimated amount thereof to the extent then feasible (which estimate shall not be conclusive of the final amount of such claim and demand). Failure to provide a Claim Notice of any such claim or demand shall not affect the Indemnifying Party's duties or obligations under this Article X, except to the extent the Indemnifying Party is materially prejudiced thereby. The Indemnifying Party shall have ten (10) days from the delivery or mailing of the Claim Notice (the "Notice Period") to notify the Indemnified Party (i) whether or not it disputes the liability of the Indemnifying Party to the Indemnified Party hereunder with respect to such claim or demand, and (ii) whether or not it desires, at the cost and expense of the Indemnifying Party, to defend the Indemnified Party against such claim or demand; provided, however, that any Indemnified Party is hereby authorized, but is not obligated, prior to and during the Notice Period, to file any motion, answer or other pleading that it shall deem necessary or appropriate to protect its interests or those of the Indemnifying Party, and provided further, that in respect of a claim or demand for Taxes no such right of the Indemnifying Party to defend shall apply unless and to the extent the Indemnified Party consents. If the Indemnifying Party notifies the Indemnified Party within the Notice Period that it desires to defend the Indemnified Party against such claim or demand, the Indemnifying Party shall, subject to the preceding sentence and the last sentence of this Section 10.03(a), have the right to control the defense against the claim by all appropriate proceedings and any settlement negotiations; provided, however, that to the satisfaction of the Indemnified Party, the Indemnifying Party shall indemnify and secure the Indemnified Party against such contested claims by posting a bond or otherwise. If the Indemnifying Party undertakes the defense of the Third Party Claim, then the Indemnified Party shall have the right to participate in the defense of the Third Party Claim at its own expense; provided,

however, that the Indemnifying Party shall bear the reasonable fees and expenses of such separate counsel for the Indemnified Party if the representation by one counsel of both the Indemnifying Party and the Indemnified Party in any such Third Party Claim would, in the reasonable opinion of counsel for the Indemnified Party, be inappropriate due to a conflict of interest; provided, further, that the Indemnifying Party shall not have the right to assume control of such defense and shall pay the fees and expenses of counsel retained by the Indemnified Party if the claim which the Indemnifying Party seeks to control (i) involves criminal allegations or (ii) is one in which the Indemnified Party is advised by counsel chosen by it that there are legal defenses available to the Indemnified Party which the Indemnifying Party cannot assert on behalf of the Indemnified Party. If the Indemnifying Party fails to respond to the Indemnified Party within the Notice Period or after electing to defend fails to commence or diligently pursue such defense, then the Indemnified Party shall have the right, but not the obligation, to undertake or continue the defense of and to compromise or settle (exercising reasonable business judgment) the claim or other matter, all on behalf of, for the account, at the expense and at the risk of the Indemnifying Party. Notwithstanding the foregoing, if the basis of the proceeding relates to a condition or operations which existed or were conducted both prior to and after the Closing Date, each party shall have the same right to participate in the proceeding without either party having the right of control.

(b) If requested by the Indemnifying Party, the Indemnified Party agrees, at the Indemnifying Party's expense, to reasonably cooperate with the Indemnifying Party and its counsel in contesting any Third Party Claim that the Indemnifying Party elects to contest.

(c) If any Indemnified Party should have a claim or demand against the Indemnifying Party hereunder that does not involve a Third Party Claim, the Indemnified Party shall send a Claim Notice with respect to such claim or demand to the Indemnifying Party. Failure to provide a Claim Notice of any such claim or demand shall not affect the Indemnifying Party's duties or obligations under this Article X, except to the extent the Indemnifying Party is materially prejudiced thereby. If the Indemnifying Party disputes such claim, such dispute shall be resolved by mutual agreement of the Indemnifying Party and Indemnified Party or by litigation in an appropriate court of competent jurisdiction.

(d) The Indemnifying Party shall not (i) settle any claim in a manner that imposes any liability upon the Indemnified Party or admits any wrongdoing by the Indemnified Party without the Indemnified Party's prior written consent, which consent may be granted or withheld in the Indemnified Party's sole and absolute discretion; or (ii) otherwise compromise or settle a claim against the Indemnified Party without the Indemnified Party's prior written consent, which consent shall not be unreasonably withheld or delayed.

#### 10.04. Limitations on Indemnity.

(a) Notwithstanding the foregoing, neither the HRB Bank Indemnifying Parties nor BofI shall be required to indemnify the BofI Indemnified Parties or the HRB Bank Indemnified Parties, as applicable, in respect of any Damages suffered by the BofI Indemnified Parties or the HRB Bank Indemnified Parties, as applicable, in connection with a breach of representation or warranty pursuant to Section 10.01(a) or Section 10.02(a), as the case may be, unless the aggregate amount of the Indemnifying Parties obligation under this Article X exceeds an amount equal to \$50,000 (the "<u>Threshold Amount</u>"), in which case the HRB Bank Indemnifying Parties or BofI, as applicable, will be required to indemnify the BofI Indemnified Parties or the HRB Bank Indemnifying Parties or BofI.

Parties, as applicable, for all Damages regardless of the Threshold Amount, back to the first dollar of Damages.

(b) No indemnification will be provided by any Indemnifying Party for any claim for indemnification which is made more than three (3) years following the Closing Date; provided, however, that claims for indemnification may be made for any Fundamental Representations until 30 days following the expiration of the applicable statute of limitations, and such Fundamental Representations shall survive until such time. Notwithstanding the foregoing, if at the end of such period there shall be pending any indemnification claim by a person, such person shall continue to have the right to seek such indemnification with respect to such claim notwithstanding such expiration.

(c) No Indemnified Party shall be required to (i) incur any material out-of-pocket costs or expenses or pay any other material amounts to third parties, except to the extent that the Indemnifying Party has acknowledged in writing that such costs, expenses or other amounts constitute indemnifiable Damages hereunder, (ii) make any claims under existing insurance policies, or (iii) take any other action to the extent such action would adversely affect such Indemnified Party in any material respect.

(d) For the avoidance of doubt, this Article X does not provide for any indemnification or other remedy relating to the performance after the Closing of the obligations of the parties under the Program Management Agreement, the Receivables Participation Agreement or any document, agreement or instrument to be executed in connection with the Program Management Agreement or the Receivables Purchase Agreement.

**10.05.** Exclusive Remedy. After the Closing Date, this Article X shall provide the exclusive remedy for any of the matters addressed herein or other claims arising out of this Agreement, provided that the foregoing shall not be deemed to limit the rights of any party to seek equitable remedies (including specific performance or injunctive relief) or seek Damages in the case of any fraud, willful or knowing breach or misrepresentation.

# ARTICLE XI

#### TRANSITIONAL AND POST-CLOSING MATTERS

#### 11.01. Notification to Customers.

(a) BofI will, jointly with HRB Bank, unless otherwise required earlier by applicable Law or Contract, on a date promptly after the Closing Date, prepare and mail to each customer whose Deposit is to be assumed by BofI and whose Loan is to be purchased by BofI, a letter, in form and substance reasonably satisfactory to the parties, informing such Person of the nature of this transaction and the services to be provided by BofI on and after the Closing Date.

(b) Either as part of or in addition to the notification to customers described in Section 11.01(a), subject to the terms of the Program Management Agreement, BofI may provide notification of changes in terms to the Transferred Assets or Assumed Liabilities that will take effect shortly following the Closing Date and will provide additional notices to customers of changes to occur upon conversion of the Deposit Accounts to BofI's processing system.

**11.02. Payment of Instruments**. Following the Closing, BofI agrees to pay all checks, drafts, and withdrawal orders (including ACH debits) with respect to the Deposits to the extent that (a) such payment is in accordance with applicable Law and the overdraft and other policies applicable to such Deposits, (b) they are otherwise properly payable, and (c) they are presented to BofI by mail, over its counters, or by any other authorized means. BofI shall in all other respects discharge, in the usual course of business, the duties and obligations of HRB Bank with respect to the Deposits and the balances due and owing thereunder. Subject to the terms of the Program Management Agreement, HRB Bank shall cooperate in a commercially reasonable fashion with BofI to enable BofI to (x) amend and reissue Contracts and disclosures relating to the Deposits, Transferred Loans and Overdrafts, (y) reissue cards (other than Emerald Cards) issued to customers with respect to the Deposits and Transferred Loans, no later than the date on which such reissuance would be required by MasterCard, and (z) otherwise facilitate the transfer of such customers' business relationships to BofI. HRB Bank will also for a period of ninety (90) days after the Closing continue to transfer to BofI any withdrawal orders (including ACH debits) received by it relating to any of the Deposits.

**11.03. Statements**. To the extent required by applicable Law or Contract, (a) HRB Bank will issue statements to its customers that include all transactions with respect to the Deposits and the Loans through an agreed-upon date, and (b) BofI will issue statements for all transactions with respect to the Deposits and the Loans thereafter. Interest and service charge calculations will be processed on behalf of HRB Bank on such customer statements as of the close of business on the Closing Date.

**11.04.** Access to Records. To the extent any books, records and files relating to the Transferred Assets and the Assumed Liabilities are not Records for purposes of this Agreement but are material in relation to any of the Transferred Assets or Assumed Liabilities or BofI's ownership or management thereof, HRB Bank will use commercially reasonable efforts to provide BofI with the material information contained in such books, records or files. In addition, HRB Bank and BofI mutually agree to maintain all other records and other documents relating to the Transferred Assets and Assumed Liabilities for such periods as provided in HRB Bank and BofI's respective record retention policies and required by applicable Law, and to permit the other party to examine, inspect, copy and reproduce such records and other documents relating to such Transferred Assets and Assumed Liabilities as may be reasonably requested by the other party. Any charges for such examination and photocopying will be at a rate not greater than the examining party's customary rates for similar requests by its customers.

**11.05. Information Reporting**. With respect to the Loans and Deposits purchased and assumed by BofI pursuant to this Agreement, HRB Bank will be responsible for reporting to the customer and to the Internal Revenue Service (and any state or local taxing authority as required by Law) all interest paid or earned by the customer prior to and including the Closing Date, and BofI will be responsible for reporting to the customer and to the Internal Revenue Service (and any state or local taxing authority as required by Law) all interest paid or earned by the Closing Date.

**11.06. Transition**. From and after the date of this Agreement, HRB Bank and BofI agree to fully cooperate with and assist one another in connection with the transition and conversion of all customer accounts, files (including data processing files) and other information which are being

purchased and assumed by BofI pursuant to the terms hereof, including providing all information that BofI may reasonably request at such times and in such formats as BofI may reasonably request. Additionally, each of BofI and HRB Bank agree to provide each other, upon reasonable prior notice, with such information and data as is necessary to allow HRB Bank and BofI to comply with all tax, regulatory reporting, audit or other compliance obligations relating to the customers, employees (to the extent applicable) and Transferred Assets and Assumed Liabilities, and each of HRB Bank and BofI agree to timely take any and all action as required by Law to comply with such tax, regulatory and/or reporting obligations.

#### ARTICLE XII

#### **GENERAL PROVISIONS**

**12.01. Amendment**. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.

**12.02.** Waiver. Any term, condition or provision of this Agreement may be waived in writing at any time by the party which is entitled to the benefits thereof. No failure or delay on the part of any party hereto to exercise any right, power, or privilege hereunder or under any instrument executed pursuant hereto operates as a waiver nor does any single or partial exercise of any right, power, or privilege preclude any other further exercise thereof or the exercise of any other right, power, or privilege.

**12.03.** Survival of Representations and Warranties. The parties hereto agree that all of the representations and warranties contained in this Agreement shall survive for a period of three (3) years subsequent to the Closing Date; other than the Fundamental Representations which shall survive until 30 days following the expiration of their respective statutes of limitation.

**12.04. Binding Effect; Assignment**. Subject to Section 2.05, no party may assign its rights or obligations under this Agreement without the prior written consent of the other parties to this Agreement. Subject to the preceding sentence, all terms of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns.

**12.05.** Severability. Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable Law, but if any provision of this Agreement if held to be prohibited by or invalid under applicable Law, such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating the remaining provisions of this Agreement.

**12.06. Headings**. Article, Section, Subsection, and Paragraph titles, captions and headings herein are inserted only as a matter of convenience and for reference, and in no way define, limit, extend, or describe the scope of this Agreement or the intent of any provision hereof.

12.07. Entire Agreement. Except as relates to the Nondisclosure Agreement, the Program Management Agreement and the Receivables Participation Agreement, this Agreement and the Schedules and Exhibits hereto constitutes the entire agreement between and among the parties with respect to the subject matter hereof, supersedes all prior negotiations, representations, warranties, commitments, offers, letters of interest or intent, proposal letters, Contracts, or understandings with respect thereto.

**12.08. Counterparts**. This Agreement may be executed in one or more counterparts, and any party to this Agreement may execute and deliver this Agreement by executing and delivering any of such counterparts, each of which when executed and delivered shall be deemed to be an original and all of which taken together shall constitute one and the same instrument. The exchange of copies of this Agreement and of signature pages by facsimile or electronic mail transmission shall constitute effective execution and delivery of this Agreement as to the parties and may be used in lieu of the original Agreement for all purposes. Signatures of the parties transmitted by facsimile or electronic mail signatures for all purposes.

**12.09.** Notices. All notices required or permitted under this Agreement must be given in writing, reference this Agreement and will be deemed delivered and given (i) upon personal delivery to the party to be notified; (ii) on the date sent by facsimile or email (with confirmation of transmission); (iii) on the date received if sent by registered or certified U.S. mail, return receipt requested, postage and charges prepaid; (iii) one Business Day after deposit with a nationally-recognized commercial overnight courier, specifying next day delivery, with written verification of receipt. All communications shall be sent to the following addresses (or at such other address for a party as shall be specified by like notice):

If to BofI: BofI Federal Bank 4350 La Jolla Village Drive, Suite 140 San Diego, California 92122 Attn: Gregory Garrabrants Email: ggarrabrants@bofifederalbank.com Telephone: 858-350-6203 Facsimile: 858-764-6561

With copies to:

(a)

BofI Federal Bank 4350 La Jolla Village Drive, Suite 140 San Diego, California 92122 Attn: Eshel Bar-Adon Email: ebaradon@bofifederalbank.com Telephone: 858-764-2905 Facsimile: 858-764-6561

 (b) If to HRB Bank: H&R Block Bank One H&R Block Way Kansas City, MO 64105 Attn: Greg Quarles Email: greg.quarles@hrblock.com Telephone: 816-854-5709 Facsimile: 816-854-8052 Sheppard Mullin Richter & Hampton, LLP 333 South Hope Street, 43<sup>rd</sup> Floor Los Angeles, CA 90071-1422 Attn: Lawrence M. Braun Email: LBraun@sheppardmullin.com Telephone: 213-617-4184 Facsimile: 213-443-2814

With copies to:

H&R Block, Inc. One H&R Block Way Kansas City, MO 64105 Attn: David Zarski, Vice President, Legal - Financial Services and Walter Pirnot, Senior Corporate Counsel Email:david.zarski@hrblock.com; wpirnot@hrblock.com Telephone: 816-8544502; 816-854-5757 Facsimile: 816-802-1065 Stinson Leonard Street LLP Attn: Mike Lochmann 1201 Walnut Street, Suite 2900 Kansas City, MO 64106 Email: mike.lochmann@ stinsonleonard.com Telephone: 816-691-3208 Facsimile: 816-412-1249

(c) If to Block Financial:

Block Financial, LLC One H&R Block Way Kansas City, MO 64105 Attn: Gregory J. Macfarlane, President Email: gregory.macfarlane@hrblock.com Telephone: 816-854-7565 Facsimile: 816-854-8500

With copies to:

H&R Block, Inc. One H&R Block Way Kansas City, Missouri 64105 Attn: Tom Gerke, Chief Legal Officer Email: tom.gerke@hrblock.com Telephone: 816-854-6060 Facsimile: 816-854-8500 Stinson Leonard Street LLP Attn: Mike Lochmann 1201 Walnut Street, Suite 2900 Kansas City, Missouri 64106 Email: mike.lochmann@ stinsonleonard.com Telephone: 816-691-3208 Facsimile: 816-412-1249

provided, however, that the providing of notice to counsel shall not, of itself, be deemed the providing of notice to a party hereto. All notices provided pursuant to this Section 12.09 shall be effective on the day of completion of all delivery requirements.

# 12.10. Governing Law; Jurisdiction; Waiver of Jury Trial.

(a) This Agreement is governed by and controlled as to validity, enforcement, interpretation, effect, and in all other respects by the internal laws of the State of New York applicable to Contracts made in that state, without reference to its conflicts of laws principles.

(b) Except as otherwise expressly provided in this Agreement, any suit, action or proceeding seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement or the transactions contemplated hereby shall be in the United States District Court for the Eastern District of Missouri, located in the City of St. Louis, or as to

those lawsuits to which the Federal Courts of the United States lack subject matter jurisdiction, before a court located in the State of Missouri in the City of St. Louis, and each of the parties hereby consents to the jurisdiction of such courts (and of the appropriate appellate courts therefrom) in any such suit, action or proceeding and irrevocably waives, to the fullest extent permitted by Law, any objection which it may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such court or that any such suit, action or proceeding which is brought in any such court has been brought in an inconvenient forum. The parties acknowledge and agree that this Agreement was executed and delivered in the State of Missouri. Process in any such suit, action or proceeding may be served on any party anywhere in the world, whether within or without the jurisdiction of any such court. Without limiting the foregoing, each party agrees that service of process on such party as provided in Section 12.09 shall be deemed effective service of process on such party.

(c) EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

**12.11.** Third Party Beneficiaries. The parties intend that this Agreement is not to benefit or create any right or cause of action in or on behalf of any Person other than the parties hereto and their affiliates. No future or present employee or customer of any of the parties is to be treated as a third party beneficiary in or under this Agreement.

**12.12. Specific Performance**. The parties acknowledge that monetary damages could not adequately compensate the parties in the event of a breach of this Agreement by one party, that the non-breaching party or parties would suffer irreparable harm in the event of such breach and that the non-breaching party or parties have, in addition to any other rights or remedies it or they may have at law or in equity, specific performance and injunctive relief as a remedy for the enforcement hereof.

**12.13. Mutual Drafting**. This Agreement is the mutual product of the parties hereto and each provision hereof has been subject to the mutual consultation, negotiation and agreement of each party and shall not be construed for or against any party.

**12.14. Interpretive Provisions**. For purposes of this Agreement, (a) the words "including" and "include" shall be deemed to be followed by the words "including without limitation" and "include without limitation," respectively and (b) the words "herein," "hereof," "hereby," "hereto" or "hereunder" refer to this Agreement. Unless the context otherwise requires, references in this Agreement: (a) to Articles, Sections, Subsections, Paragraphs, Exhibits and Schedules mean the Articles, Sections, Subsections and Paragraphs of, and the Exhibits and Schedules attached to, this Agreement; (b) to an agreement, instrument or other document means such agreement, instrument or other document as amended, supplemented and modified from time to time to the extent permitted by the provisions thereof and by this Agreement; and (c) to a statute means such statute as amended from time to time and includes any successor legislation thereto and any regulations promulgated thereunder. The Schedules and Exhibits referred to in this Agreement shall be construed with and as an integral part of this Agreement to the same extent as if they were set forth verbatim herein.

[Execution page follows]

**IN WITNESS WHEREOF**, the parties hereto have caused this Agreement to be signed by their respective officers thereunto duly authorized as of date first above written.

#### **BOFI FEDERAL BANK**

By: <u>/s/ Gregory Garrabrants</u>

Name: Gregory Garrabrants Title: President and Chief Executive Officer

# **H&R BLOCK BANK**

By: <u>/s/ Greg M. Quarles</u> Name: Greg M. Quarles Title: President and Chief Executive Officer

# **BLOCK FINANCIAL LLC**

By: <u>/s/ Gregory J. Macfarlane</u> Name: Gregory J. Macfarlane Title: President

Purchase and Assumption Agreement Signature Page



List of Schedules and Similar Attachments Omitted from Purchase and Assumption Agreement

Schedule 3.02	Sample Transfer Amount Calculation
Schedule 3.03	Prior Engagements
Schedule 3.06	Items of Income and Expenses to be Prorated
Schedule 6.01	Transactions Outside the Ordinary Course of Business
Schedule 8.01(h)	Required Third Party Consents
Exhibit A	Assignment and Assumption of Deposit Liabilities Agreement
Exhibit B	Assignment and Assumption of Assumed Contracts Agreement
Exhibit C	Bill of Sale
Exhibit D	Assignment and Assumption of Loans Agreement
Exhibit E	Assignment, Transfer and Appointment of Successor Custodian for Custodial Accounts
Exhibit F	Limited Power of Attorney
Exhibit J	Company Financial Products Distribution Agreement
Exhibit K	Franchisee Financial Products Distribution Agreement
Exhibit L	Trademark Licensing Agreement
Exhibit M	Trademark Licensing Agreement
Exhibit N	Collection Services Agreement
Exhibit O	Amended and Restated Legacy Emerald Advance Participation Agreement
Exhibit P	Amended and Restated Legacy Emerald Unsecured Credit Card Participation Agreement
Exhibit G - Program Management Agreement, Exhibit H - Guaranty Agreement, and Exhibit I - Emerald Advance Receivables	
Participation Agreement were filed as exhibits to the registrant's Current Report on Form 8-K filed April 10, 2014.	

Upon request, the registrant agrees to furnish supplementally to the Securities and Exchange Commission a copy of any omitted schedule or similar attachment to the Purchase and Assumption Agreement; provided, however, that the registrant may request confidential treatment of omitted items prior to any public disclosure.

# **PROGRAM MANAGEMENT AGREEMENT**

By and Between

# EMERALD FINANCIAL SERVICES, LLC,

and

**BofI FEDERAL BANK** 

Dated as of

\_\_\_\_\_, 2014

# **TABLE OF CONTENTS**

# ARTICLE 1 DEFINITIONS; ORDER OF PRECEDENCE; RULES OF INTERPRETATION2

- 1.1 Definitions 2
- 1.2 Order of Precedence 12
- 1.3 Rules of Interpretation 12

# ARTICLE 2 PROGRAM DESCRIPTION 13

- 2.1 Establishment of the Program 13
- 2.2 Financial Products 13
- 2.3 Bank Obligations 14
- 2.4 EFS Obligations 15
- 2.5 Program Economics 16
- 2.6 Distributors 16
- 2.7 Franchisees 16
- 2.8 Website Supporting the Program 17
- 2.9 Promotion of Program; Program Marketing Plan 17
- 2.10 Back-Up Account Originator 18
- 2.11 Program Eligibility Policy 18
- 2.12 Servicing Responsibilities 18
- 2.13 Service Level Agreements 19
- 2.14 Expenses 19
- 2.15 Settlement Statements 19
- 2.16 Nevada Office 19
- 2.17 Program Infrastructure 19

#### ARTICLE 3 PROGRAM SUPPORT20

- 3.1 Communications; Dispute Resolution 20
- 3.2 Dispute Resolution 21

ARTICLE 4 DECISION-MAKING AUTHORITY; REGULATORY COORDINATION; COMPLIANCE OBLIGATIONS; 22 BANK OVERSIGHT; BANK COMPLIANCE PROGRAM

- 4.1 Decision-Making Authority 22
- 4.2 Regulatory Coordination 23
- 4.3 Compliance Obligations 23
- 4.4 Bank Oversight 24
- 4.5 Bank Compliance Program 24

#### ARTICLE 5 EXCLUSIVITY; NEW PRODUCTS; CHANGES TO EXISTING PRODUCTS AND PROGRAM 25

- 5.1 Bank's Right to Offer Financial Products to Others 25
- 5.2 EFS to Offer Only Financial Products of Bank 25
- 5.3 EFS Right to Suspend Offering of Financial Product 26
- 5.4 Enhancements to Financial Products 26
- 5.5 Changes to Program During Tax Season 26

- 5.6 New Products 26
- 5.7 Staffing Plan 27

### ARTICLE 6 ACCOUNT DOCUMENTATION; MARKETING; CROSS MARKETING27

- 6.1 Account Documentation, Marketing Materials and Servicing Materials 27
- 6.2 Cross Marketing 28
- 6.3 Customers of Bank 29

# ARTICLE 7 OUTSOURCING RESTRICTIONS; SERVICE PROVIDERS 29

- 7.1 Outsourcing Restrictions 29
- 7.2 Service Providers 29

# ARTICLE 8 AUDIT RIGHTS; REPORTING30

- 8.1 Bank Audit Rights and Obligations 30
- 8.2 EFS Audit Rights and Obligations 31
- 8.3 Audits by Regulatory Authorities 33
- 8.4 Reporting 33
- 8.5 SEC Reporting and Public Announcements 34
- 8.6 OCC 2013-29 34
- 8.7 OCC Oversight 35

# ARTICLE 9 INTELLECTUAL PROPERTY; LICENSE TO USE MARKS; OWNERSHIP RIGHTS 35

- 9.1 Licensing Agreements 35
- 9.2 Ownership and Licenses of Intellectual Property 35

# ARTICLE 10 REPRESENTATIONS, WARRANTIES AND COVENANTS35

- 10.1 EFS's Representations, Warranties and Covenants 35
- 10.2 Bank's Representations, Warranties and Covenants 37
- 10.3 Mutual Covenants 39

# ARTICLE 11 CONFIDENTIALITY39

- 11.1 Confidential Information 39
- 11.2 Limits on Use and Disclosure 40
- 11.3 Regulatory Requests 41
- 11.4 Disposition of Confidential Information 41
- 11.5 Unauthorized Use or Disclosure 41

#### ARTICLE 12 PRIVACY AND DATA SECURITY42

- 12.1 Privacy 42
- 12.2 Data Security 42
- 12.3 Collection, Ownership and Use of Program Customer Data 44

# ARTICLE 13 EVENTS OF DEFAULT44

- 13.1 EFS Event of Default 44
- 13.2 Bank Event of Default 45

ARTICLE 14 TERM AND TERMINATION46

- 14.1 Term 46
- 14.2 Mutual Termination Rights 46
- 14.3 Additional Termination Rights of EFS 46
- 14.4 Cross Termination of Product Schedules 47

#### ARTICLE 15 RIGHTS UPON TERMINATION47

- 15.1 EFS Purchase Options 47
- 15.2 Evaluation Data 48
- 15.3 Determination of Purchase Price 48
- 15.4 Purchase Mechanics 49
- 15.5 Duties After Termination or Expiration 49
- 15.6 Wind-Down by Bank 50
- 15.7 Communication with Accountholders 51
- 15.8 Applicability 51

#### ARTICLE 16 INDEMNIFICATION; LIMITATION OF LIABILITY51

- 16.1 Indemnification of Bank by EFS 51
- 16.2 Indemnification of EFS by Bank 52
- 16.3 Notice 52
- 16.4 Right to Defend Claims; Coordination of Defense 53
- 16.5 Settlement of Claims 53
- 16.6 Subrogation 54
- 16.7 Indemnification Payments 54
- 16.8 Apportionment of Costs 54
- 16.9 Limitation of Liability 55

#### ARTICLE 17 MISCELLANEOUS55

- 17.1 Assignment 55
- 17.2 Entire Agreement; Amendments 55
- 17.3 No Third-Party Beneficiaries 55
- 17.4 Non-Waiver of Default 55
- 17.5 Severability 56
- 17.6 Further Assurances 56
- 17.7 Notices 56
- 17.8 Force Majeure 57
- 17.9 Interpretation 58
- 17.10 Exhibits and Schedules 58
- 17.11 No Agency 58
- 17.12 Relationship of Parties 58
- 17.13 Governing Law 58
- 17.14 Consent to Jurisdiction 58
- 17.15 Waiver of Trial by Jury 59
- 17.16 Cumulative Remedies; Waivers 59
- 17.17 Binding Agreement 59
- 17.18 Survival 59
- 17.19 Multiple Counterparts and Facsimile Signatures 59

Signature PageS-1

# List of Schedules

Schedule 2.1(a):Financial Products By TerritorySchedule 2.4(f):EFS Audit PlanSchedule 3.1(a)(i):Initial Designated ExecutivesSchedule 3.1(b)(i):Initial Senior Program ManagersSchedule 7.1:List of Internationally Outsourced Service ProvidersSchedule 7.2(a):Material Third Party Service ProvidersSchedule 15.3:Calculation of Fair Value of Emerald Advance Participation Interests

# **List of Product Schedules**

Schedule A: Prepaid Products Product Schedule

Schedule B: Refund Transfer Product Schedule

Schedule C: Emerald Advance Product Schedule

Schedule D: Credit Card Product Schedule

Schedule E: Deposit Products Schedule

Schedule F: Service Level Agreements

#### PROGRAM MANAGEMENT AGREEMENT

THIS PROGRAM MANAGEMENT AGREEMENT (this "<u>Agreement</u>"), dated as of \_\_\_\_\_\_, 2014 is made by and between **EMERALD FINANCIAL SERVICES**, LLC, a limited liability company organized under the laws of Delaware ("<u>EFS</u>"), and **BofI FEDERAL BANK**, a federal savings bank ("<u>Bank</u>"). EFS and Bank are at times hereinafter referred to as the "<u>Parties</u>" and each individually as a "<u>Party</u>."

#### **RECITALS:**

A. Bank has purchased certain assets and assumed certain liabilities related to accounts ("<u>Purchased Accounts</u>") held by H&R Block Bank ("<u>HRB Bank</u>") under a purchase and assumption agreement, dated \_\_\_\_\_\_, 2014 ("<u>Purchase Agreement</u>").

B. HRB Tax Group, Inc., a Missouri corporation ("<u>HRB Tax Group</u>"), and certain of its subsidiaries are in the business of providing (or making available through Franchisees) tax preparation and related products and services to consumer customers (including customers of Franchisees) ("<u>Company Customers</u>") throughout the Program Territory.

C. HRB Tax Group and EFS are indirect subsidiaries of H&R Block, Inc., a Missouri corporation ("<u>Block, Inc.</u>"; and together with all its subsidiaries, the "<u>Company</u>").

D. In conjunction with other products and services offered by the HRB Tax Group and its subsidiaries, HRB Bank developed and offered a prepaid card program and companion financial products and services to Company Customers (collectively, the "<u>HRB Bank Financial Products</u>").

E. Concurrent with closing under the Purchase Agreement (the "<u>Closing</u>"), HRB Bank will no longer offer the HRB Bank Financial Products.

F. Bank has developed, and shall continue to develop, various prepaid card programs, under which it issues reloadable and non-reloadable prepaid cards, as well as a series of card-related products and services in conjunction therewith, including products and services similar to the HRB Bank Financial Products.

G. It is the intention of the Parties that upon Closing the BINs, ICA numbers and customer relationships related to the HRB Bank Financial Products be transferred by HRB Bank to Bank, and that Bank thereafter offer to Company Customers Financial Products upon terms and conditions that are the same or substantially similar to those applicable to the HRB Financial Products, as well as such future Bank sponsored and EFS marketed products and services as the Parties may from time to time mutually agree (collectively, the "Program").

H. Bank now desires to engage EFS, and EFS desires to be engaged, to serve as program manager and provide program management and processing services in connection with the Program subject to the terms and conditions set forth herein.

#### AGREEMENT

ACCORDINGLY, in consideration of the mutual covenants and agreements of the Parties herein contained and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties hereby agree as follows:

#### ARTICLE 1 DEFINITIONS; ORDER OF PRECEDENCE; RULES OF INTERPRETATION

1.1 <u>Definitions</u>. Capitalized terms used in this Agreement shall have the meanings set forth below; provided, however, that with respect to any capitalized term specifically defined in any Product Schedule, such term shall have the meaning set forth in such Product Schedule:

"<u>Account</u>" means (a) each loan account, deposit account and prepaid account originated by Bank for a Company Customer associated with any Financial Products offered in connection with the Program, and (b) each Purchased Account.

"<u>Account Documentation</u>" means, with respect to an Account, the applications, Accountholder agreements, disclosures, privacy notices, change of terms notices, including any and all amendments or modifications thereto, however and wherever stored or kept, and any other written information relating to such Account's terms and conditions.

"Accountholder" means any Person who holds or has held an Account.

"<u>Accountholder Data</u>" means all Personally Identifiable Information regarding an Accountholder received by or on behalf of Bank or by EFS in connection with a Financial Product, or obtained by or on behalf of Bank or EFS in connection with an Account but shall not include Company Customer Data.

"<u>Accounts Receivable</u>" means any and all amounts owing from time to time by an Accountholder on an Account, whether billed or unbilled, including any unpaid balances for purchases, accrued finance charges, late fees and any other charges and fees assessed on an Account.

"Acquiring IP Party" has the meaning set forth in Section 9.2(a) (Ownership of Intellectual Property).

"Affected Party" has the meaning set forth in Section 12.2(b) (Data Security).

"<u>Affiliate</u>" means any Person that, directly or indirectly, through one or more intermediaries, (a) owns or controls another Person, (b) is owned or controlled by another Person, or (c) is under common control or ownership with another Person, and "ownership" means the direct or indirect beneficial ownership of more than 30% of the equity securities of a Person, or, in the case of a Person that is not a corporation, more than 30% of the voting and/or equity interest of such Person.

"<u>Agreement</u>" has the meaning set forth in the Preamble.

"<u>Applicable Law</u>" means any and all laws, treaties, rules, regulations, regulatory guidance and determinations of a Regulatory Authority, mandatory written direction from a Regulatory Authority, and orders opinions and interpretations of any Regulatory Authority, including under the Bank Secrecy Act, UDAAP, laws relating to anti-money laundering (including customer due diligence and enhanced due diligence necessary to meet "know your customer" requirements), identity theft, fraud schemes, and predatory, unfair or deceptive acts, any and all sanctions or regulations enforced by OFAC, and statutes or regulations of any state relating to gift cards, money transmission or unclaimed property, that are applicable to the Program, or otherwise applicable to any of the Parties, or any Distributor or Franchisee.

"Applicant" means any Person who has submitted an application to Bank for a Financial Product.

"<u>Applicant Data</u>" means all Personally Identifiable Information regarding an Applicant received by or on behalf of Bank (including by EFS as servicer) in connection with such Applicant's application for a Financial Product, but shall not include Company Customer Data or Bank Customer Data.

"Assumed Accounts Purchase Date" has the meaning set forth in Section 15.1 (EFS Purchase Options).

"Assumed Liabilities" has the meaning set forth in Section 15.1 (EFS Purchase Options).

"Back-Up Account Originator" has the meaning set forth in Section 2.10 (Back-Up Account Originator).

"Back-Up Notice" has the meaning set forth in Section 2.10 (Back-Up Account Originator).

"Bank" has the meaning set forth in the Preamble.

"Bank Corrective Plan" has the meaning set forth in Section 8.2(b) (Bank Audit Rights and Obligations).

"Bank Customer" means any individual who has or had a banking relationship with Bank that was not originated through the Program or acquired under the Purchase Agreement.

"Bank Customer Data" means data relating to Bank Customers, other than Program Customer Data.

"Bank Event of Default" has the meaning set forth in Section 13.2 (Bank Event of Default).

"Bank Indemnified Parties" has the meaning set forth in Section 16.1 (Indemnification of Bank by EFS).

"<u>Bank Licensed Marks</u>" means the trademarks, trade names, service marks, logos and other proprietary designations of Bank licensed to EFS by Bank under the EFS Licensing Agreement.

"Bank Licensing Agreement" means the Trademark Licensing Agreement substantially in the form attached as Exhibit L to the Purchase Agreement, pursuant to which HRB Innovations is granting a license to Bank.

"Bank Matters" has the meaning set forth in Section 4.1(a) (Bank Matters).

"<u>Bank Rules</u>" means the policies, procedures, operating rules and regulations of Bank, as amended from time to time by Bank in the exercise of its reasonable discretion, and incorporated into the Operating Procedures or Product Schedules.

"Bank Senior Program Manager" has the meaning set forth in Section 3.1(b)(i) (Program Managers).

"<u>Bank Service Provider</u>" means a third-party service provider (including an Affiliate of Bank or a Material Bank Service Provider) used by Bank in connection with the performance of Bank's obligations under this Agreement, other than EFS or an EFS Service Provider.

"BIN" has the meaning set forth in Section 15.5(b) (ABA Routing Number; BIN; ICA).

"Block, Inc." has the meaning set forth in the Recitals.

"<u>Block Licensing Agreement</u>" means the Trademark Licensing Agreement substantially in the form attached as Exhibit M to the Purchase Agreement, pursuant to which Bank is granting a license to certain subsidiaries of Block, Inc.

"Business Day" means any day, except a Saturday, Sunday or a federal legal holiday.

"<u>Claim</u>" means any claim, demand, suit, legal action, regulatory action, administrative action, arbitration or proceeding, including those brought in connection with allegations of misrepresentations, breach of warranty, breach of contract, violation of Applicable Law, unfair or deceptive acts or practices, or otherwise seeking to recover Indemnified Losses.

"Closing" has the meaning set forth in the Recitals.

"Closing Date" has the meaning set forth in Section 15.4 (Purchase Mechanics).

"<u>Code</u>" has the meaning set forth in <u>Section 6.2(a)</u> (Cross Marketing).

"Company" has the meaning set forth in the Recitals.

"<u>Company Applicable Law</u>" has the meaning set forth in <u>Section 4.3(b)</u> (Compliance Obligations).

"Company Customer" has the meaning set forth in the Recitals.

"<u>Company Customer Data</u>" means all Personally Identifiable Information regarding a Company Customer or tax return information (as defined in IRC § 7216) obtained in connection with the provision of Company products and services to such Company Customer.

"<u>Company Distribution Agreement</u>" means the Company Financial Products Distribution Agreement in substantially the form attached as Exhibit J to the Purchase Agreement.

"<u>Company Licensed Marks</u>" means the trademarks, trade names, service marks, logos and other proprietary designations of Company licensed to Bank under the Bank Licensing Agreement.

"<u>Company Location</u>" means a location that is using Company Licensed Marks in the Program Territory doing business with the public, including an electronic location (such as a Company Website), other direct access media within the Program Territory that is owned or operated by Company or any Franchisee Locations.

"Company Website" means that portion of the worldwide web internet sites operated by Company in support of the Program.

"<u>Confidential Information</u>" has the meaning set forth in <u>Section 11.1(a)</u> (Confidential Information).

"Conversion Date" has the meaning set forth in Section 15.5(a) (Interim Servicing).

"Credit Card Product" means the H&R Block MasterCard Credit Card as described in <u>Schedule D</u> (Credit Card Product Schedule).

"Data Security Requirements" has the meaning set forth in Section 12.2(a) (Data Security).

"Deposit Products" means the individual retirement accounts as described in Schedule E (Deposit Product Schedule).

"Designated Executive" has the meaning set forth in Section 3.1(a)(i) (Designated Executives).

"Disclosing Party" has the meaning set forth in Section 11.2(b) (Limits on Use and Disclosure).

"Dispute" has the meaning set forth in Section 3.2(b) (Dispute Resolution).

"<u>Disputed Program Change Notification</u>" means written notice (that has been delivered in compliance with <u>Section 17.7</u> (Notices)) describing with particularity the Party's basis for disputing a Program Change covered by such notice.

"Distributors" means HRB Tax Group, HRB Enterprises, HRB Eastern Enterprises and HRB Digital.

"<u>Durbin Regulatory Event</u>" shall have occurred if, for any reason, the applicable Financial Products, as currently structured, do not qualify to receive the highest interchange fees permitted for federally insured depository institutions.

"Effective Date" means the Closing Date (as defined in the Purchase Agreement).

"<u>EFS</u>" has the meaning set forth in the Preamble.

"<u>EFS Audit Plan</u>" means an audit plan with respect to the Financial Products maintained by EFS and approved by Bank in the form set forth as <u>Schedule 2.4(f)</u> (EFS Audit Plan), except as otherwise from time to time agreed between the Parties in writing or required by Applicable Law, as determined by Bank in its reasonable discretion.

"EFS Audit Parties" means EFS and any of its Affiliates that are specifically covered by the EFS Audit Plan.

"EFS Corrective Plan" has the meaning set forth in Section 8.1(b) (EFS Audit Rights and Obligations).

"EFS Event of Default" has the meaning set forth in Section 13.1 (EFS Event of Default).

"EFS Matters" has the meaning set forth in Section 4.1(b) (EFS Matters).

"EFS Purchase Option" has the meaning set forth in Section 15.1 (EFS Purchase Options).

"EFS Senior Program Manager" has the meaning set forth in Section 3.1(b)(i) (Program Managers).

"EFS Corrective Plan" has the meaning set forth in Section 8.1(b) (Bank Audit Rights and Obligations).

"EFS Indemnified Parties" has the meaning set forth in Section 16.2 (Indemnification of EFS by Bank).

"<u>EFS Service Provider</u>" means a third-party service provider (including an Affiliate of EFS but excluding Franchisees) used by EFS in connection with the performance of EFS's obligations under this Agreement, including a Material EFS Service Provider. For the avoidance of doubt, a third-party service provider that enters into a tri-party agreement among itself, EFS and Bank is an EFS Service Provider.

"<u>Emerald Advance</u>" means an open-end line of credit offered by Bank under the Program whereby Company Customers may obtain credit as further described in <u>Schedule C</u> (Emerald Advance Product Schedule).

"<u>Emerald Card</u>" means a reloadable, general purpose debit card associated with a demand deposit account offered by Bank to Company Customers under the Program as further described in <u>Schedule A</u> (Prepaid Products Product Schedule).

"Event of Default" means any EFS Event of Default or Bank Event of Default.

"Executive Officer" has the meaning set forth in Section 3.2(b) (Dispute Resolution).

"Exercise Notice" has the meaning set forth in Section 15.1 (EFS Purchase Options).

"Final Wind-Down Date" has the meaning set forth in Section 15.6 (Wind-Down by Bank).

"<u>Financial Products</u>" means the Prepaid Products, Refund Transfer, Emerald Advance, Credit Card Product, and any New Product, in each case offered by Bank and distributed by EFS pursuant to this Agreement.

"FIS" means Fidelity National Information Services, Inc., and its Affiliates.

"Force Majeure Event" has the meaning set forth in Section 17.8(a) (Force Majeure).

"<u>Franchisee</u>" means a Person party to a Franchisee Distribution Agreement with Bank and EFS. No H&R Block franchisee that has not signed a Franchisee Distribution Agreement with Bank may participate in the Program or be considered a Franchisee under this Agreement.

"<u>Franchisee Distribution Agreement</u>" means the Franchisee Financial Products Distribution Agreement in substantially the form attached as Exhibit K to the Purchase Agreement.

"Franchisee Location" means any physical retail office open to the public for the preparation of Returns operated by any Franchisee.

"<u>GLBA</u>" means, collectively, Title V - Privacy of the Gramm-Leach-Bliley Act, P.L. 106-102 and implementing regulations promulgated thereunder, and the standards for safeguarding customer information set forth in 12 CFR Part 364 and 16 CFR Part 314, all as they may be amended, supplemented and/or interpreted in writing from time to time by any federal Regulatory Authority.

"HRB Bank" has the meaning set forth in the Recitals.

"HRB Bank Financial Products" has the meaning set forth in the Recitals.

"HRB Digital" means HRB Digital LLC, a Delaware limited liability company.

"HRB Eastern Enterprises" means H&R Block Eastern Enterprises, Inc., a Missouri corporation.

"HRB Enterprises" means H&R Block Enterprises LLC, a Missouri limited liability company.

"HRB Innovations" means HRB Innovations, Inc., a Delaware corporation, and its successors and assigns.

"<u>HRB Tax Group</u>" has the meaning set forth in the Recitals.

"ICA" has the meaning set forth in Section 15.5(b) (ABA Routing Number; BIN; ICA).

"Indemnification Threshold Amount" has the meaning set forth in Section 16.7(a) (Indemnification Payments).

"Indemnified Losses" means any and all losses, liabilities, costs and expenses of any kind, nature or description imposed or incurred in connection with this Agreement (including reasonable attorneys' fees and expenses, reasonable out-of-pocket costs, interest and penalties), settlements, equitable relief, judgments, damages (including liquidated damages), claims (including counter and cross-claims, and allegations whether or not proven) demands, offsets, defenses, actions, investigations or proceedings by whomsoever asserted (including Regulatory Authorities).

"Indemnified Party" has the meaning set forth in Section 16.3 (Notice).

"Indemnifying Party" has the meaning set forth in Section 16.3 (Notice).

"Intellectual Property" means, on a worldwide basis, any and all: (i) rights associated with works of authorship, including copyrights, moral rights and mask-works; (ii) trademarks and service marks and the goodwill associated therewith; (iii) trade secret rights; (iv) patents, designs, algorithms and other industrial property rights; (v) other intellectual and industrial property rights of every kind and nature, however designated, whether arising by operation of law, contract, license or otherwise; and (vi) applications, registrations, renewals, extensions, continuations, divisions or reissues thereof now or hereafter in force (including any rights in any of the foregoing).

"IP Owner" has the meaning set forth in Section 9.2(a) (Ownership of Intellectual Property).

"<u>Knowledge</u>" means, with respect to EFS the actual and imputed knowledge, after reasonable inquiry, of the Chief Financial Officer of Block, Inc., President or most senior executive officer of EFS, CFO or most senior financial officer of EFS and COO or most senior operations officer of EFS, and with respect to Bank, the actual and imputed knowledge, after reasonable inquiry, of the chief executive officer, the chief financial officer and the chief operating officer of Bank.

"<u>Marketing Deadline</u>" means (a) with respect to Emerald Advance, September 15th of each Program Year, and (b) for all other Financial Products, October 15th of each Program Year.

"<u>Marketing Materials</u>" means all advertisements, brochures, applications, telemarketing scripts, point of purchase displays, packaging, television advertisements, radio advertisements, electronic web pages, electronic web links, and any other type of advertisement, solicitation material, or interactive media developed, launched or distributed for purposes of marketing or promoting the Program, including any and all amendments or modifications thereto, however stored or kept.

"Marketing Templates" has the meaning set forth in Section 2.9(a) (Promotion of Program; Program Marketing Plan).

7

"Material Bank Service Provider" has the meaning set forth in Section 7.2(a) (Service Providers).

"Material EFS Service Provider" has the meaning set forth in Section 7.2(a) (Service Providers).

"New Product" has the meaning set forth in Section 5.6(a) (New Products).

"New Product Offering Plan" has the meaning set forth in Section 5.6(a) (New Products).

"No Interest Notice" has the meaning set forth in Section 15.1 (EFS Purchase Options).

"Nominated Purchaser" has the meaning set forth in Section 15.1 (EFS Purchase Options).

"<u>Non-Surviving Obligations</u>" means the obligations of a Party set forth in this Agreement that do not continue following a Termination Date, which include the obligations under Sections 5.2 and 6.2(a).

"<u>Notified Party</u>" has the meaning set forth in <u>Section 8.4(d)</u> (Reporting).

"<u>Notifying Party</u>" has the meaning set forth in <u>Section 8.4(d)</u> (Reporting).

"<u>OCC</u>" means the Office of the Comptroller of the Currency.

"<u>OFAC</u>" means the Office of Foreign Assets Control.

"Operating Procedures" has the meaning set forth in Section 2.12(a) (Operating Procedures).

"<u>Party</u>" has the meaning set forth in the Preamble.

"<u>Payment Network</u>" means MasterCard, Visa and any other network, including reload networks, as selected by EFS and approved by Bank (such approval not to be unreasonably withheld), facilitating financial transactions through the use of a credit, debit or prepaid product.

"<u>Payment Network Rules</u>" means (i) the applicable bylaws, rules, bulletins, regulations, documentation and manuals promulgated or adopted by each of the Payment Networks, and (ii) any applicable rule or requirement of NACHA - The Electronic Payments Organization, in each case as such rules, manuals and other items may be amended or supplemented from time to time.

"<u>Performance Material Adverse Effect</u>" means, with respect to a Party, any event, change, occurrence or effect that, individually or in the aggregate, has or could be reasonably anticipated to have a material and adverse effect on such Party's ability to perform its duties and obligations under the terms of this Agreement (including the Product Schedules).

"Person" means and includes any individual, partnership, joint venture, corporation, limited liability company, bank, trust, unincorporated organization, government or any department, agency or instrumentality thereof.

"<u>Personally Identifiable Information</u>" means any information that alone, or in combination with other information, relates to a specific, identifiable individual or can be used to identify an individual, including any information defined as "nonpublic personal information" for purposes of GLBA.

"Pilot Program" has the meaning set forth in Section 2.10(b) (Back-Up Account Originator)

"<u>Prepaid Products</u>" means the Emerald Card, employee incentive cards and gift cards co-branded with Company's name as offered by Bank under the Program, in each case as further described in <u>Schedule A</u> (Prepaid Products Product Schedule).

"Privacy Notice" has the meaning set forth in Section 12.1(b) (Privacy Notice).

"<u>Processing Services</u>" means any and all services necessary or convenient for operation of the Program, including the issuance of cards or the processing of transactions or settlements in accordance with applicable Payment Network Rules and Applicable Law. Such services shall include: set-up and maintenance of cardholder accounts, transaction authorization, processing,

clearing and settlement, Payment Network access, cardholder dispute resolution, Payment Network compliance, regulatory compliance, security and fraud control, customer service, collections and activity reporting.

"Processor" means a third-party provider of some or all of the Processing Services, including any EFS Service Provider and any Bank Service Provider.

"<u>Product Schedules</u>" means the product schedules set forth in <u>Schedule A</u> (Prepaid Products), <u>Schedule B</u> (Refund Transfer), <u>Schedule C</u> (Emerald Advance), <u>Schedule D</u> (Credit Card Product), and Schedule E ( Deposit Product).

"Program" has the meaning set forth in the Recitals.

"<u>Program Change</u>" means any change to the terms, pricing, conditions, underwriting or other characteristics of, or the Account Documentation, Marketing Materials, Servicing Materials or any other Program documentation or requirements for, any Financial Product, or any other aspect of the Program or the obligations of EFS or Bank under the Program, made after the date of this Agreement.

"Program Customer" means any Accountholder or any Applicant.

"Program Customer Data" means Accountholder Data and Applicant Data.

"Program Data Site" means a secure content management website containing the Operating Procedures and other documents agreed by the Parties for the Program as set forth in this Agreement.

"<u>Program Eligibility Policy</u>" means the policies, procedures, strategies, guidelines and implementation procedures for the establishment and maintenance of Financial Products and the extension of credit thereunder, if applicable, including all Account eligibility decisioning activities (including application decisioning criteria/procedures, score cutoff strategies, judgmental decisioning policies/procedures and screenings relating to anti-money laundering, Bank Secrecy Act, OFAC and customer identification programs) and Account management activities (including authorizations, line management, delinquency management, fraud monitoring and plastic reissue strategies).

"<u>Program Guidelines</u>" means the policies, procedures and guidelines for the operation of the Program and the distribution of the Financial Products by the Distributors and the Franchisees.

"Program Marketing Plan" has the meaning set forth in Section 2.9(b) (Promotion of Program; Program Marketing Plan).

"<u>Program Material Adverse Effect</u>" means any event, change, occurrence or effect that, individually or in the aggregate, has or could be reasonably anticipated to have a material and adverse effect upon the financial or other performance of the Program as set forth on its pro forma financial statements as developed by the Parties.

"<u>Program Territory</u>" means the United States, Guam, Puerto Rico and certain U.S. military bases outside the United States as identified in <u>Schedule 2.1(a)</u> (Financial Products by Territory).

"<u>Program Year</u>" means the twelve-month period commencing on July 1st of each calendar year (or as applicable, a shorter period for the first Program Year) and ending on June 30th of the following calendar year.

"<u>Prospect Data</u>" means all Personally Identifiable Information regarding a Prospective Customer used by the Parties in connection with determining whom to solicit for the Program.

"Prospective Customer" means Company Customers and any other Persons selected by the Parties to receive Program offers.

"<u>Purchase Agreement</u>" has the meaning set forth in the Recitals.

"Purchase Option Exercise Period" has the meaning set forth in Section 15.1 (EFS Purchase Options).

"<u>Purchase Option Expiration Date</u>" means, with respect to any one or more Product Schedules or the entire Agreement, as applicable, the earliest date upon which one of the following occurs: (a) EFS issues a No Interest Notice, (b) the Purchase Option Exercise Period has expired without EFS issuing an Exercise Notice, (c) the time period for EFS or its designee to enter into a purchase agreement(s) for the purchase of the Purchased Assets and the assumption of the Assumed Liabilities has expired and EFS or its designee to consummate the purchase of the Purchased Assets and the assumption of the Assumed Liabilities under the purchase agreement(s) has expired and EFS or its designee has not consummated such purchase as set forth in <u>Section 15.4</u> (Purchase Mechanics), (d) the time period for EFS or its designee to consummate the purchase of the Purchased Assets and the assumption of the Assumed Liabilities under the purchase agreement(s) has expired and EFS or its designee has not consummated such purchase as set forth in <u>Section 15.4</u> (Purchase Mechanics), (e) the purchase and assumption agreement for the Purchased Assets and the Assumed Liabilities has been terminated by a Party as permitted pursuant to the terms thereof, or (f) EFS (or its permitted designee) otherwise ceases to have the option to purchase the Purchased Assets and assume the Assumed Liabilities.

"Purchase Price" has the meaning set forth in Section 15.3(a) (Determination of Purchase Price).

"Purchased Accounts" has the meaning set forth in the Recitals.

"Purchased Assets" has the meaning set forth in Section 15.1 (EFS Purchase Options).

"<u>Reasonable and Related Outside Counsel Transaction Expenses</u>" has the meaning set forth in <u>Section 15.5(c)</u> (Conversion Costs).

"<u>Receivables Participation Agreement</u>" means the Emerald Advance Receivables Participation Agreement in substantially the form of Exhibit I attached to the Purchase Agreement, governing the ongoing purchases of Emerald Advance Accounts Receivables and other unsecured Accounts Receivables by HRB Participant I, LLC from Bank, which purchases and sales of receivables are intended to be true sales for all purposes.

"Receiving Party" has the meaning set forth in Section 11.2(b) (Limits on Use and Disclosure).

"<u>Refund Transfer</u>" means a banking service offered by Bank under the Program in which Program Customers receive income tax refunds under the Program in a limited-use deposit account via direct deposit from the U.S. Treasury, from which amounts authorized by Program Customers are deducted and as further described in <u>Schedule B</u> (Refund Transfer Product Schedule).

"<u>Regulatory Authority</u>" means, as the context requires, any federal, state or local, domestic, foreign or supranational governmental, regulatory or self-regulatory authority, agency, court,

tribunal, commission or other governmental, regulatory or self-regulatory entity with jurisdiction over a Party, any Distributor or any Franchisee.

"<u>Regulatory Request</u>" has the meaning set forth in <u>Section 11.2(b)</u> (Limits on Use and Disclosure).

"<u>Relevant Audit Portions</u>" has the meaning set forth in <u>Section 8.2(d)</u> (EFS Audit Rights and Obligations).

"<u>Returns</u>" means the federal, state and local income tax returns prepared by Company or any of the Franchisees or by a Person using the Company Website.

"<u>SEC</u>" means the U.S. Securities and Exchange Commission.

"Security Breach" has the meaning set forth in Section 12.2(b) (Data Security).

"Senior Program Manager" has the meaning set forth in Section 3.1(b)(i) (Program Managers).

"Servicing Materials" means any correspondence, documents and reports relating to the customer servicing and collections for the Accounts, including any and all amendments or modifications thereto, however stored or kept.

"Settlement Account" means an account used for settlement of all transactions with respect to the Program.

"Settlement Statement" has the meaning set forth in Section 2.15 (Settlement Statements).

"<u>Solvent</u>" as to a Person, means (a) the present fair salable value of such Person's assets is in excess of the total amount of its liabilities, (b) such Person is presently able generally to pay its debts as they become due, and (c) such Person does not have unreasonably small capital to carry on such Person's business as theretofore operated and all business in which such Person is about to engage. The phrase "present fair salable value" of a Person's assets is intended to mean that value which can be obtained if the assets are sold within a reasonable time in arms'-length transactions in an existing and not theoretical market.

"Specified Party" means those Persons listed in Schedule 15.6 (Block Specified Parties).

"Suspended Product" has the meaning set forth in Section 5.3 (Right to Suspend Offering of Financial Product).

"Tax Season" means the period from November 1st of a given year through April 30th of the following year.

"Term" has the meaning set forth in Section 14.1 (Term).

"<u>Termination Date</u>" means, as applicable, the date on which this Agreement or a Product Schedule terminates or expires in accordance with <u>Article 14</u> (Term and Termination).

"Third Party Guidance" has the meaning set forth in Section 8.5 (OCC 2013-29).

"<u>Transaction</u>" means a card usage that is processed through a Payment Network and its members, or through a Processor, including a load or reload, a deposit, a purchase, a cash withdrawal, or a refund.

"<u>UDAAP</u>" means Sections 1031 and 1036 of the Dodd Frank Wall Street Reform and Consumer Protection Act, 12 USC § 5536, and implementing regulations issued pursuant thereto, and Federal Reserve Regulation AA, which prohibit unfair, deceptive or abusive acts or practices.

"<u>United States</u>" means each of the fifty (50) states or commonwealths comprising the United States of America and the District of Columbia.

1.2 Order of Precedence. This Agreement and the Schedules and Exhibits contain the base terms that govern the relationship between EFS and Bank. In the event any provision of any Schedules (including any Product Schedule) or Exhibits conflicts with a provision of this Agreement, the provision of this Agreement shall control, unless such provision of the Schedule (including any Product Schedule) or Exhibit expressly states that it supersedes a specifically-identified section of this Agreement. To the extent that there are any inconsistencies or conflicts arising between the provisions of this Agreement and any other agreements entered into between Bank and EFS, the provisions of this Agreement shall control unless otherwise expressly provided in such other agreements.

1.3 <u>Rules of Interpretation</u>. Unless otherwise expressly provided in this Agreement or the context otherwise requires, the following rules apply hereto:

(a) the singular includes the plural and the plural includes the singular;

(b) all references to the masculine gender include the feminine gender (and vice versa);

(c) "include," "includes" and "including" are not limiting and are deemed to be followed by the words "without limitation";

(d) references to a particular agreement, instrument or document also refer to and include all renewals, extensions, modifications, amendments and restatements of such agreement, instrument or document;

(e) a reference in this Agreement to an Article, Section, Schedule or Exhibit is to the Article, Section, Schedule or Exhibit of or to this Agreement;

(f) a reference to an Article or Section in this Agreement refers to all sub-parts or sub-components of any such article or section;

(g) words such as "hereunder," "hereto," "hereof," and "herein," and other words of like import refer to the whole of this Agreement and not to any particular section, subsection or clause hereof;

(h) a lower-case reference in this Agreement to a "party" or "parties" includes any Person;

(i) the headings and subheadings of the sections of this Agreement are inserted for convenience of reference only and do not control or affect the meaning or construction of any of the agreements, terms, covenants and conditions of this Agreement in any manner;

(j) a reference to "unreasonably withheld" means "unreasonably withheld, delayed or conditioned;"

(k) any approval, consent or notice required hereunder means "written approval," "written consent" or "written notice," as applicable; and

(1) any reference made in this Agreement to Applicable Law means such Applicable Law as may be amended from time to time, and to any successor Applicable Law relating to the same subject.

(m) Any provision in this Agreement that allows a Party to unilaterally exercise (i) reasonable discretion, (ii) a right of final approval, or (iii) similar decision-making authority, shall automatically include a requirement that, upon exercise of such unilateral right, and a written request by the other Party (signed by a person, and delivered to a person, in each case included in the definition of Knowledge), the Party exercising such unilateral right shall provide a written explanation of the basis for such Party's exercise of such right.

#### ARTICLE 2 PROGRAM DESCRIPTION

2.1 Establishment of the Program.

(a) <u>Commencement of Program; Program Scope</u>. Bank and EFS shall establish the Program within the Program Territory. Specifically, Bank and EFS shall offer Financial Products within the United States under the Program pursuant to the terms and conditions of this Agreement and the Product Schedules. Additionally, Bank and EFS shall offer such Financial Products to qualified consumers in the United States Territories of Puerto Rico, Guam and on certain U.S. military bases as further described in <u>Schedule</u> <u>2.1(a)</u> (Financial Products by Territory) pursuant to the terms and conditions of this Agreement and the Product Schedules.

(b) <u>Account Ownership</u>. Bank shall be the sole owner of the Accounts under the Program. The Parties acknowledge (i) EFS's participation interests as set forth in the Receivables Participation Agreement, and (ii) the EFS Purchase Option.

2.2 Financial Products.

Schedule).

(a) <u>Prepaid Products</u>. A description of the Prepaid Products is set forth in <u>Schedule A</u> (Prepaid Products Product

(b) <u>Refund Transfer</u>. A description of the Refund Transfer product is set forth in <u>Schedule B</u> (Refund Transfer Product Schedule).

(c) <u>Emerald Advance</u>. A description of Emerald Advance product is set forth in <u>Schedule C</u> (Emerald Advance Product Schedule).

(d) <u>Credit Card Product</u>. A description of the Credit Card Product is set forth in <u>Schedule D</u> (Credit Card Product Schedule).

(e) <u>Deposit Product</u>. A description of the Deposit Products is set forth in <u>Schedule E</u> (Deposit Products Schedule).

(f) <u>New Products</u>. Descriptions of New Products shall be set forth in new product schedules, as may be agreed by the Parties from time to time pursuant to <u>Section 5.6</u> (New Products).

2.3 <u>Bank Obligations</u>. Subject to the terms and conditions of this Agreement, Bank shall perform the following tasks, and such other tasks as Bank and EFS shall mutually agree:

(a) exercise final decision-making authority with respect to whether Account Documentation, Marketing Materials and Servicing Materials comply with Applicable Law, as further described in <u>Section 6.1(c)</u> (Account Documentation, Marketing Materials and Servicing Materials);

(b) be willing and able to offer the Financial Products to the markets described in the Product Schedules in volumes no less than the greater of (i) the historical volumes set forth in the Product Schedules or (ii) the projected volumes for the current Tax Season as reasonably determined by the Parties; subject, however, in each case to compliance with Bank's underwriting policies and standards for each Financial Product, and Bank's overall risk management policies;

(c) subject at all times to its compliance obligations under Applicable Law, fulfill all of Bank's obligations hereunder in accordance with the Operating Procedures and the applicable Product Schedules;

(d) maintain all appropriate books and records reflecting Accountholder Data with respect to the Program in accordance with the Operating Procedures;

(e) establish the Program Eligibility Policy in accordance with <u>Section 2.11</u> (Program Eligibility Policy), including establishing credit criteria for the Emerald Advance and Credit Card Product;

(f) review and process Financial Product applications in accordance with the Program Eligibility Policy and this Agreement;

(g) open and administer Accounts in connection with the Financial Products it provides to each Applicant whom Bank has approved for issuance of a Financial Product, including extending credit where applicable on Accounts;

(h) jointly prepare and update the Operating Procedures with EFS pursuant to <u>Section 2.12(a)</u> (Servicing Responsibilities);

(i) supervise, monitor and review the offering of Financial Products at Company Locations as set forth in <u>Section</u> <u>4.4</u> (Bank Oversight);

(j) design, establish and maintain a detailed compliance program as set forth in <u>Section 4.5</u> (Bank Compliance Program);

(k) in its reasonable discretion and in consultation with EFS, develop appropriate training and informational support for Company Location personnel in support of the Program;

(l) jointly review with EFS, and, if approved by Bank, sign, the annual Program Marketing Plan for the marketing and promotion of the Program;

(m) review and, if Bank deems appropriate, approve the EFS Audit Plan and any changes thereto;

(n) comply with Applicable Law, the Operating Procedures and the Payment Network Rules applicable to Bank's conduct of its activities with respect to the Program; and

(o) in consultation with EFS, use commercially reasonable efforts to maintain and enhance the technical and operational systems required to support the Program in a manner that is competitive in the marketplace for financial products and services that are substantially similar to, or substantially competitive with, the Financial Products in the manner in which they are offered to Prospective Customers, all as determined by Bank in its reasonable discretion.

2.4 <u>EFS Obligations</u>. Subject to the terms and conditions of this Agreement, EFS shall be exclusively responsible for the day-to-day management of the Program and shall perform the following tasks and such other tasks as Bank and EFS shall mutually agree:

(a) subject to the oversight of and final approval by Bank as set forth in <u>Section 6.1(b)</u> (Account Documentation, Marketing Materials and Servicing Materials), develop and implement Account Documentation, Marketing Materials and Servicing Materials, and the Operating Procedures;

(b) develop, and jointly review with Bank, and after approval by Bank in Bank's reasonable discretion, sign, the annual Program Marketing Plan for the marketing and promotion of the Program, and then implement the approved Program Marketing Plan in accordance with <u>Section 6.1(a)</u> (Account Documentation, Marketing Materials and Servicing Materials) and the Operating Procedures;

(c) subject at all times to its compliance obligations under Applicable Law, fulfill all of EFS's obligations hereunder in accordance with the Operating Procedures and the applicable Product Schedules, including distribution of all materials developed pursuant to Section 2.4(a) and Section 2.4(b);

(d) maintain all appropriate books and records reflecting Account data with respect to the Program as set forth in the Product Schedules and in accordance with the Operating Procedures;

(e) provide servicing for the Financial Products and Accounts as set forth in the Product Schedules;

(f) maintain and comply in all material respects with the EFS Audit Plan, which is set forth in <u>Schedule 2.4(f)</u>;

(g) comply with Applicable Law, Bank Rules and the Operating Procedures applicable to EFS's conduct of its activities with respect to the Program and as from time to time instructed in writing by Bank, in the exercise its reasonable discretion;

(h) comply with UDAAP for the operation of the Program, as applicable to the Program and/or as from time to time instructed in writing by Bank, in the exercise of its reasonable discretion;

(i) subject to Bank's approval, which approval may be granted or withheld in Bank's reasonable discretion, which shall be exercised in a prompt fashion, and to Bank's continuing oversight, develop and provide appropriate training and informational support for Company Location personnel in support of the Program;

(j) in consultation with Bank, use commercially reasonable efforts to maintain and enhance the technical and operational systems and equipment required to facilitate the

distribution and offering of customer servicing for the Financial Products through Company Locations in a manner that is competitive in the marketplace for financial products and services that are substantially similar to, or substantially competitive with, the Financial Products in the manner in which they are offered to Prospective Customers, all as determined by EFS in its reasonable discretion;

Procedures;

(k) collect, verify and record all required customer identification information in accordance with the Operating

(1) subject to Bank's approval, which approval may be granted or withheld in Bank's reasonable discretion, and Bank's continuing oversight, implement the Program Guidelines to be followed by the Franchisees and the Distributors participating in the Program;

(m) provide Bank with reports of its customer service activities in such frequency and format as mandated by the Operating Procedures, or as Bank may otherwise from time to time designate by written notice in the exercise of its reasonable discretion; and

(n) regularly notify Bank of all consumer or third party complaints received in connection with the Program, and promptly respond to, and resolve, such complaints as instructed by Bank in the exercise of its reasonable discretion. In addition, EFS shall cooperate with Bank in assessing and evaluating the frequency, nature or underlying causes for any consumer complaints, and preventing recurrence thereof.

2.5 <u>Program Economics</u>. The economics and accounting for the Program will be structured on a Product Scheduleby-Product Schedule basis as set forth in each Product Schedule.

2.6 <u>Distributors</u>. EFS and the Distributors will make available the Financial Products offered by Bank to Company Customers as set forth in the Product Schedules. Except as may otherwise be required by Applicable Law, as from time to time determined by Bank in its reasonable discretion, Bank shall make all Financial Products available in Company Locations (including the on-line, digital and DIY channels) owned or operated by the Distributors as provided in the Company Distribution Agreement. The Company Distribution Agreement shall establish the obligations and liabilities of each of Bank, EFS and the Distributors with respect to the other parties thereto. Any responsibilities of EFS with respect to the Distributors are as set forth in the Company Distribution Agreement.

2.7 Franchisees.

(a) Subject to <u>Section 4.4</u> (Bank Oversight), and except as may otherwise be required by Applicable Law, as from time to time determined by Bank in its reasonable discretion, Bank shall make all Financial Products available to each Franchisee in the Program Territory as provided in the Franchisee Distribution Agreement executed by such Franchisee.

(b) Franchisees shall not be deemed third-party beneficiaries of this Agreement.

(c) Each Franchisee Distribution Agreement shall establish the obligations and liabilities of each of Bank, EFS and the Franchisee with respect to the other parties thereto. Any responsibilities of EFS with respect to Franchisees are as set forth in the respective Franchisee Distribution Agreements. Notwithstanding the foregoing, except to the extent Bank is otherwise directed by a Regulatory Authority, or Bank's legal counsel determines that such Franchisee's continued distribution of Financial Products or involvement with the Program is not consistent with

safe and sound banking practices and thus that immediate termination of such Franchisee or the taking of enforcement action under the Franchisee Distribution Agreement is required, Bank shall provide notice to EFS prior to (i) withdrawing any Financial Products from distribution to any Franchisee, or (ii) taking any enforcement actions against a Franchisee under the Franchisee Distribution Agreement.

2.8 <u>Website Supporting the Program</u>. EFS shall be responsible for the establishment of Company Websites in support of and to service the Program, subject to Bank's approval of content related to the Program or the Financial Products to the extent such content describes the Financial Products or is required by Applicable Law (other than Company Applicable Law), as determined by Bank in its reasonable discretion.

2.9 Promotion of Program; Program Marketing Plan.

(a) EFS shall market and support the Program, including complying with its obligations regarding promotion of the Program as set forth in the then-current Program Marketing Plan. EFS acknowledges that any use or distribution of Marketing Materials that may not be in strict compliance with all legal and regulatory requirements may pose regulatory, legal and reputational risks to Bank. In order to protect against any inadvertent noncompliance with said legal and regulatory requirements, Bank shall develop and make available to EFS by September 1 of each Program Year, pre-approved templates of the Marketing Materials (the "<u>Marketing Templates</u>").

(b) Prior to distributing or using any Marketing Materials, EFS shall either:

i. Ensure that the Marketing Materials conform with the then-current version of the Marketing Templates; or

ii. Provide a copy of any Marketing Materials that may not conform with the Marketing Templates to Bank for its review and approval, which approval may be given or withheld in Bank's sole discretion to be exercised in good faith.

(c) Promptly after receipt of any Marketing Materials proposed to be used by EFS, Bank shall give EFS written notice of (i) approval and authorization for implementation of the Marketing Materials as presented, (ii) approval and authorization for implementation of the Marketing Materials as revised by Bank, or (iii) rejection of the Marketing Materials.

(d) Notwithstanding anything to the contrary herein, in no event shall EFS, or any of its Affiliates, Franchisees, Distributors, officers, directors, employees, representatives or agents, use or distribute any Marketing Materials prior to meeting the requirements of either subpart (b)(i) or (b)(ii) of this <u>Section 2.9</u>. It is expressly understood that Bank's review and approval of Marketing Materials shall be for Bank's independent purposes, and shall not (i) constitute an assumption of risk on the part of Bank, (ii) give rise to any claim, action or cause of action by EFS against Bank, or (iii) release EFS from any liability with respect to such Marketing Materials.

(e) Subject to <u>Sections 3.1(a)</u> (Designated Executives) and <u>3.1(b)</u> (Program Managers), EFS will propose, and Bank will (i) review a marketing plan for the upcoming year for the Program, including a marketing calendar for the next Tax Season, forecast of Program transactional volume, and seasonal check-ins (such annual plan, the "<u>Program Marketing Plan</u>"),

and (ii) if Bank so determines in its reasonable discretion, approve the Program Marketing Plan by no later than September 15th of each Program Year.

# 2.10 Back-Up Account Originator.

(a) Provided that it has given Bank at least thirty (30) days' prior written notice (the "<u>Back-Up Notice</u>") thereof, EFS may contract with another federally insured depository institution (the "<u>Back-Up Account Originator</u>") at any time to make prior arrangements for the fulfillment of certain duties (i) after this Agreement or any Product Schedule is terminated or expires or (ii) in the event Bank is unable to perform its obligations under this Agreement or any of the Product Schedules. The Back-Up Notice shall identify the Back-Up Account Originator. EFS may provide the Back-Up Account Originator with access to the Program agreements and such other information as is reasonably necessary to assume Bank's obligations under this Agreement and any of the Product Schedules, provided that such Back-Up Account Originator agrees to confidentiality, data security and privacy covenants in favor of Bank substantially similar to those set forth in <u>Articles 11</u> and <u>12</u>.

(b) Notwithstanding any other provision of this Agreement, during the Term the Back-Up Account Originator and EFS may issue a limited number of Financial Products in one or more pilot programs (each a "<u>Pilot Program</u>") to ensure that the backup system is operational, provided that such Pilot Programs shall not involve substantial promotional efforts, or more than the minimum number of customers reasonably necessary to test the backup system. Moreover, EFS shall take all reasonable steps to ensure that any Pilot Program does not serve to undermine the Program or impugn Bank or its reputation in any manner. In addition, any Pilot Program may be expanded to cover any volume of Financial Products necessary to achieve projected volumes desired by EFS and not agreed to by Bank pursuant to <u>Section 2.3(b)(ii)</u> (Bank Obligations).

2.11 <u>Program Eligibility Policy</u>. Bank shall, in its sole discretion but with such consultation with EFS as Bank reasonably deems appropriate, (a) establish the Program Eligibility Policy for each Financial Product and Account, and (b) provide prior written notice of any material change to the Program Eligibility Policy to EFS.

# 2.12 Servicing Responsibilities.

(a) The Parties agree to, in good faith, jointly develop and implement joint operating agreements that address the operating policies and procedures applicable to the Program (the "<u>Operating Procedures</u>"), provided, however, that in order to ensure full compliance with Applicable Law, Bank shall have a right of final approval with respect to the Operating Procedures. The Parties shall cooperate to review, update and modify the Operating Procedures on an ongoing basis as appropriate. The Parties agree that they shall continue to use such Operating Procedures in substantially the same form throughout the Term unless changes are otherwise agreed in writing by the Parties or are required by Applicable Law as determined by Bank in its reasonable discretion.

(b) Bank shall be responsible for monitoring and updating the Operating Procedures to comply with changes in Applicable Law that relate to Financial Products covered hereunder.



2.13 <u>Service Level Agreements</u>. Except as otherwise provided in a Schedule applicable to any Financial Product, the applicable service level agreements for each of Bank and EFS shall be as described in <u>Schedule F</u> (Service Level Agreements).

2.14 <u>Expenses</u>. Except as expressly set forth in this Agreement or any of the Product Schedules or otherwise agreed by the Parties, each Party shall be responsible for costs associated with its respective obligations under this Agreement.

2.15 <u>Settlement Statements</u>. No later than the fifteenth (15th) day of each month (if a Business Day, or if not, the next Business Day), EFS will provide to Bank one or more monthly settlement statements (each a " <u>Settlement Statement</u>") setting forth:

(a) the total amounts for the month owed to or by EFS, Bank, any participant or other third party pursuant to each Product Schedule and the Receivables Participation Agreement, with line item specificity; and

(b) any other amounts owed to or by EFS, Bank, any participant or other third party, as explicitly provided for herein, in the Receivables Participation Agreement or as otherwise agreed by the Parties in writing, with line item specificity, which amounts may be netted.

Each Settlement Statement shall be prepared and delivered in accordance with the procedures and requirements set forth in the applicable Product Schedule.

2.16 <u>Nevada Office</u>. Bank will (i) maintain its Nevada Branch and will issue and book the Financial Products at the Nevada Branch, (ii) take all reasonable actions at the Nevada Branch necessary for Bank to export Nevada interest rates (and rely upon Nevada usury rates) on the Emerald Advance and other credit products, including the actions specified in the Emerald Advance Product Schedule, and (iii) provide that all Account Documents are subject to and governed by U.S. federal and Nevada law.

2.17 <u>Program Infrastructure</u>. During the Term, EFS shall maintain, operate, or engage in, as the case may be, the following items in support of the Program:

(a) <u>Compliance Management System</u>. EFS will maintain a compliance risk management system, including appropriate and necessary internal controls designed to reasonably ensure that all EFS duties, obligations, and services provided pursuant to the Program, and the actions of any Distributors or Franchisees participating in the Program, will be in compliance with Applicable Law, and conform to the standards set forth in the Agreement. The Program will be substantially in its current form, modified as reasonably requested by Bank, to ensure compliance with Applicable Law.

(b) <u>Quality Control</u>. EFS shall maintain such systems and quality controls as may be necessary or as Bank may reasonably request to (i) enable Bank to adequately monitor the operations of the Program, (ii) react to fraud and promptly, when necessary, respond to and resolve consumer complaints and inquiries so that risk is managed and complaints are reasonably addressed and (iii) provide reports summarizing information as Bank may reasonably require.

(c) <u>Documentation Review and Approval</u>. EFS shall ensure that any of Account Documentation, Marketing Materials, Program Eligibility Policy, Program Guidelines, and Program Documents (including any changes thereto) over which it has ownership and responsibility are in compliance with Applicable Law. EFS shall submit such Marketing Materials, Program Guidelines,

and Program Documents to Bank for review and approval prior to use, such approval not to be unreasonably withheld.

(d) <u>Distributor Relationships.</u> EFS shall maintain Bank approved Company Distribution Agreements with all Distributors, maintain appropriate records with respect to each and every Distributor performing under the Program, and provide reports to Bank with respect to such Distributors in a form and manner as the Bank may reasonably require.

(e) <u>System Access</u>. EFS shall provide Bank with access to all reports and such other data and information as Bank reasonably requests to facilitate settlement, balance, and reconcile accounts, monitor for fraudulent financial transactions and to comply with Bank Secrecy Act and OFAC obligations. Program Manager shall provide summaries in various forms as reasonably requested by Bank in order to monitor performance of EFS.

#### **ARTICLE 3**

#### **PROGRAM SUPPORT**

- 3.1 <u>Communications; Dispute Resolution</u>.
- (a) <u>Designated Executives</u>.
  - i. Each Party shall designate one or more of its senior executives to serve as a high level point of contact to facilitate the efficient operation of the Program (each a "<u>Designated Executive</u>"). The initial Designated Executives are set forth in <u>Schedule 3.1(a)(i)</u> (Initial Designated Executives).
  - ii. The Designated Executives shall communicate as needed to discuss the Program and address any issues that may arise in connection therewith.
- (b) <u>Program Managers</u>.
  - i. Each Designated Executive shall appoint a senior officer to facilitate the overall management of the Program under this Agreement (respectively, the "<u>Bank Senior Program Manager</u>" and "<u>EFS Senior Program Manager</u>", and each, a "<u>Senior Program Manager</u>"). The names of the individuals initially designated as the EFS Senior Program Manager and Bank Senior Program Manager are set forth in <u>Schedule 3.1(b)(i)</u> (Initial Senior Program Managers). Each Senior Program Manager shall have sufficient knowledge and experience to effectively and efficiently perform his or her responsibilities. Each Party shall endeavor to provide stability and continuity in the Senior Program Manager positions.
  - ii. The Senior Program Managers shall review all changes to terms and conditions of the Financial Products offered by Bank as part of the Program, including Bank pricing and Program Eligibility Policy, and to EFS's manner of offering the Financial Products and administering the Program through the Distributors and Franchisees.



- (c) <u>Roles and Responsibilities of Senior Program Managers</u>.
  - i. The Senior Program Managers shall have the responsibility to coordinate, handle and make decisions regarding the day-to-day operations for the Program.
  - ii. In the event of a situation that either Party determines in good faith to be an "emergency," each of the Senior Program Managers shall have the responsibility to promptly notify the other Senior Program Manager and each shall promptly notify the appropriate officers within its respective organization of such emergency.

(d) The Parties shall work together in good faith to facilitate the implementation of the Program, conduct the Program in an efficient manner, provide for appropriate advance planning and such other coordination as shall be necessary or appropriate.

3.2 Dispute Resolution.

(a) In the event of any dispute, controversy, or claim arising out of or relating to this Agreement or the making, construction, interpretation, performance, breach, termination, enforceability or validity thereof (hereinafter, a " <u>Dispute</u>"), the Party raising such Dispute shall provide written notice to the other Party promptly. The Parties shall cooperate and attempt in good faith to resolve any Dispute promptly by negotiating between Persons who have authority to settle the Dispute and who are at a higher level of management than the Persons with direct responsibility for administration and performance of the provisions or obligations of this Agreement that are the subject of the Dispute. Notwithstanding the foregoing, the failure of a Party to promptly provide notice of a Dispute, does not waive any rights of such Party with respect to such Dispute (except to the extent of harm caused by the failure to give prompt notice).

(b) To the extent that Disputes remain unresolved within seven (7) Business Days of being raised, or such other date as agreed between the Chief Financial Officer of Block Inc. and the Chief Executive Officer of Bank, such Dispute shall immediately be referred to the Chief Financial Officer of Block Inc. and the Chief Executive Officer of Bank, or their respective designees (each, an "Executive Officer"), for their review and resolution.

(c) If and only after Executive Officers are unable to reach an agreement within seven (7) Business Days of such referral, the Parties may pursue their respective remedies or exercise their respective rights pursuant to the Program, this Agreement, any of the Product Schedules, any of the Financial Products, or the agreements and transactions contemplated hereby and thereby, including a court action for appropriate remedies, including damages and injunctive relief.

(d) Except as expressly set forth herein, nothing in this <u>Section 3.2</u> shall limit a Party's right to give notice of termination or otherwise pursue its right to terminate this Agreement or any Product Schedule or pursue any other rights set forth in this Agreement or relevant Product Schedule.

(e) Notwithstanding anything to the contrary in this Agreement, with respect to any Financial Product being offered during a Program Year or planned for an upcoming Tax Season, if a Dispute arises regarding a particular Financial Product with respect to the (i) terms and conditions of the then-current offering of such Financial Product, including such Financial Product's features,

Accountholder terms and conditions, functionality or availability compared to those of its prior offering or (ii) manner in which it is being offered or was most recently offered, then either Party may unilaterally elect to have the Dispute immediately escalated to the Executive Officers for review and resolution.

(f) The foregoing provisions of this <u>Section 3.2</u> notwithstanding, each of Bank and EFS shall have the right to take any action that it is advised by outside counsel is required by Applicable Law, or refrain from taking any action that it reasonably determines or is advised by counsel is prohibited by Applicable Law, without first complying with the Dispute resolution procedures set forth herein.

#### **ARTICLE 4**

# DECISION-MAKING AUTHORITY; REGULATORY COORDINATION; COMPLIANCE OBLIGATIONS; BANK OVERSIGHT; BANK COMPLIANCE PROGRAM

4.1 Decision-Making Authority.

(a) <u>Bank Matters</u>. Subject to Applicable Law and the provisions hereunder, Bank shall have decision-making authority with respect to the following matters:

- i. compliance with respect to Applicable Law (other than Company Applicable Law);
- ii. use of Bank Licensed Marks;
- iii. changes to Bank information technology and processing systems that would not be reasonably likely to have a Program Material Adverse Effect;
- iv. compliance with respect to bank-related regulatory matters, including changes required by Applicable Law;
- v. Bank capital expenditures;
- vi. management and retention of Bank personnel;
- vii. establishment of Program Eligibility Policy;
- viii. Financial Product descriptions and Account Documentation required by Applicable Law;

(collectively, the "Bank Matters").

(b) <u>EFS Matters</u>. Except to the extent reserved to Bank in <u>Section 4.1(a)</u>, subject to Applicable Law and the provisions hereunder, EFS shall have decision-making authority with respect to the following matters:

- i. compliance with respect to Applicable Law;
- ii. use of Company Licensed Marks;
- iii. changes to EFS information technology and processing systems that would not be reasonably likely to have a Program Material Adverse Effect;



- iv. compliance with respect to tax preparation-related matters, including changes required by Applicable Law;
- v. EFS capital expenditures;
- vi. management and retention of EFS personnel;
- vii. execution of the Program Marketing Plan in Company Locations;
- viii. marketing design and development of the Program Marketing Plan;
- ix. distribution of Financial Products through Company Locations; and
- x. customer service for the Financial Products and Accounts arranged by EFS consistent with the Product Schedules.

# (collectively, the "EFS Matters").

4.2 <u>Regulatory Coordination</u>. The Parties shall cooperate in (i) analyzing pending material Applicable Law changes and discussing regulatory developments affecting the Program, including any of the Financial Products or Accounts established thereunder, and (ii) subject to Applicable Law regarding the confidentiality of Bank supervisory matters, responding to Regulatory Authorities regarding regulatory-related requests to the Parties.

## 4.3 Compliance Obligations.

(a) Bank shall have final decision making authority with respect to any issues involving compliance of the Program or any aspect thereof with Applicable Law, Bank Rules and Payment Network Rules, including issues arising in connection with disclosure and compliance requirements for Financial Products and Accounts. Subject to the requirements of <u>Section 4.2</u> (Regulatory Coordination), if Bank, as determined by Bank in its reasonable discretion, determines that certain changes to the Program are necessary solely to comply with Applicable Law and Payment Network Rules, the Parties shall work together in good faith with respect to any such changes prior to their implementation.

(b) Except to the extent that any such matters relate to compliance of the Program with Applicable Law as set forth in <u>Section 4.3(a)</u>, EFS shall be solely responsible for decisions regarding compliance with Applicable Law related to EFS and Company operations, including (i) tax preparation and tax-related products and services (other than the Financial Products), and (ii) disclosure, licensing and compliance requirements of EFS and Company related to EFS and Company operations (" <u>Company</u> <u>Applicable Law</u>").

(c) Each Party shall cooperate in a commercially reasonable manner with the other Party in support of and compliance with any Program policies implemented by such Party that are required by Applicable Law.

(d) Each Party shall offer Financial Products and maintain the Accounts in compliance with Applicable Law. In addition to the compliance obligations set forth in this Agreement, any of the Product Schedules or the Program Data Site, EFS shall advise Bank with respect to any Applicable Law with which EFS reasonably believes Bank must comply in connection with the Program, and Bank shall advise EFS with respect to any Applicable Law with which Bank reasonably believes EFS must comply in connection with the Program.

4.4 <u>Bank Oversight</u>. The Parties acknowledge and agree that Bank has the right and the duty to supervise, monitor and review the provision of the Financial Products, Accounts and related services offered under the Program, including at Company Locations. Subject to at least three (3) days' advance notice to EFS and Company Locations (other than with respect to Bank's rights to use "mystery shoppers" pursuant to <u>Section 8.1(a)</u> (Bank Audit Rights and Obligations), for which advance notice shall not be required), Bank and its designated third-party representatives shall have access rights to Company Locations in order to supervise, monitor, review and audit the activities at such Company Locations to ensure that such activities comply with the Operating Procedures, and Applicable Law. EFS shall, as soon as reasonably practicable, take all actions reasonably necessary to remedy any non-compliance identified pursuant to Bank's supervisory rights under this <u>Section 4.4</u>, including causing such remedial actions to be taken at the applicable Company Locations.

## 4.5 Bank Compliance Program.

(a) Bank shall design, establish and maintain a detailed compliance program consistent with this Agreement, to ensure adequate monitoring, supervision and control over EFS and the Program activities that EFS performs for Bank and the Financial Products offered and Accounts maintained by Bank. The compliance program shall include, at a minimum, the following features:

- i. The compliance program shall be reviewed by Bank's board of directors and senior management not less frequently than annually.
- ii. Bank shall designate a compliance officer responsible for the development, implementation and management of Bank's compliance program. The compliance officer shall have responsibility for the oversight of EFS's performance of customer servicing activities related to the Program, the Financial Products and Accounts offered by Bank.
- iii. Not less frequently than annually, Bank shall conduct a compliance risk assessment for the Program. Bank and EFS shall cooperate to develop a true and comprehensive depiction of actual risks in the Program.
- iv. Not less frequently than annually, the Bank compliance officer shall review the compliance program to determine if EFS is operating in accordance with Bank's established policies and procedures regarding the activities relating to the Program, the Financial Products and Accounts offered by Bank.
- v. Bank shall conduct an annual internal or external audit review of the compliance program, which shall include a review and update of the training program and training materials.
- vi. Bank shall require the compliance officer to provide annual written compliance and audit reports to Bank's board of directors. Such reports shall include evidence of appropriate remedial actions taken (or to be taken) to address any identified deficiencies in the compliance program.

- vii. Bank shall develop and maintain a system for tracking and recording consumer complaints regarding the Program in a timely manner. The compliance officer shall provide an annual written report of consumer complaints regarding the Program, and the resolution of such complaints, to Bank's board of directors.
- viii. Bank shall maintain a review process for all Account Documentation, Marketing Materials and Servicing Materials used in the Program.
- ix. Bank shall comply with any other requirements or conditions that a Regulatory Authority deems appropriate for Bank with regard to the Program.
- x. EFS may, with the consent of Bank (such consent not to be unreasonably withheld), implement compliance standards and practices for the Program that supplement, but do not conflict, with those prescribed by Bank.
- xi. EFS may, with the consent of Bank (such consent not to be unreasonably withheld), implement compliance standards and practices for the Program that implement legal stipulations, settlements and contractual agreements with third parties.

## ARTICLE 5 EXCLUSIVITY; NEW PRODUCTS; CHANGES TO EXISTING PRODUCTS AND PROGRAM

5.1 <u>Bank's Right to Offer Financial Products to Others</u>. Except as set forth in <u>Section 6.2(b)</u> (Cross Marketing) and <u>Article 12</u> (Privacy and Data Security), nothing in this Agreement is intended or shall be construed to prohibit or limit Bank's right to offer financial products and services that are similar to the Financial Products, with or through any Person.

5.2 EFS to Offer Only Financial Products of Bank.

(a) At Company Locations, EFS and its Affiliates will offer and distribute only Bank's Financial Products, and will not offer or distribute financial products issued or originated by other insured depository institutions that are substantially similar to the Financial Products, except:

i. as otherwise provided in this Agreement;

ii. for financial products issued or originated after the termination of this Agreement or, with respect to a particular financial product, after the termination of the Product Schedule for the substantially similar Bank Financial Product;

iii. EFS and its Affiliates shall have the right to offer a "second look" program with a different set of eligibility criteria for selected Financial Products for which Bank has declined to offer based on those eligibility criteria; and

iv. any entity that becomes an Affiliate of Company after the date of this Agreement may continue to offer financial products of another insured depository institution that are substantially similar to the Financial Products, if such Affiliate was

offering such financial products prior to such Affiliate being acquired by, or otherwise becoming an Affiliate of, Company.

(b) Nothing in this Agreement is intended or shall be construed to prohibit or limit the rights of EFS and its Affiliates to offer or distribute any other financial product or service that is not substantially similar, in all respects, to a Financial Product.

5.3 <u>EFS Right to Suspend Offering of Financial Product</u>. Notwithstanding any other provision of this Agreement, EFS and its Affiliates shall have the right, in their sole discretion, to suspend the offering of one or more Financial Products (a "<u>Suspended Product</u>") in one or more states (or other jurisdictions) without terminating the relevant Product Schedules or triggering a termination right under this Agreement; provided, however, that following any such suspension if EFS elects to resume offering any such Suspended Product during the Term, EFS shall offer such Suspended Product only through Bank.

5.4 <u>Enhancements to Financial Products</u>. Bank and EFS may agree upon enhancements or other changes to the Financial Products offered under the Program. The Parties shall periodically meet to discuss new features or functionalities to existing Financial Products under the Program. After approval by both Parties, the offering of such new features or functionalities to existing Financial Products shall be implemented by a written amendment of the relevant Product Schedule(s).

5.5 <u>Changes to Program During Tax Season</u>. Notwithstanding any other provision of this Agreement or any Product Schedule:

(a) Unless otherwise permitted under this Agreement, any changes proposed for the Program shall require the prior consent of Bank and EFS. Bank shall use commercially reasonable efforts in good faith to delay any changes to Financial Products until the conclusion of the Tax Season. Bank may not require that EFS make changes to or cease offering any Financial Product during a Tax Season, unless Bank has received a specific written directive from its primary federal regulator that changes to, or cessation of the offering of, a Financial Product are required, provided, however, that before requiring EFS to cease offering a Financial Product during the Tax Season, Bank shall make reasonable efforts to work with EFS to make changes to the Financial Product or the distribution of such product to address regulatory concerns.

(b) If there are any potential material changes to the Program of which Bank is aware or contemplates implementing, Bank shall promptly notify EFS of such potential material changes.

5.6 New Products.

(a) EFS and Bank may agree to offer a new financial product or service by developing, negotiating and executing a plan to offer the new product (a "<u>New Product Offering Plan</u>") and a new product schedule (when so finalized, a "<u>New Product</u>"). Upon execution of the new product schedule and the New Product Offering Plan, the New Product will become a Financial Product and the product schedule will become a Product Schedule, covered by the Program and subject to the terms of this Agreement.

(b) Nothing in this Agreement is intended or shall be construed to require (i) Bank to offer or distribute any proposed new product developed by EFS; (ii) Bank to offer or distribute

through EFS and its Affiliates any proposed new product that Bank develops; (iii) EFS to give Bank a first look or right of first refusal on any proposed new product developed by EFS; or (iv) Bank to give EFS a first look or right of first refusal to distribute or service any proposed new product developed by Bank.

5.7 <u>Staffing Plan</u>. Bank shall develop and implement a Staffing Plan designed to ensure adequate support of the Program, including the compliance obligations associated therewith, which is acceptable to Bank's regulators. Bank shall make each of Bank's employees or independent contractors assigned to assist on matters relating to or in connection with the Program or this Agreement aware of the confidentiality provisions of <u>Section 11.2</u> (Limits on Use and Disclosure) (unless such contractors are subject to independent duties of confidentiality).

# **ARTICLE 6**

# ACCOUNT DOCUMENTATION; MARKETING; CROSS MARKETING

## 6.1 Account Documentation, Marketing Materials and Servicing Materials.

(a) EFS shall be responsible for implementing the Program Marketing Plan consistent with Company's tax services marketing plan and the terms of this Agreement and will review the Program Marketing Plan, Marketing Materials and Marketing Template with Bank.

(b) EFS shall be responsible for developing, modifying, and implementing the Account Documentation, Marketing Materials, and Servicing Materials, subject to final approval by Bank, including approval with respect to any content (i) required by Applicable Law, or (ii) containing Bank Licensed Marks, unless use is in a format preapproved by Bank. Material changes to Account Documentation, Marketing Materials, and Servicing Materials by EFS shall require the prior approval of Bank, as soon as practicable and in no event later than five (5) Business Days after notice, and shall be paid for in their entirety by EFS. EFS shall administer Accounts in compliance with Account Documentation, Applicable Law, Bank Rules and Payment Network Rules, and clearly disclose in any cardholder agreement all fees to be paid by Accountholders and/or Applicants. All dormant Accounts will be processed in accordance with Applicable Law, including relevant escheat and abandonment statutes.

(c) Bank shall have final decision-making authority with respect to Account Documentation, Marketing Materials, Marketing Templates and Servicing Materials and shall have the right to require changes to those materials as required by Applicable Law or relating to any changes in Bank Licensed Marks. Bank has the right to set, and make changes to, the terms of the Financial Products, including interest rates and fees for the Financial Products. Any material changes to Account Documentation, Marketing Materials, Marketing Templates and Servicing Materials by Bank that (i) are not required by Applicable Law or (ii) do not relate to any changes in Bank Licensed Marks shall: (x) require the prior approval of EFS as soon as practicable and in no event later than five (5) Business Days after notice, such approval not to be unreasonably withheld and (y) be implemented by EFS within thirty (30) days of EFS approval. Notwithstanding Section 6.1(b), any material changes to Account Documentation, Marketing Templates and Servicing Materials made by Bank without the concurrence of EFS that (i) are not required by Applicable Law or (ii) which relate to changes in Bank Licensed Marks, shall be paid for in their entirety by Bank. Solely for purposes of the preceding sentence, "required by Applicable Law" means as reasonably determined by Bank upon advice of outside counsel.



#### 6.2 Cross Marketing.

(a) Without obligating either Party, EFS and Bank agree to explore ways to cross-market their respective products to the other Party's customers on terms satisfactory to each of the Parties, in their sole discretion, and appropriately documented. Any such marketing shall in all cases be subject to Applicable Law, including Sections 6713 and 7216 of the Internal Revenue Code of 1986 (the "<u>Code</u>") and GLBA, and also subject to the ownership and rights to customer data provisions of this Agreement under <u>Section 12.3</u> (Collection, Ownership and Use of Program Customer Data).

(b) Except as expressly authorized in this Agreement or otherwise agreed in writing by EFS:

i. Bank shall not use Applicant Data, Prospect Data, Company Customer Data and/or Accountholder Data to do any of the following: (A) solicit Applicants, Prospective Customers or Program Customers for product or service offerings (including offerings by third parties); or (B) use or disclose the names of Applicants, Prospective Customers or Program Customers, or any other information relating to the Accounts or to the Applicants, Prospective Customers or Program Customers; and

ii. Bank shall not insert any Bank or any third-party's offerings in statements provided to Program Customers or target Company Customers for Bank products and services (other than the Financial Products) or offerings by third parties. For the avoidance of doubt, any Applicant Data, Prospect Data, Company Customer Data and/or Accountholder Data that Bank obtains in connection with the Program may only be used for purposes of fulfilling its obligations and exercising its rights under this Agreement and the Program.

(c) EFS may use Program Customer Data, and other information relating to Program Customers, to the fullest extent permitted by Applicable Law and the Privacy Notice.

(d) Notwithstanding the restrictions set forth in <u>Section 6.2(b)</u>:

i. Bank may make solicitations for goods and services to the public in its own name, and may use prospect lists it develops independently of the Program or that are provided by third parties, which may include the names of one (1) or more Program Customers; provided that Bank does not (A) target such solicitations to Program Customers or Company Customers, (B) other than as expressly authorized by this Agreement, use or permit a third-party to use any list of or information regarding Program Customers obtained by Bank pursuant to its activities related to the Program, or (C) refer to or otherwise use the name of Company or EFS in connection with such solicitation; and

ii. Bank shall not be obligated to redact the names of Program Customers from general marketing lists it develops independently of the Program or acquires from third parties (e.g., subscription lists) that it uses for solicitations.

iii. Bank shall be permitted to engage in business with any Person who contacts it independently of activities involving the Financial Products.

6.3 <u>Customers of Bank</u>. For all purposes hereof, Accountholders are, and for all intents and purposes shall be deemed to be, customers of Bank, having a direct relationship with Bank. Bank shall be free to contact Accountholders and potential Accountholders regarding matters specifically related to their existing Accounts, without notice to EFS, and at all times in compliance with this Agreement, including the restrictions in <u>Section 6.2(b)</u> (Cross Marketing) and in <u>Article 12</u> (Privacy and Data Security).

# ARTICLE 7 OUTSOURCING RESTRICTIONS; SERVICE PROVIDERS

# 7.1 Outsourcing Restrictions.

Neither Party shall, without the written consent of the other Party, outsource any servicing functions relating to the Accounts that would result in Program Customer Data being transmitted outside of the United States; provided, however, EFS shall be permitted to utilize the servicing functions of the service providers located outside the United States as set forth in <u>Schedule 7.1</u> (List of Internationally Outsourced Service Providers).

## 7.2 Service Providers.

(a) The material third-party service providers for each of Bank and EFS ("<u>Material Bank Service Providers</u>" and "<u>Material EFS Service Providers</u>," respectively) are set forth in <u>Schedule 7.2(a)</u> (Material Third-Party Service Providers). Modification of the engagement by Bank of Material Bank Service Providers or EFS of Material EFS Service Providers, as applicable, for the Program shall be subject to the other Party's prior notice and consultation rights; provided, however, neither Party shall change or add a Material Bank Service Provider or Material EFS Service Provider, as applicable, without prior written notice to and consultation with the other Party.

(b) The use by a Party of Bank Service Providers or EFS Service Providers, as applicable, to perform services related to this Program shall not relieve such Party of any of its obligations under this Agreement, and for the avoidance of doubt, such Bank Service Providers or EFS Service Providers, as applicable, shall be held to the same standards of care as would be applicable to such Party if it were to perform the service itself. The Party engaging any Bank Service Providers or EFS Service Providers, as applicable, will be responsible for all payments to such Bank Service Providers or EFS Service Providers, as applicable, and shall be responsible for (i) ensuring the performance or non-performance of such Bank Service Providers or EFS Service Providers, as applicable, as if such performance or nonperformance were that of such Party and (ii) requiring such Bank Service Providers or EFS service Providers or E

(c) Neither Party shall make material changes to the services provided under agreements with Material Bank Service Providers or Material EFS Service Providers in connection with the Program without prior written notice to and consultation with the other Party other than pricing changes for which Bank or EFS, as applicable, is responsible.

(d) Except as otherwise agreed in writing among Bank, EFS and any third-party service provider, each Party shall continue to manage all third-party relationships managed by such Party as of the Effective Date and deemed necessary and appropriate for such Party to perform its obligations pursuant to this Agreement and the Product Schedules. EFS shall continue to be entitled to all incentives and benefits arising from servicing contracts with third parties, including its existing incentive agreement with MasterCard.

## ARTICLE 8 AUDIT RIGHTS; REPORTING

#### 8.1 Bank Audit Rights and Obligations.

In addition to the other rights set forth in this Agreement, subject to the confidentiality provisions set forth in (a) Article 11 (Confidentiality), applicable privacy laws and other Applicable Law, the EFS Audit Parties and service providers for which EFS and Bank have agreed to enter into tri-party agreements shall, during normal business hours, in a manner designed to be least disruptive and no more than once per Program Year unless more frequent audits are required by any Regulatory Authority or are necessary to confirm that prior audit exceptions have been rectified: (i) permit Bank, its officers, employees, accountants, lawyers and consultants in such a manner as to minimize unreasonable interference with the EFS Audit Parties' normal business operations, to, by any means reasonably acceptable to Bank, examine and audit operations and audit, inspect, copy and make copies of all of the data, records, files and books of account (including non-financial information) under the control of the EFS Audit Parties if such operations, data, records, files and books of account relate to any obligation of EFS under this Agreement or any of the Product Schedules, including any calculation required to be made pursuant to the terms of this Agreement or the Product Schedules, as applicable, and as required by Applicable Law or Payment Network Rules; and (ii) use commercially reasonable efforts to facilitate Bank's exercise of such right granted in subparagraph (i) of this Section 8.1(a) (including a good faith effort to obtain any consents that may be necessary or desirable to avoid a breach of any contractual obligations). In furtherance of and without limiting the foregoing, (x) the EFS Audit Parties shall permit Bank to use "mystery shoppers" at Company Locations to audit the offering of the Financial Products under the Program; (y) the EFS Audit Parties shall permit Bank to examine and audit operations and audit, inspect, copy and make copies of all of the data, records, files and books of account (including non-financial information) at Company Locations if such operations, data records, files and books of account relate to any obligation of EFS under this Agreement or any of the Product Schedules.

(b) To the extent an audit conducted pursuant to <u>Section 8.1(a)</u> reveals any error, deficiency or other failure to perform on the part of EFS, EFS will (i) as soon as reasonably possible following the date on which it becomes aware of such error, deficiency or other failure to perform and, in any event, no later than fifteen (15) Business Days following such date (unless a shorter timeframe is reasonably deemed necessary by Bank because of the critical nature of the error, deficiency or other failure or is required by a Regulatory Authority), deliver to Bank a written corrective plan (an "<u>EFS Corrective Plan</u>") that, if followed, is designed to correct the error, deficiency or other failure to perform, (ii) following the approval of the EFS Corrective Plan by Bank, promptly execute the EFS Corrective Plan and (iii) permit Bank to conduct additional follow-up audits as Bank may deem reasonably necessary for Bank to audit EFS's compliance with this provision, including the correction of EFS's error, deficiency or other failure to perform. The EFS

Audit Parties shall use commercially reasonable efforts to deliver any document or instrument necessary for Bank to obtain such information from any Person maintaining records for the EFS Audit Parties. The reasonable cost and expense of any such follow-up audits shall be expenses of EFS.

(c) Except as otherwise required for Bank to be in full compliance with Applicable Law, notwithstanding anything to the contrary contained herein, EFS shall not be required to provide Bank or any other Person with access to information or records to the extent that such access (i) is prohibited by Applicable Law or Payment Network Rules; provided, however, that to the extent that access to information or records is so prohibited, EFS (A) shall notify Bank in writing regarding the Applicable Law or Payment Network Rules which prohibit such access and (B) shall deliver to Bank copies of all requested information or records, redacted as may be necessary to comply with the cited Applicable Law or Payment Network Rules or (ii) would reasonably be expected to cause EFS to be a consumer credit reporting agency as set forth in the Fair Credit Reporting Act.

(d) EFS shall use commercially reasonable efforts to facilitate the maximum level of access by Bank, its officers, employees, accountants, lawyers and consultants in light of constraints under Applicable Law and Payment Network Rules. No action taken by (or on behalf of) Bank pursuant to this <u>Section 8.1</u> shall diminish or obviate any of the representations, warranties, covenants or agreements of EFS contained herein.

(e) The reasonable out of pocket costs and expenses of annual examinations or audits conducted by (or on behalf of) Bank pursuant to Section 8.1(a) shall be the expenses of Bank (other than costs incurred by EFS in responding to the needs of Bank or such other Person performing the audit on behalf of Bank).

(f) It is expressly understood that Bank's exercise of its audit rights hereunder, or the granting thereof herein, shall be for Bank's independent purposes, and shall not constitute an assumption of risk on the part of Bank or release EFS from any liability hereunder or under Applicable Law.

8.2 EFS Audit Rights and Obligations.

(a) In addition to the other rights set forth in this Agreement, subject to the confidentiality provisions set forth in <u>Article 11</u> (Confidentiality), applicable privacy laws and other Applicable Law, Bank shall, during normal business hours, in a manner designed to be least disruptive, and no more than once per Program Year unless more frequent audits are required by any Regulatory Authority or are necessary to confirm that prior audit exceptions have been rectified: (i) permit EFS, its officers, employees, accountants, lawyers and consultants in such a manner as to minimize unreasonable interference with Bank's normal business operations, to, by any means reasonably acceptable to EFS, examine and audit operations and audit, inspect, copy and make copies of all of the data, records, files and books of account (including non-financial information) under the control of Bank if such operations, data, records, files and books of account to the terms of this Agreement or the Product Schedules, as applicable, and as required by Applicable Law or Payment Network Rules; and (ii) use commercially reasonable efforts to facilitate EFS's exercise of such right granted in subparagraph (i) of this <u>Section 8.2(a)</u>

(including a good faith effort to obtain any consents that may be necessary or desirable to avoid a breach of any contractual obligations). In furtherance of and without limiting the foregoing, Bank shall permit EFS to examine and audit operations and audit, inspect, copy and make copies of all of the data, records, files and books of account (including non-financial information) at Bank if such operations, data records, files and books of account relate to any obligation of Bank under this Agreement or any of the Product Schedules. The Parties acknowledge and agree that that EFS's audit rights do not extend to Bank's regulatory reports of examination, regulatory communications or other documents or materials that Bank is prohibited by Applicable Law from sharing with EFS.

(b) To the extent an audit conducted pursuant to <u>Section 8.2(a)</u> reveals any error, deficiency or other failure to perform on the part of Bank, Bank will (i) as soon as reasonably possible following the date on which it becomes aware of such error, deficiency or other failure to perform and, in any event, no later than fifteen (15) Business Days following such date (unless a shorter timeframe is reasonably deemed necessary by EFS because of the critical nature of the error, deficiency or other failure or is required by a Regulatory Authority), deliver to EFS a written corrective plan that, if followed, is designed to correct the error, deficiency or other failure to perform (a "<u>Bank Corrective Plan</u>"), (ii) following the approval of the Bank Corrective Plan by EFS, promptly execute the Bank Corrective Plan and (iii) permit EFS to conduct additional follow-up audits as EFS may deem reasonably necessary for EFS to audit its compliance with this provision, including the correction of its error, deficiency or other failure to perform. Bank shall use commercially reasonable efforts to deliver any document or instrument necessary for EFS to obtain such information from any Person maintaining records for Bank.

(c) Except as otherwise required for Bank to be in full compliance with Applicable Law, notwithstanding anything to the contrary contained herein, Bank shall not be required to provide EFS or any other Person with access to information or records to the extent that such access (i) is prohibited by Applicable Law or Payment Network Rules; provided, however, that to the extent that access to information or records is so prohibited, Bank (A) shall notify EFS in writing regarding the Applicable Law or Payment Network Rules which prohibit such access and (B) shall deliver to EFS copies of all requested information or records, redacted as may be necessary to comply with the cited Applicable Law or Payment Network Rules; or (ii) would reasonably be expected to cause Bank to be a consumer credit reporting agency as set forth in the Fair Credit Reporting Act.

(d) The EFS Audit Parties shall conduct audits in accordance with, and on the schedule required by, the EFS Audit Plan. The EFS Audit Parties shall provide to Bank the portions of such audits relating to the Program and the Financial Products (the "<u>Relevant Audit Portions</u>"). The reasonable cost and expense of the EFS Audit Parties providing the Relevant Audit Portions to Bank shall be at EFS's expense. The EFS Audit Parties shall deliver to Bank (i) any Relevant Audit Portion relating to the offering of Emerald Advance no later than December 31 of each Program Year relating to pre-season Emerald Advance, through February 15 of the Tax Season no later than March 31 of each Program Year, and (iii) any Relevant Audit Portion relating to the offering of the Financial Products after February 15 no later than June 30 of each Program Year; provided, however, that if any audit conducted by the EFS Audit Parties pursuant to the first sentence of this <u>Section 8.2(c)</u> reveals any adverse material findings, the EFS

Audit Parties must provide to Bank interim reports relating to such adverse material findings within five (5) Business Days of EFS's receipt of such audit.

(e) Bank shall use commercially reasonable efforts to facilitate the maximum level of access by EFS, its officers, employees, accountants, lawyers and consultants in light of constraints under Applicable Law and Payment Network Rules. No action taken by (or on behalf of) EFS pursuant to this <u>Section 8.2</u> shall diminish or obviate any of the representations, warranties, covenants or agreements of Bank contained herein.

(f) The reasonable cost and expense of annual examinations or audits conducted by (or on behalf of) EFS pursuant to Section 8.2 or the EFS Audit Plan shall be expenses of EFS.

(g) It is expressly understood that EFS's exercise of its audit rights hereunder, or the granting thereof herein, shall be for EFS's independent purposes, and shall not constitute an assumption of risk on the part of EFS or release Bank from any liability hereunder or under Applicable Law.

8.3 <u>Audits by Regulatory Authorities</u>. In addition to access as provided in <u>Sections 8.1</u> (Bank Audit Rights and Obligations) and <u>8.2</u> (EFS Audit Rights and Obligations), each Party, at its own expense, will permit, in accordance with Applicable Law, any Regulatory Authority to visit its facilities related to the Program. Each Party will also permit, in accordance with Applicable Law, any Regulatory Authority to review and obtain copies of the books and records in its possession or under its control relating to the Program. The access granted under this <u>Section 8.3</u> shall occur during normal business hours with reasonable advance notice unless otherwise required by any such Regulatory Authority and Applicable Law.

## 8.4 Reporting.

(a) To the extent permitted by Applicable Law, each Party shall have the right to require the other Party upon ten (10) days' prior written notice or such shorter timeframe as may be required at the request of a Regulatory Authority, to provide reasonably necessary internal and external reports (including all data, electronic or otherwise, forming the basis for such reports) requested in good faith with respect to each of the Financial Products under the respective Product Schedules in a timely manner to it from time to time, including reporting sufficient for each Party to satisfy its reporting compliance obligation under Applicable Law (such as reconciliation reports, fraud reports and complaint reports). The Parties shall provide the reports as set forth in the Program Data Site and as otherwise agreed in writing by the Parties.

(b) EFS shall cause Bank to have direct access to management reports available to EFS from Material EFS Service Providers for the Financial Products. Each of EFS and Bank shall provide to the other Party such additional management reports, including reporting related to fraud losses, suspicious activity and consumer complaints relating to the Program. Upon a Party's reasonable request, the other Party shall also provide any customized reports related to the Program from time to time, to the extent reasonably practicable.

(c) In the event that a Party requests additional reports or information that will require the other Party to incur reasonable additional out-of-pocket costs, the Party preparing the report shall be required to provide such requested reports or information. The requesting Party shall pay the other Party for such additional out-of-pocket costs unless the requesting Party has made

such request based on a good faith determination that such additional report or information is required for purposes of satisfying its regulatory obligations.

(d) Each Party will cooperate with the other Party in a commercially reasonable manner to build any necessary system interfaces with respect to each of the Financial Products under the respective Product Schedules, including Program and Financial Product transaction and Account level data.

8.5 <u>SEC Reporting and Public Announcements</u>. The Parties will work in good faith to coordinate public disclosures that reference the Financial Products, the Program or the other Party. The Parties shall mutually agree on the content and timing of press releases and Form 8-Ks announcing the execution of this Agreement.

To the extent permitted by Applicable Law, each Party (the "<u>Notifying Party</u>") shall provide the other Party (the "<u>Notified Party</u>") with advance notice and copies of all relevant portions of the Notifying Party's anticipated regulatory filings and other public disclosures required by U.S. securities laws that reference the Financial Products or the Program, or that mention the Notified Party's name, prior to the public disclosure or filing thereof. The Notified Party shall have the opportunity to review with its counsel and provide comments to the Notifying Party on such disclosures and filings prior to their public disclosure or filings with the regulators. The Notified Party's comments and consider incorporating such comments into its regulatory filings and other public disclosures required by U.S. securities laws prior to their filing or public disclosure; provided that, notwithstanding anything in this Agreement to the contrary, a Notifying Party shall at all times control and be responsible for the content and timing of its securities law filings and public disclosures and be permitted to make any disclosures that it reasonably believes are required under Applicable Law.

To the extent permitted by Applicable Law and to the extent not already covered by the above paragraph, each Party should provide the other Party with advance notice and copies of all relevant portions of any press releases, announcements, similar materials and the planned text of other statements of a public nature that reference the Financial Products or the Program, or that mention the Notified Party's name, prior to the public disclosure or filing thereof. The Notified Party shall have the opportunity to review and provide comments to the Notifying Party on such materials prior to their public disclosure or filing. The Notifying Party shall in good faith discuss and consider the Notified Party's comments and consider incorporating such comments into its communications prior to their public disclosure or filing; provided that, notwithstanding anything in this Agreement to the contrary, a Notifying Party shall at all times control and be responsible for the content and timing of its communications and be permitted to make any such communications.

Nothing contained in this section 8.5 shall prevent either Party from (i) publicly discussing general plans, forecasts or other materials that do not reference Financial Products, the Program or the other Party or (ii) issuing press releases, announcements, similar materials or communications or making other statements consistent with content previously shared with the other Party pursuant to this Section 8.5 or otherwise in public domain (other than as a result of a violation of this Section 8.5 by the Party desiring to make the disclosure).

8.6 <u>OCC 2013-29</u>. The Parties acknowledge that the relationships contemplated hereunder fall within the purview of Risk Management Guidance OCC 2013-29, issued by the OCC

on October 30, 2013 (the "<u>Third Party Guidance</u>"). EFS has provided and shall continue to promptly provide to Bank such information regarding itself, its Affiliates, and the EFS Service Providers as Bank may from time to time reasonably request, in order to ensure compliance with Bank's obligations as set forth in the Third Party Guidance.

8.7 <u>OCC Oversight</u>. EFS acknowledges that (i) the performance of activities by external parties for Bank is subject to OCC examination oversight, including access to all work papers, drafts, and other materials; (ii) the OCC treats, as subject to 12 USC 1867(c) and 12 USC 1464(d)(7), situations in which a bank arranges, by contract or otherwise, for the performance of any applicable functions of its operations; and (iii) the OCC has the authority to examine and to regulate the functions or operations performed or provided by third parties.

# ARTICLE 9

# INTELLECTUAL PROPERTY; LICENSE TO USE MARKS; OWNERSHIP RIGHTS

9.1 <u>Licensing Agreements</u>. On the date hereof, (a) Bank is entering into the Bank Licensing Agreement with certain subsidiaries of Block, Inc. and (b) HRB Innovations is entering into the Block Licensing Agreement with Bank.

## 9.2 Ownership and Licenses of Intellectual Property.

(a) <u>Ownership of Intellectual Property</u>. Each Party and HRB Innovations, Inc. shall continue to own all of their respective Intellectual Property that existed as of the Effective Date. Each Party and HRB Innovations, Inc. also shall own all right, title and interest in the Intellectual Property each of them develops independently of the other Party or HRB Innovations, Inc., as applicable, during the Term. To the extent a Party (the "<u>Acquiring IP Party</u>") acquires any rights in or to such Intellectual Property of the other Party or HRB Innovations, Inc. (the "<u>IP Owner</u>"), the Acquiring IP Party hereby assigns all such right, title and interest in and to such Intellectual Property back to the IP Owner. Acquiring IP Party shall execute any documents in connection with such assignment that IP Owner may reasonably request.

(b) <u>Joint Intellectual Property</u>. All Intellectual Property jointly developed by the Parties in connection with the Program shall be owned by EFS.

## ARTICLE 10 REPRESENTATIONS, WARRANTIES AND COVENANTS

## 10.1 EFS's Representations, Warranties and Covenants.

To induce Bank to establish the Program, EFS makes the following representations, warranties and covenants to Bank, each and all of which shall survive the execution and delivery of this Agreement, and each and all of which shall be deemed to be restated and remade with the same force and effect on each day of the Term, except as otherwise stated.

(a) <u>Corporate Existence</u>. EFS: (i) is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Delaware, (ii) is duly licensed or qualified to do business as a limited liability company and is in good standing in all jurisdictions in which the nature of the activities conducted or proposed to be conducted by it or the character of the assets owned or leased by it makes such licensing or qualification necessary to perform its obligations required hereunder, and (iii) has all necessary licenses, permits, consents or approvals from or by, and has made all necessary notices to, all Regulatory Authorities, to the extent required

for EFS's business, and for the performance of its obligations pursuant to this Agreement, except where the failure to have such licenses, permits, consents or approvals would not have a Performance Material Adverse Effect with respect to EFS or a Program Material Adverse Effect.

(b) <u>Capacity: Authorization; Validity.</u> EFS has all necessary corporate power and authority to execute and enter into this Agreement, and perform the obligations required of EFS hereunder and the other documents, instruments and agreements relating to the Program and this Agreement executed by EFS pursuant hereto. The execution and delivery by EFS of this Agreement and all documents, instruments and agreements executed and delivered by EFS pursuant hereto, and the consummation by EFS of the transactions specified herein have been duly and validly authorized and approved by all necessary corporate action of EFS. This Agreement (i) has been duly executed and delivered by EFS, (ii) constitutes the valid and legally binding obligation of EFS, and (iii) is enforceable in accordance with its terms (subject to applicable bankruptcy, insolvency, reorganization, receivership or other laws affecting the rights of creditors generally and by general equity principles including those respecting the availability of specific performance).

(c) <u>Conflicts; Defaults; Etc</u>. The execution, delivery and performance of this Agreement by EFS, its compliance with the terms hereof, and its consummation of the transactions specified herein will not:

i. conflict with, violate, result in the breach of, constitute an event which would, or with the lapse of time or action by a third-party or both would, result in a default under, or accelerate the performance required by, the terms of any material contract, instrument or agreement to which EFS is a party or by which it is bound, or by which EFS assets are bound, except for conflicts, breaches and defaults which would not have a Performance Material Adverse Effect with respect to EFS;

ii. conflict with or violate the articles of incorporation or by-laws, or any other equivalent organizational documents, of EFS;

iii. violate any Applicable Law or conflict with, or require any consent or approval under any judgment, order, writ, decree, permit or license, to which EFS is a party or by which it is bound or affected, except to the extent that such violation or the failure to obtain such consent or approval would not have a Performance Material Adverse Effect with respect to EFS;

iv. require the consent or approval of any other party to any contract, instrument or commitment to which EFS is a party or by which it is bound, except to the extent that the failure to obtain such consent or approval would not have a Performance Material Adverse Effect with respect to EFS; or

v. require any filing with, notice to, consent or approval of, or any other action to be taken with respect to, any regulatory authority, except to the extent that the failure to obtain such consent or approval would not have a Performance Material Adverse Effect with respect to EFS.

(d) <u>Solvency</u>. EFS is Solvent.

(e) <u>No Default</u>. Neither EFS nor, to the best of its Knowledge, its Affiliates are in default with respect to any contract, agreement, lease, or other instrument to which it is a party

or by which it is bound, except for defaults which would not have a Performance Material Adverse Effect with respect to EFS, nor has EFS received any notice of default under any contract, agreement, lease or other instrument which default or notice of default would materially and adversely affect the performance by EFS of its obligations under this Agreement. No EFS Event of Default has occurred and is continuing.

(f) <u>Books and Records</u>. All of EFS's and, to the best of its Knowledge, its Affiliates' records, files and books of account relating to the Program, including records provided to Bank regarding Account activities, are in all material respects complete and correct and are maintained in accordance with Applicable Law.

(g) <u>No Litigation</u>. As of the Effective Date, no action, claim or any litigation, proceeding, arbitration, investigation or controversy is pending against EFS, at law, in equity or otherwise, before any court, board, commission, agency or instrumentality of any federal, state, or local government or of any agency or subdivision thereof or before any arbitrator or panel of arbitrators, to which EFS is a party, which, if adversely determined, would have a Performance Material Adverse Effect with respect to EFS.

(h) <u>Ownership of Trademarks</u>. HRB Innovations, Inc. is the sole legal and beneficial owner of the trademarks and other Intellectual Property licensed to Bank pursuant to the Block Licensing Agreement.

(i) <u>No Convicted Individuals</u>. No officer, director, or employee of EFS subject to a restriction under 12 U.S.C. §1829 shall participate in any of EFS's activities provided for by this Agreement.

10.2 Bank's Representations, Warranties and Covenants.

To induce EFS to establish the Program, Bank makes the following representations, warranties and covenants to EFS, each and all of which shall survive the execution and delivery of this Agreement, and each and all of which shall be deemed to be restated and remade with the same force and effect on each day of the Term, except as otherwise stated.

(a) <u>Corporate Existence</u>. Bank (i) is a federal savings bank duly organized and validly existing under the laws of the United States, and (ii) has all necessary licenses, permits, consents or approvals from or by, and has made all necessary notices to, all Regulatory Authorities, to the extent required for Bank's business, and for the performance of its obligations pursuant to this Agreement, except where the failure to have such licenses, permits, consents or approvals would not have a Performance Material Adverse Effect with respect to Bank or a Program Material Adverse Effect.

(b) <u>Capacity: Authorization: Validity</u>. Bank has all necessary corporate power and authority to execute and enter into this Agreement and perform the obligations required of Bank hereunder and the other documents, instruments and agreements relating to the Program and this Agreement executed by Bank pursuant hereto. The execution and delivery by Bank of this Agreement and all documents, instruments and agreements executed and delivered by Bank pursuant hereto, and the consummation by Bank of the transactions specified herein have been duly and validly authorized and approved by all necessary corporate action of Bank. This Agreement (i) has been duly executed and delivered by Bank, (ii) constitutes the valid and legally binding obligation

of Bank, and (iii) is enforceable in accordance with its terms (subject to applicable bankruptcy, insolvency, reorganization, receivership or other laws affecting the rights of creditors generally and by general equity principles including those respecting the availability of specific performance).

(c) <u>Conflicts: Defaults: Etc</u>. The execution, delivery and performance of this Agreement by Bank, its compliance with the terms hereof, and its consummation of the transactions specified herein will not:

i. conflict with, violate, result in the breach of, constitute an event which would, or with the lapse of time or action by a third-party or both would, result in a default under, or accelerate the performance required by, the terms of any material contract, instrument or agreement to which Bank is a party or by which it is bound, or by which Bank assets are bound, except for conflicts, breaches and defaults which would not have a Performance Material Adverse Effect with respect to Bank;

ii. conflict with or violate the articles of incorporation or by-laws, or any other equivalent organizational documents, of Bank;

iii. violate any Applicable Law or conflict with, or require any consent or approval under any judgment, order, writ, decree, permit or license, to which Bank is a party or by which it is bound or affected, except to the extent that such violation or the failure to obtain such consent or approval would not have a Performance Material Adverse Effect with respect to Bank;

iv. require the consent or approval of any other party to any contract, instrument or commitment to which Bank is a party or by which it is bound, except to the extent that the failure to obtain such consent or approval would not have a Performance Material Adverse Effect with respect to Bank; or

v. require any filing with, notice to, consent or approval of, or any other action to be taken with respect to, any regulatory authority, except to the extent that the failure to obtain such consent or approval would not have a Performance Material Adverse Effect with respect to Bank.

(d) <u>Solvency</u>. Bank is Solvent.

(e) <u>No Default</u>. Neither Bank nor, to the best of its Knowledge, its Affiliates are in default with respect to any contract, agreement, lease, or other instrument to which it is a party or by which it is bound, except for defaults which would not have a Performance Material Adverse Effect with respect to Bank, nor has Bank received any notice of default under any contract, agreement, lease or other instrument which default or notice of default would materially and adversely affect the performance by Bank of its obligations under this Agreement. No Bank Event of Default has occurred and is continuing.

(f) <u>Books and Records</u>. All of Bank's and, to the best of its Knowledge, its Affiliates' records, files and books of account relating to the Program, including records provided to EFS regarding Account activities, are in all material respects complete and correct and are maintained in accordance with Applicable Law.

(g) <u>No Litigation</u>. As of the Effective Date, no action, claim or any litigation, proceeding, arbitration, investigation or controversy is pending against Bank, at law, in equity or

otherwise, before any court, board, commission, agency or instrumentality of any federal, state, or local government or of any agency or subdivision thereof or before any arbitrator or panel of arbitrators, to which Bank is a party, which, if adversely determined, would have a Performance Material Adverse Effect with respect to Bank.

(h) <u>MasterCard and Visa Membership</u>. Bank will join and/or maintain membership in any Payment Network as required to offer the Financial Products under the Program. Bank also provides, or shall, pursuant to this Agreement and the Purchase Agreement, acquire, one or more BINs to be used exclusively for the Prepaid Products and the Credit Card Product. EFS is responsible for any costs incident to initially enrolling on or transferring a Financial Product to or from any Payment Network.

10.3 Mutual Covenants.

(a) <u>Notice of Litigation</u>. To the extent permitted by Applicable Law, each Party shall promptly notify the other Party if it receives written notice of any pending or threatened action, claim or litigation, proceeding, arbitration, investigation or controversy against such Party or its Affiliates, at law, in equity or otherwise, before any court, board, commission, agency or instrumentality of any federal, state, or local government or of any agency or subdivision thereof or before any arbitrator or panel of arbitrators (other than customer complaints in the ordinary course of business), to which such Party is a party, which, if adversely determined, would reasonably be expected to have a Performance Material Adverse Effect on such Party's ability to perform its obligations under this Agreement.

(b) <u>Non-Solicitation</u>. Each Party covenants and agrees that, until twelve (12) months after the Final Wind-Down Date, it will not, directly or indirectly, solicit or employ any employees of the other Party who have direct managerial responsibility for the Program; provided, that: (i) nothing herein shall prohibit a Party from making offers of employment to or hiring persons whose employment with the non-soliciting Party has been terminated by the non-soliciting Party without cause; and (ii) general advertising or searches regarding the availability of employment opportunities (whether by print, search firm, radio, electronic media, website or otherwise) shall not constitute "solicitation."

(c) <u>Disaster Recovery Plan</u>. Each Party agrees to maintain a disaster recovery plan, which it shall test regularly, but at a minimum one time per calendar year, as well as systems, equipment, facilities and trained personnel, that shall enable it to perform its essential obligations under this Agreement consistent with such Party's disaster recovery plan continuously through a disaster. Either Party may request the other Party to make its disaster recovery plan available for review. Either Party may make changes to its disaster recovery plan from time to time without the other Party's consent; provided that such changes do not materially decrease the level of protection offered by the disaster recovery plan.

# ARTICLE 11 CONFIDENTIALITY

# 11.1 Confidential Information.

(a) "<u>Confidential Information</u>" of a Party means (i) information that is provided by or on behalf of such Party to the other Party, its Affiliates or to EFS Service Providers or Bank

Service Providers, as applicable, in connection with the Program, or (ii) information about such Party, its Affiliates, or their respective businesses or employees, that is otherwise obtained by the other Party in connection with the Program, in each case including: (A) information concerning Program Marketing Plans, marketing philosophies, objectives and financial results; (B) information regarding business systems, methods, processes, financing data, programs and products; (C) information unrelated to the Program obtained by the other Party in connection with this Agreement, including by accessing or being present at the business location of the other Party; (D) proprietary technical information, including source and object codes; (E) competitive advantages and disadvantages, technological development, sales volume(s), merchandise mix, business relationships and methods of transacting business, product design, product features and functionalities, operational and data processing capabilities, and systems software and hardware and the documentation thereof; (F) other information regarding the business or affairs of the other Party or its Affiliates or the transactions contemplated by this Agreement that such other Party or its Affiliates reasonably considers confidential or proprietary; and (G) any copies, excerpts, summaries, analyses or notes of the foregoing. Subject to any disclosure required by Applicable Law, the Parties agree that the terms of this Agreement and the Product Schedules shall be Confidential Information of both Parties.

(b) The obligations hereunder with respect to Confidential Information shall not apply to any information that is sourced from information that (i) is publicly known without breach of this Agreement, or (ii) either Party or its Affiliates, EFS Service Providers or Bank Service Providers, as applicable, (A) already knows at the time it is disclosed as shown by their written records, (B) receives from a third-party permitted to disclose it without restriction, or (C) develops independently without use of Confidential Information.

11.2 Limits on Use and Disclosure.

(a) Each Party shall comply with, and cause its respective directors, officers, employees, representatives and as applicable, EFS's Affiliates, EFS Service Providers, Bank's Affiliates or Bank Service Providers, to comply with the provisions of this <u>Section 11.2</u>.

(b) If a Party (the "<u>Receiving Party</u>") receives Confidential Information of the other Party (the "<u>Disclosing</u> <u>Party</u>"), the Receiving Party shall not use or disclose Confidential Information of the Disclosing Party except:

i. to perform its obligations or enforce its rights with respect to the Program or this Agreement;

- ii. as expressly permitted by this Agreement;
- iii. with the prior consent of the Disclosing Party;

iv. to respond to a valid subpoena, order or request of any Regulatory Authority ("<u>Regulatory Request</u>"), or of any recognized stock exchange;

v. based on advice of legal counsel, to the extent either Party is required by Applicable Law or valid court or governmental agency order to disclose, in which case the Party receiving such an order must, if permitted by Applicable Law, give prompt notice to the other Party, allowing it to seek a protective order;

vi. to comply with any Applicable Law, or legal or regulatory process, including the Securities Act of 1933, the Securities Exchange Act of 1934 and the rules and regulations promulgated thereunder and related thereto; or

vii. as otherwise required by Applicable Law.

(c) A Receiving Party shall limit access to the Disclosing Party's Confidential Information to those employees, consultants and, as applicable, to EFS's Affiliates, EFS Service Providers, Bank's Affiliates or Bank Service Providers, that have a reasonable need to access such Confidential Information in connection with the Program or other purposes permitted by this Agreement and only if that Person has agreed to confidentiality obligations at least as restrictive as those set forth in this <u>Article 11</u> (Confidentiality) prior to disclosure. The Receiving Party shall remain responsible to the Disclosing Party for acts or omissions of individuals referred to in the preceding sentence that if committed by the Receiving Party would constitute a violation of the Receiving Party's confidential obligations hereunder.

(d) Notwithstanding anything else contained in this Agreement, a Party will not be obligated to take any action with respect to the collection, use or disclosure of information in the Program that such Party believes in good faith would cause, or is reasonably likely to cause, either Party to violate any Applicable Law (including privacy and security laws and the reuse and redisclosure provisions of GLBA).

11.3 <u>Regulatory Requests</u>. If a Receiving Party receives a Regulatory Request to disclose any Confidential Information of the Disclosing Party, the Receiving Party shall, to the extent permitted by Applicable Law:

- (a) notify the Disclosing Party thereof promptly after receipt of such Regulatory Request;
- (b) consult with the Disclosing Party with respect to such Regulatory Request; and

(c) if disclosure is required or deemed advisable based on advise of legal counsel of the Receiving Party, at the Disclosing Party's request and expense, reasonably cooperate with the Disclosing Party in any attempt by the Disclosing Party to obtain a protective order or other reliable assurance that confidential treatment will be accorded to the Confidential Information of the Disclosing Party.

11.4 <u>Disposition of Confidential Information</u>. Upon the termination or expiration of this Agreement or a Product Schedule, the Receiving Party shall maintain any retained Confidential Information in accordance with the terms of this Agreement and Applicable Law and shall dispose of any Confidential Information in accordance with the Disclosing Party's reasonable instructions; provided, however, the Receiving Party in possession of tangible property containing the Disclosing Party's Confidential Information may retain one (1) archived copy of such material.

11.5 <u>Unauthorized Use or Disclosure</u>. Each Receiving Party agrees that any unauthorized use or disclosure of Confidential Information of the Disclosing Party may cause immediate and irreparable harm to the Disclosing Party for which damages may not constitute an adequate remedy. In that event, the Receiving Party agrees that injunctive relief may be warranted in addition to any other remedies the Disclosing Party may have at law or in equity all of which

shall be cumulative and in addition to any rights and remedies available by contract, law, rule, regulation or order. In addition, the Receiving Party agrees to promptly advise the Disclosing Party in writing of any unauthorized misappropriation, disclosure or use by any Person of the Confidential Information in violation of the Receiving Party's obligations under this Agreement which may come to its attention and Receiving Party shall take appropriate steps, at its own expense, as reasonably requested by the Disclosing Party to limit, stop or otherwise remedy such misappropriation, disclosure or use.

# ARTICLE 12 PRIVACY AND DATA SECURITY

12.1 Privacy.

(a) Subject to each Party's compliance obligations in <u>Section 4.3</u> (Compliance Obligations), each Party shall comply with Applicable Law relating to the use and disclosure of Program Customer Data and Prospect Data, including the applicable terms and provisions of GLBA and the Code. Without limiting the foregoing, each Party shall implement and maintain appropriate administrative, technical and physical safeguards to protect the security, confidentiality and integrity of all Program Customer Data and Prospect Data.

(b) In furtherance of and without limiting the foregoing, the Parties agree to, in good faith, jointly develop and prepare, and comply with, the privacy notice applicable to the Program (the "<u>Privacy Notice</u>").

(c) Each Party shall ensure that any third-party (other than attorneys, accountants and any third-party advisors that are bound by a professional duty of confidentiality to such Party) to whom Program Customer Data or Prospect Data is transferred or made available by or on behalf of such Party signs a written contract with the contracting Party in which such third-party agrees: (i) to restrict its use of Program Customer Data and Prospect Data, as applicable, to the use specified in the agreement between EFS or Bank and the third-party (which use must be in compliance with the Party's permitted uses of the information, including as provided in this <u>Article 12</u> (Privacy and Data Security)); (ii) to comply with all Applicable Law, the applicable Payment Network Rules and the Privacy Notice; and (iii) to implement and maintain appropriate administrative, technical and physical safeguards to protect the security, confidentiality and integrity of all Program Customer Data and Prospect Data, including compliance with the provisions of the Data Security Requirements.

(d) The Parties agree to negotiate in good faith, if necessary in the reasonable business judgment of EFS or Bank, an amendment to (i) the provisions of this Agreement related to the use and disclosure of Program Customer Data and Prospect Data and (ii) any other applicable documents, such as privacy policies and privacy guidelines, for the purpose of aligning the privacy policies and practices of the Parties and assuring continued compliance with Applicable Law.

12.2 Data Security.

(a) The Parties agree to, in good faith, jointly develop and prepare, and comply with, the information security and business continuity policies and procedures applicable to the Program as required by Applicable Law (the "<u>Data Security</u> <u>Requirements</u>"). In addition, EFS and Bank will each establish, maintain and implement an information security program, including

appropriate administrative, technical and physical safeguards, that is designed to meet the objectives of the Interagency Guidelines Establishing Standards for Safeguarding Information Security Data, including, at a minimum, maintenance of an information security program that is designed to: (i) ensure the security and confidentiality of Program Customer Data and Prospect Data, as applicable; (ii) protect against any reasonably anticipated threats or hazards to the security or integrity of Program Customer Data or Prospect Data; (iii) protect against unauthorized access to or use of Program Customer Data or Prospect Data that could result in substantial harm or inconvenience to any Program Customer or Prospective Customer; and (iv) ensure the proper disposal of Program Customer Data and Prospect Data, as applicable. Each Party shall use the same degree of care in protecting Program Customer Data and Prospect Data against unauthorized disclosure as it accords to its own confidential customer information, but in no event less than a reasonable standard of care.

Each of EFS and Bank shall notify the other Party promptly, and in any event within twenty-four (24) (b) hours, following discovery or notification of any actual or suspected Security Breach (as defined below) of the information systems maintained by EFS (including through EFS Service Providers) or Bank (including through Bank Service Providers), respectively, unless such systems do not access, process or store Program Customer Data or Prospect Data. A Party that suffers an actual or suspected Security Breach (the "Affected Party") agrees to take action promptly, at its own expense, to investigate the actual or suspected Security Breach and, if an actual Security Breach is confirmed, to identify and mitigate its effects and to implement reasonable and appropriate measures in response. The Affected Party shall also permit the other Party to conduct or require the Affected Party to engage a qualified third-party to conduct its own investigation of an actual or suspected Security Breach; provided, however, that such investigation may not unreasonably interfere with the investigation being conducted by the Affected Party or the operations of the Affected Party. The Affected Party also will provide the other Party with information reasonably requested by the other Party regarding such Security Breach to assist such other Party in implementing its information security response program and, if applicable, in notifying affected Program Customers and Prospective Customers, as applicable, as well as other third parties as required by Applicable Law. The Affected Party shall pay for the reasonable costs of any such investigation and notification, which notification shall be mutually agreed upon by the Parties (such agreement not to be unreasonably withheld) or other remediation reasonably deemed necessary. "Security Breach" means the unauthorized access to or acquisition of any record containing Program Customer Data or Prospect Data, whether in paper, electronic, or other form, in a manner that renders misuse of the information reasonably possible or that otherwise compromises the security, confidentiality, or integrity of the information; provided, however, that it shall also include any data or information security breach under Applicable Law.

(c) EFS and Bank, respectively, will use reasonable measures designed to properly dispose of all records containing Program Customer Data and Prospect Data, as applicable, whether in paper, electronic, or other form, including adhering to policies and procedures that require the destruction or erasure of electronic media containing such Program Customer Data and Prospect Data, as applicable, so that the information cannot practicably be read or reconstructed.

## 12.3 Collection, Ownership and Use of Program Customer Data.

(a) <u>Ownership of Program Customer Data</u>. The Parties recognize that Program Customer Data is owned by Bank, subject to <u>Article 15</u> (Rights Upon Termination), and that Program Customers are Company Customers as well as Bank Customers. Accordingly, each Party has certain ownership and use rights in the information relating to such Program Customers. The Parties acknowledge that the same or similar information may be included in Company Customer Data, Prospect Data, Bank Customer Data and Program Customer Data and to the extent that there is overlapping information in Company Customer Data, Prospect Data, Bank Customer Data and Program Customer Data, subject to the limitations set forth in this Agreement and Applicable Law: (i) Bank shall retain its ownership and use rights in such Program Customer Data and Prospect Data; and (ii) EFS or EFS Affiliates, as applicable, shall retain their respective ownership and use rights in Company Customer Data and Prospect Data.

(b) EFS, its Affiliates and EFS Service Providers shall be entitled to access and may use and disclose all Program Customer Data and Prospect Data in any manner permitted by this Agreement, Applicable Law and the Privacy Notice. For the avoidance of doubt, (i) Company Customer Data obtained by EFS or any EFS Affiliate in connection with the provision of tax preparation related goods and services that are not part of the Program and (ii) Prospect Data, in each case, shall belong exclusively, to the extent permitted by Applicable Law, to EFS or such EFS Affiliate, as applicable.

(c) Notwithstanding any rights they may have to use or disclose Program Customer Data, Prospect Data and Applicant Data under Applicable Law or the Privacy Notice, neither Bank, nor its Affiliates nor Bank Service Providers shall use or disclose Program Customer Data, Prospect Data or Applicant Data for any purpose other than: (i) to perform its obligations or enforce its rights with respect to the Program; (ii) as expressly permitted by this Agreement; (iii) pursuant to a Regulatory Request; (iv) by a Party to a Regulatory Authority as required or requested by such Regulatory Authority; or (v) as otherwise required by Applicable Law.

(d) Each Party shall develop the appropriate tools and resources to allow for efficient access to Program and Financial Product transaction data and Account and Applicant level data, subject to any restrictions under Applicable Law. The operational method for such access shall be set forth in the Program Data Site.

(e) Each of Bank and EFS shall maintain the Privacy Notice to permit the broadest rights allowable under Applicable Law for sharing of Program Customer Data with the other Party and such other Party's Affiliates.

## ARTICLE 13 EVENTS OF DEFAULT

# 13.1 EFS Event of Default.

The occurrence of any one (1) or more of the events specified in this <u>Section 13.1</u> (regardless of the reason therefor) with respect to EFS shall be an "<u>EFS Event of Default</u>."

(a) <u>Failure to Make Payments When Due</u>. EFS fails to make a payment of any material amount due and payable pursuant to this Agreement that is not disputed in good faith and such failure shall remain unremedied for a period of five (5) Business Days after Bank shall have

given notice thereof. For purposes of this Section 13.1(a), the term "material amount" shall mean an amount greater than \$100,000.00.

(b) <u>Failure to Settle</u>. Notwithstanding <u>Section 13.1(a)</u>, if EFS fails to settle any amount that is not disputed in good faith, within three (3) Business Days after the date on which Bank shall have given notice thereof in accordance with <u>Section 2.15</u> (Settlement Statements).

(c) <u>Breach of EFS's Representations, Warranties and Covenants</u>. (i) Any representation or warranty of EFS in this Agreement or any of the Product Schedules shall fail to be true and correct in any material respect as of the date when made or reaffirmed, or (ii) EFS shall fail to perform its covenants as set forth in this Agreement or any of the Product Schedules, or any other material covenant or other agreement contained in this Agreement or any of the Product Schedules that EFS is required to perform; and, in either case, such failure has a Performance Material Adverse Effect with respect to EFS and the same shall remain uncured for a period of thirty (30) days after Bank provides written notice thereof.

(d) <u>Solvency</u>. EFS (i) shall not be Solvent; (ii) shall admit in writing its inability to pay its debts generally; (iii) shall make a general assignment for the benefit of its creditors; (iv) shall have any proceeding instituted by or against it seeking to adjudicate it as bankrupt or insolvent or seeking liquidation, reorganization or any similar alternative under any law relating to bankruptcy or relief of debtors, or seeking the entry of an order for relief or the appointment of a receiver or other similar official for it or for any substantial part of its property, and, in the case of any proceeding instituted against it (but not instituted by it), either such proceeding shall remain undismissed or unstayed for a period of thirty (30) days, or any of the actions sought in such proceeding (including the entry of an order for relief against, or the appointment of a receiver or other similar official for, it or any substantial part of its property) shall occur; or (v) shall take any corporate action to authorize any of the actions set forth in clause (iii) or (iv) of this <u>Section 13.1(d)</u>.

## 13.2 Bank Event of Default.

The occurrence of any one (1) or more of the events specified in this <u>Section 13.2</u> (regardless of the reason therefor) with respect to Bank shall be a "<u>Bank Event of Default</u>."

(a) <u>Failure to Make Payments when Due</u>. Bank fails to make a payment of any material amount due and payable pursuant to this Agreement that is not disputed in good faith and such failure shall remain unremedied for a period of five (5) Business Days after EFS shall have given notice thereof. For purposes of this <u>Section 13.2(a)</u>, the term "material amount" shall mean an amount greater than \$100,000.00.

(b) <u>Failure to Settle</u>. Notwithstanding <u>Section 13.2(a)</u>, if Bank fails to settle (i) any amount that is not disputed in good faith, within three (3) Business Days after the date on which EFS shall have given notice thereof in accordance with <u>Section 2.15</u> (Settlement Statements), or (ii) any Payment Network transactions.

(c) <u>Breach of Bank's Representations, Warranties and Covenants</u>. (i) Any representation or warranty of Bank in this Agreement or any of the Product Schedules shall fail to be true and correct in any material respect as of the date when made or reaffirmed, or (ii) Bank shall fail to perform its covenants as set forth in this Agreement or any of the Product Schedules, or any

other material covenant or other agreement contained in this Agreement that Bank is required to perform including the failure to meet its obligations under any Program Marketing Plans; and, in either case, such failure results in a Performance Material Adverse Effect with respect to Bank and the same shall remain uncured for a period of thirty (30) days after EFS provides written notice thereof.

(d) <u>Solvency</u>. Bank (i) shall not be Solvent; (ii) shall admit in writing its inability to pay its debts generally; (iii) shall make a general assignment for the benefit of its creditors; (iv) shall have a receiver appointed (or sought to be appointed) by Bank's primary federal regulator or shall have any proceeding instituted by or against it seeking to adjudicate it as bankrupt or insolvent or seeking liquidation, reorganization or any similar alternative under any law relating to bankruptcy or relief of debtors, or seeking the entry of an order for relief; or (v) shall take any corporate action to authorize any of the actions set forth in clause (iii) or (iv) of this Section 13.2(d).

(e) <u>Undercapitalization</u>. Bank is deemed to be less than adequately capitalized or determined to be in a troubled condition within the meaning of Section 38 of the Federal Deposit Insurance Act by the appropriate federal banking agency having primary supervisory jurisdiction over Bank.

# ARTICLE 14 TERM AND TERMINATION

14.1 <u>Term</u>. The term of this Agreement and the Product Schedules shall commence upon the Effective Date and end June 30, 2021, unless earlier terminated as provided in this Agreement (the "<u>Term</u>"). Subject to the provisions of this <u>Article 14</u> (Term and Termination), the term of all Product Schedules will be co-terminus with the Term. This Agreement will remain in full force and effect to the extent any Product Schedule remains in full force and effect.

14.2 Mutual Termination Rights.

(a) <u>Events of Default</u>. Each Party shall have the right to terminate this Agreement or a Product Schedule upon not less than twenty (20) days' prior written notice if any Event of Default with respect to the other Party shall occur and be continuing pursuant to <u>Article 13</u> (Events of Default).

(b) <u>Force Majeure Event</u>. If a Force Majeure Event with respect to a Party has occurred pursuant to <u>Section 17.8</u> (Force Majeure) that materially prevents or impedes the other Party's performance hereunder and such Force Majeure Event continues for a period of more than thirty (30) days, the other Party shall have the right to terminate the affected Product Schedule(s) while the Force Majeure Event continues by providing written notice to the Party experiencing the Force Majeure Event, such termination to be effective on the date specified in the notice of termination.

14.3 Additional Termination Rights of EFS.

(a) <u>Early Termination</u>. EFS shall have the one-time right to unilaterally terminate this Agreement for any or no reason effective June 30, 2017, by giving written notice of termination to Bank no later than April 30, 2017.

(b) <u>Durbin Regulatory Event</u>. If a Durbin Regulatory Event occurs, EFS shall have the right, in its sole discretion, to terminate this entire Agreement, or any of the Product Schedules with respect to Emerald Advance, Emerald Card or Refund Transfer, upon thirty (30) days' prior written notice to Bank.

(c) <u>Changed Circumstances Termination</u>. If Bank unilaterally changes (i) the terms, pricing, conditions, underwriting or other characteristics of, or the Account Documentation, Marketing Materials, Servicing Materials or any other Program documentation or requirements for, any Financial Product, or (ii) any other aspect of the Program or the obligations of EFS under the Program, that (A) individually or collectively have, or are reasonably expected to have, a material adverse effect on the economic benefits that EFS and its Affiliates reasonably expect to derive from the Financial Product based upon the pricing, terms, conditions, underwriting or other characteristics of the Financial Product as in effect on the date of this Agreement, or (B) expose EFS or any of its Affiliates to what EFS, in good faith believes, is a significant increase in the level of legal, regulatory, or litigation risk (and EFS provides thirty (30) days' prior written notice advising Bank's chief executive officer that EFS disagrees with such change and the basis for its disagreement, with particularity, and the change remains in effect after such thirty (30) day period), then notwithstanding anything in this Agreement to the contrary, EFS shall have the right, in its sole discretion, to terminate this entire Agreement, or any one or more of the Product Schedules, upon at least fifteen (15) days' prior written notice to Bank. For purposes of this <u>Section 14.3(c)</u>, "material" means \$5,000,000 in consolidated pre-tax net income, or \$15,000,000 in consolidated gross revenue, of Block, Inc.

Notwithstanding the foregoing, EFS shall have no termination right with respect to an industry wide change that is clearly necessary to comply with Applicable Law or Payment Network Rules. The Parties acknowledge that the current Program and Financial Products as in effect on the date of this Agreement are in compliance with Applicable Law.

14.4 Cross Termination of Product Schedules.

(a) Subject to the terms contained herein, an individual Product Schedule may be terminated upon the occurrence of a termination event described in <u>Section 14.2</u> or <u>14.3</u> in accordance with this <u>Article 14</u> (Term and Termination) without causing the termination of any of the other Product Schedules or this Agreement.

(b) If any of the Product Schedules with respect to Emerald Advance, Emerald Card or Refund Transfer is terminated by Bank in accordance with this <u>Article 14</u>, EFS, in its sole discretion, may terminate any or all of the other Product Schedules or this Agreement; provided, however, that EFS shall have the right to exercise the EFS Purchase Option set forth in <u>Section 15.1</u> (EFS Purchase Options) with respect to the Product Schedules that are terminated.

(c) If any of the Product Schedules with respect to the Emerald Card or Refund Transfer is terminated by EFS in accordance with this <u>Article 14</u>, Bank in its sole discretion, may terminate this Agreement.

## ARTICLE 15 RIGHTS UPON TERMINATION

15.1 <u>EFS Purchase Options</u>. When this Agreement expires in accordance with its Term, or if this Agreement or any Product Schedule is terminated early for any reason, then EFS

shall have the option (each an "EFS Purchase Option") to purchase, have an Affiliate purchase or to arrange for a federally-insured depository institution selected by EFS to purchase (each of EFS, an EFS Affiliate or a federally-insured depository institution, a "Nominated Purchaser"), all of the Accounts associated with the Program (other than those Accounts relating to individual retirement accounts, which will remain with Bank), or if only a Product Schedule is terminated, the Accounts associated with the related Financial Product, and the Accounts Receivable and related rights, funds, interest and fees ("Purchased Assets") (which, for the avoidance of doubt, shall include all of the related assets of the type and nature of the "Transferred Asset" as defined in Section 2.01 of the Purchase Agreement), free and clear of all liens, claims and encumbrances created by Bank, and to assume the payment and performance obligations and other liabilities arising after the closing date relating to such Purchased Assets (the "Assumed Liabilities") (which, for the avoidance of doubt, shall include all of the related Liabilities of the type and nature of the "Assumed Liabilities" as defined in Section 2.03 of the Purchase Agreement). Such closing date is the "Assumed Accounts Purchase Date." EFS shall notify Bank in writing of a Nominated Purchaser's intent to purchase the Purchased Assets and the Assumed Liabilities (an "Exercise Notice") no later than one hundred twenty (120) days after the Termination Date of this Agreement or of any Product Schedule (the "Purchase Option Exercise Period"). EFS may designate different Nominated Purchasers with respect to different EFS Purchase Options and with respect to different Financial Products, and may designate a substitute or replacement Nominated Purchaser, but each thereof shall be deemed to be the "Nominated Purchaser" as defined in this Section 15.1. If EFS does not provide the Exercise Notice to Bank before the expiration of the Purchase Option Exercise Period, then the EFS Purchase Option under this Section 15.1 shall expire. If EFS determines not to exercise the EFS Purchase Option, then EFS shall provide Bank with a written notice of no interest (" No Interest Notice").

15.2 <u>Evaluation Data</u>. Bank shall cooperate with EFS and the Nominated Purchaser to provide EFS and Nominated Purchaser with access (subject to customary confidentiality agreements) to all Program-related data and information (other than Bank's proprietary work product) reasonably requested by the Nominated Purchaser for the sole purpose of due diligence regarding the purchase of the Accounts, provided that such information is not otherwise reasonably available from EFS or its Affiliates.

#### 15.3 Determination of Purchase Price.

(a) In the event that EFS issues an Exercise Notice, the purchase price for the Purchased Assets and the Assumed Liabilities ("<u>Purchase Price</u>") shall be calculated as the sum of (i) the book value of deposits associated with Emerald Card accounts; (ii) the fair value of Emerald Advance participation interests and receivables; and (iii) the amount of one dollar (\$1.00) for all Emerald Advance, Emerald Card, Refund Transfer, and Credit Card Accounts and the other Program assets owned by Bank. Other Program assets will include all of the related assets of the type and nature of the "Transferred Asset" as defined in Section 2.01 of the Purchase Agreement. "Assumed Liabilities" shall include all of the related liabilities of the type and nature of the "Assumed Liabilities" as defined in Section 2.03 of the Purchase Agreement. The Purchase Price shall be determined as of the Assumed Accounts Purchase Date. The fair value of Emerald Advance participation interests shall be calculated in a manner consistent with <u>Schedule 15.3</u>; provided however, that in the event of a dispute in the determination of fair value, the parties shall rely on the determination of an independent accountant firm to be selected by Bank, EFS and Nominated Purchaser, consistent with pricing used in the Purchase Agreement.

(b) In the event that the EFS Purchase Option is validly exercised, Bank and EFS shall, in good faith, use commercially reasonable efforts to consummate the purchase of the Purchased Assets and the assumption of the Assumed Liabilities by Nominated Purchase methanics) at the Purchase Price). EFS shall notify Nominated Purchaser of EFS's expectation that the Nominated Purchaser shall, in good faith, use commercially reasonable efforts to consummate the purchase Assets and the assumption of the Purchased Assets and the assumption of the Purchased Assets and the purchaser of the Purchased Assets and the assumption of the Assumed Liabilities at the Purchase Price.

(c) <u>No Purchase</u>. If a Purchase Option Expiration Date occurs, the Accounts retained by Bank shall be liquidated pursuant to the provisions of <u>Section 15.6</u> (Wind-down by Bank); provided, however, that nothing in this <u>Section 15.3(c)</u> shall be construed as limiting or relieving the obligation of Bank and EFS to use commercially reasonable efforts to consummate the purchase of the Purchased Assets and the assumption of the Assumed Liabilities after EFS delivers an Exercise Notice. For the avoidance of doubt, as long as EFS uses commercially reasonable efforts to consummate such purchase, EFS shall not be liable for a Nominated Purchaser's failure to purchase the Purchased Assets and assume the Assumed Liabilities, and in no event shall EFS be obligated to purchase the Purchased Assets and assume the Assumed Liabilities for itself.

15.4 <u>Purchase Mechanics</u>. After EFS has delivered an Exercise Notice, Bank, EFS and Nominated Purchaser shall, with reasonable expedition, negotiate in good faith, execute and deliver all necessary agreements, instruments and other documentation customary for a transaction of this kind, including a purchase and sale agreement, which agreements may require each of Bank, EFS and Nominated Purchaser to agree to certain representations, warranties, covenants, indemnities, transition services and other terms and conditions usual and customary for a transaction of this kind. All such agreements shall be in a form reasonably acceptable to the Parties. Bank, EFS and the Nominated Purchaser shall in good faith use commercially reasonable efforts to expeditiously consummate the purchase of the Purchased Assets and the assumption of the Assumed Liabilities as contemplated hereby no later than two hundred seventy (270) days from the date on which EFS has provided an Exercise Notice to Bank. Notwithstanding the foregoing, (a) if the Assumed Accounts Purchase Date falls within the Tax Season, at EFS's option, the Assumed Accounts Purchase Date shall be extended up to one hundred eighty (180) days in the event that any application for Regulatory Authority approval that is required for the Assumed Accounts Purchase Date remains pending.

## 15.5 Duties After Termination or Expiration.

(a) <u>Continuation of Rights and Obligations under Agreement</u>. Notwithstanding the occurrence of a Termination Date with respect to this Agreement, the rights and obligations of the Parties under this Agreement (other than the Non-Surviving Obligations) shall continue until the latest of (i) the expiration of the EFS Purchase Option or delivery by EFS of a No Interest Notice; (ii) if EFS has delivered an Exercise Notice, the consummation of the purchase of the Purchased Assets and the assumption of the Assumed Liabilities; or (iii) if EFS has delivered an Exercise Notice, the expiration of the purchase delivered set forth in <u>Section 15.4</u> (Purchase Mechanics) without the consummation of the purchase of the Purchased Assets and the assumption of the Assumed Liabilities. The Parties shall mutually agree on a conversion plan and shall not unreasonably

withhold or delay execution of the conversion plan. Bank shall be prepared to participate in the conversion of the Accounts to the purchaser thereof by the conversion date to be specified in the conversion plan (the "<u>Conversion Date</u>").

In addition, notwithstanding anything else to the contrary in this Agreement, upon the occurrence of a Termination Date, the rights and obligations of the Parties under this Agreement (other than the Non-Surviving Obligations) shall continue during the wind down period described in <u>Section 15.6</u> until the earliest of (i) the Final Wind Down Date; (ii) the completion of the wind down prior to the Final Wind Down Date; (iii) the date that Bank completes the implementation of a plan to replace or substitute one or more Financial Products with similar financial products offered by Bank that are not branded with Company Licensed Marks; or (iv) such other date as may be agreed upon in writing by the Parties and a Nominated Purchaser pursuant to <u>Section 15.4</u>.

(b) <u>ABA Routing Number; BIN; ICA</u>. If EFS delivers an Exercise Notice, the Bank, EFS and Nominated Purchaser shall work together to assign or transfer to the Nominated Purchaser the ABA routing numbers, bank identification numbers ("<u>BIN</u>") or interbank card association numbers ("<u>ICA</u>") range applicable to the Financial Products (to the extent permissible by the Payment Networks and to the extent related to or utilized by the Financial Products) and the account numbers relating to the Financial Products.

(c) <u>Conversion Costs</u>. Each Party shall bear its own costs associated with the sale and conversion of the Purchased Assets and the assumption of the Assumed Liabilities to the initial Nominated Purchaser; provided, however, that (i) EFS shall pay for Bank's reasonable outside legal fees and expenses incurred to negotiate, draft, execute and deliver the purchase agreement and related agreements and regulatory filings contemplated by this <u>Article 15</u> ("<u>Reasonable and Related Outside Counsel Transaction</u> <u>Expenses</u>"), to the extent such outside legal fees and expenses exceed \$200,000.00, and (ii) if for any reason EFS designates a substitute or replacement Nominated Purchaser or if EFS extends the timeframe beyond the initial 270 day period provided by <u>Section 15.4</u>, then EFS shall pay for all of Bank's Reasonable and Related Outside Counsel Transaction Expenses. Third party costs and expenses, to the extent directly related to the conversion to a Nominated Purchaser pursuant to this <u>Article 15</u>, shall be paid or reimbursed by EFS.

15.6 <u>Wind-Down by Bank</u>. If (i) the Purchase Option Exercise Period expires, (ii) EFS issues a No Interest Notice or (iii) EFS delivers an Exercise Notice, but the periods set forth in <u>Section 15.4</u> expire without the consummation of the purchase of the Purchased Assets and the assumption of the Assumed Liabilities, then as promptly as reasonably practicable and in any event within one (1) year thereafter (the "<u>Final Wind-Down Date</u>"), Bank shall wind down the remaining Accounts in the Program or relevant Product Schedule in any lawful manner which may be expeditious or economically advantageous to Bank, including the issuance of a replacement or a substitute card associated with an Account, as applicable; provided, however, that Bank shall (a) remove all Company Licensed Marks from the Financial Products and related materials as promptly as reasonably practicable and in any event prior to the Final Wind-Down Date; (b) discontinue originating Financial Products with Company Licensed Marks within sixty (60) days after the Termination Date; and (c) not re-brand the Financial Products with a Specified Party, sell the Financial Products or Program Customer Data to a Specified Party or a Specified Party's designee, or otherwise permit the Financial Products or Program Customer Data to benefit a Specified Party.

During the wind down period, the rights and obligations of the Parties under this Agreement will continue as provided in the second paragraph of Section 15.5(a). During the wind down period, Bank and EFS shall act in good faith and shall reasonably cooperate with each other to wind down the Program in a reasonable and efficient manner, with the least disruption to Company Customers and Bank Customers as possible.

15.7 <u>Communication with Accountholders</u>. When this Agreement expires in accordance with its Term, or if this Agreement or any Product Schedule is terminated early for any reason, except as required by Applicable Law or Payment Network Rules, Bank and EFS shall mutually agree upon all communications with Accountholders regarding the termination of this Agreement and the Program or of a Product Schedule. If Bank is required by Applicable Law or Payment Network Rules to communicate with Accountholders, then EFS may review and approve (which approval shall not be unreasonably withheld) such communication prior to its distribution to Accountholders. EFS shall pay for the costs of such communications to Accountholders.

15.8 <u>Applicability</u>. The Parties acknowledge and agree that the provisions of this <u>Article 15</u> shall be applicable to the termination of one or more Product Schedules, or this entire Agreement, and that there may be more than one exercise of the EFS Purchase Option. The respective rights and obligations of the Parties hereunder shall apply with respect to each exercise of the EFS Purchase Option.

## ARTICLE 16 INDEMNIFICATION; LIMITATION OF LIABILITY

16.1 <u>Indemnification of Bank by EFS</u>. EFS agrees to protect, indemnify, defend and hold harmless Bank, its parent, subsidiaries and Affiliates, and its and their respective shareholders, directors, officers. employees, agents, representatives and permitted assigns (collectively, "<u>Bank Indemnified Parties</u>"), from and against any and all Indemnified Losses to the extent such Indemnified Losses arise out of or result from:

(a) any third-party Claim brought against any of the of Bank Indemnified Parties in connection with or in any way related to this Agreement, the Program, a Financial Product or a Transaction, except to the extent that an Indemnified Loss resulted from (i) a Program Change made by Bank (or at the direction of Bank) as to which EFS has delivered a Disputed Program Change Notification to Bank, or (ii) any cross-marketing by Bank Indemnified Parties pursuant to <u>Section 6.2(a)</u> (Cross Marketing);

(b) any negligent, willful, or fraudulent act or omission of the part of EFS, its Affiliates, any Distributor, any Franchisee, or and EFS Service Provider, or any of their directors, officers. employees, agents, or representatives, in connection with the Program;

(c) any breach by EFS, its Affiliates, any Distributor or any Franchisee of any representation, warranty, covenant or other provision contained in this Agreement or any other instrument or document delivered by EFS to Bank in connection with the Program; and

(d) the failure of EFS, its Affiliates, any Distributor or any Franchisee to comply with Company Applicable Law or any applicable Payment Network Rule in connection with the Program;

provided, however, that EFS shall have no obligation to indemnify any Bank Indemnified Party under this <u>Section 16.1</u> (Indemnification of Bank by EFS) against any Indemnified Losses to the extent that such Indemnified Losses result from (x) a grossly negligent, willful, or fraudulent act or omission of any Bank Indemnified Party or (y) any breach by any Bank Indemnified Party of any representation or warranty (other than representations and warranties in <u>Section 10.2(c)(iii)</u>) of this Agreement, or the covenants set forth in <u>Section 2.16</u> (Nevada Office), <u>Section 6.2(b)</u> (Cross Marketing), <u>Article 11</u> (Confidentiality) and <u>Article 12</u> (Privacy and Data Security). For the avoidance of doubt, "gross negligence," as used in this <u>Article 16</u>, includes repeated acts of the same or similar negligence for which Bank has received notice from EFS identifying the existence of the negligence and a reasonable remediable action plan, and Bank has failed to implement such remedial action plan within a reasonable period of time to avoid further Indemnification Losses. In addition, EFS agrees to comply with the last two paragraphs of Section 10.02 of the Purchase Agreement.

16.2 <u>Indemnification of EFS by Bank</u>. Bank agrees to protect, indemnify, defend and hold harmless EFS, its parent, subsidiaries and Affiliates, and its and their respective shareholders, directors, officers, employees, agents, representatives and permitted assigns (collectively, "<u>EFS Indemnified Parties</u>"), from and against any and all Indemnified Losses to the extent such Indemnified Losses arise out of or result from:

(a) any third-party Claim brought against any of the of EFS Indemnified Parties in connection with or in any way related to this Agreement, the Program, a Financial Product or a Transaction only to the extent that the Indemnified Loss resulted from a Program Change made by Bank (or at the direction of Bank) as to which EFS has delivered a Disputed Program Change Notification to Bank;

(b) any grossly negligent, willful or fraudulent act or omission of the part of Bank, its parent, subsidiaries or Affiliates, or any Bank Service Provider, or any of their directors, officers. employees, agents, or representatives, in connection with the Program; and

(c) any breach by Bank, its parent, subsidiaries or Affiliates, or any Bank Service Provider, of any representation or warranty (other than representations and warranties in <u>Section 10.2(c)(iii)</u>) of this Agreement, or the covenants, agreements or undertakings set forth in <u>Section 2.16</u> (Nevada Office), <u>Section 6.2(b)</u> (Cross Marketing), <u>Article 11</u> (Confidentiality) and <u>Article 12</u> (Privacy and Data Security);

provided, however, that Bank shall have no obligation to indemnify any EFS Indemnified Party under this <u>Section 16.2</u> (Indemnification of EFS by Bank) against any Indemnified Losses to the extent that such Indemnified Losses result from (x) a negligent, willful or fraudulent act or omission of any EFS Indemnified Party or (y) any breach by any EFS Indemnified Party of any representation, warranty, covenant or other provision contained in this Agreement, or any other agreement, instrument or documents delivered to Bank in connection with the Program. In addition, Bank agrees to comply with the last two paragraphs of Section 10.02 of the Purchase Agreement.

16.3 <u>Notice</u>. If a Party (the "<u>Indemnified Party</u>") receives notice of any third-party claim for which indemnification may be available under this Agreement, the Indemnified Party must promptly notify the other Party (the "<u>Indemnifying Party</u>") in writing of the third-party claim, including, if possible, the amount or estimate of the amount of liability arising from it. The Indemnified Party shall use its commercially reasonable efforts to provide notice to the Indemnifying

Party no later than fifteen (15) days after receipt by the Indemnified Party in the event a suit or action has commenced, or thirty (30) days under all other circumstances; provided, however, that the failure to give such notice shall not relieve an Indemnifying Party of its obligation to indemnify except to the extent the Indemnifying Party is materially prejudiced by such failure.

## 16.4 Right to Defend Claims; Coordination of Defense.

(a) Subject to <u>Section 16.5</u> (Settlement of Claims), the Indemnifying Party shall have the right to defend any such third-party claim at its expense and in the name of the Indemnified Party and shall select the counsel for the defense of such third-party claim as approved by the Indemnified Party, such approval not to be unreasonably withheld, and the Indemnifying Party shall reasonably cooperate with the Indemnified Party in the conduct of the defense against such third-party claim. The Indemnified Party may participate, at its own expense, in such defense and in any settlement discussions directly or through counsel of its choice on a monitoring, non-controlling basis, or at the Indemnified Party's expense and with full control if the Indemnifying Party does not fulfill its obligations to appoint counsel to defend the Indemnified Party which is reasonably satisfactory to the Indemnified Party. The Parties agree to cooperate in good faith to coordinate the defense of any third-party claim that may give rise to indemnification obligations of more than one Indemnifying Party or that may include allegations that are not subject to indemnification.

(b) Notwithstanding the foregoing, the Indemnifying Party shall not have the right to defend any such thirdparty claim on behalf of the Indemnified Party if: (i) it contests (in whole or in part) its indemnification obligations (but only as to the obligations specific to the Indemnifying Party's obligation to indemnify under this <u>Article 16</u> in the event a third-party claim gives rise to indemnification obligations of more than one Indemnifying Party); (ii) it fails to employ counsel approved by the Indemnified Party (such approval not to be unreasonably withheld) to assume the defense of such third-party claim or refuses to replace such counsel upon the Indemnified Party's reasonable request, as provided for herein; (iii) the Indemnified Party reasonably determines that there are issues which could raise possible conflicts of interest between the Indemnifying Party and the Indemnified Party or that the Indemnified Party has claims or defenses that are separate from or in addition to the claims or defenses of the Indemnified Party. In each such case described in clauses (i) - (iv) above, the Indemnified Party shall have the right to direct the defense of the third-party claim and retain its own counsel, and the Indemnifying Party shall pay the cost of such defense, including reasonable attorneys' fees and expenses. If the Indemnifying Party does not assume the defense of any such third-party claim or refuses to replace such counsel upon the Indemnified Party's reasonable request as provided in this <u>Section 16.4(b)</u>, the Indemnifying Party shall have the right to participate, at its own expense, in the defense of such third-party claim with counsel deemed satisfactory to it in its sole discretion.

16.5 <u>Settlement of Claims</u>. The Indemnifying Party shall not be liable for any settlement of any such third-party claim effected without its written consent (which consent shall not be unreasonably withheld) but, if settled with such consent or if there is a judgment for the plaintiff, the Indemnifying Party shall indemnify the Indemnified Party from and against any Indemnified Losses by reason of such settlement or judgment to the extent required by <u>Section 16.1</u>

(Indemnification of Bank by EFS) or <u>16.2</u> (Indemnification of EFS by Bank), as applicable. If the Indemnifying Party assumes the defense of any such third-party claim, it shall be entitled to settle such third-party claim (a) with the consent of the Indemnified Party (which consent shall not be unreasonably withheld) or, (b) if such settlement provides for an unconditional, irrevocable release of the Indemnified Party in connection with all matters relating to the action or proceeding that have been asserted against such Indemnified Party in connection with such third-party claim by the other parties to such settlement, and such release does not include a statement as to, or otherwise imply an admission of fault, culpability or a failure to act by or on behalf of such Indemnified Party, or can reasonably be expected to result in an adverse action against the Indemnified Party by a Regulatory Authority, without the consent of such Indemnified Party which consent may be granted or withheld in the Indemnified Party's reasonable discretion. The amount paid or payable by an Indemnified Party as a result of the Indemnified Party in connection with investigating or defending any such third-party claim, except where the Indemnified Party is required to bear such expenses pursuant to this <u>Article 16</u>, which expenses the Indemnifying Party shall pay as and when incurred, at the request of the Indemnified Party, to the extent it is reasonable to believe that the Indemnifying Party will be ultimately obligated to pay such expenses. If any expenses so paid by the Indemnifying Party are subsequently determined to not be required to be borne by the Indemnifying Party under this <u>Article 16</u>, the Indemnified Party that received the benefit of such payment shall promptly refund the amount so paid to the Indemnifying Party.

16.6 <u>Subrogation</u>. The Indemnifying Party shall be subrogated to any counterclaims, claims in recoupment, or similar rights of the Indemnified Party as against any third-party that directly relate to any claim made by such third-party for which the Indemnified Party seeks indemnification under this <u>Article 16</u>, but only to the extent of any amount which the Indemnifying Party is liable to pay in satisfaction or settlement of such third-party claim under this <u>Article 16</u>. The Indemnified Party shall reasonably cooperate with the Indemnifying Party, at the Indemnifying Party's expense, in the assertion by the Indemnifying Party of any such claims against such third-party claimant.

## 16.7 Indemnification Payments.

(a) An Indemnifying Party shall not be liable in respect of any indemnification obligations under this Agreement until the cumulative aggregate amount of Indemnified Losses exceed \$100,000 ("<u>Indemnification Threshold Amount</u>"); provided, however, that once the Indemnification Threshold Amount has been exceeded, the Indemnified Party shall be entitled to recover the Indemnification Threshold Amount and any additional amounts owed pursuant to this <u>Article 16</u>.

(b) Amounts owing under this <u>Article 16</u> shall be paid promptly upon written demand for indemnification containing in reasonable detail the facts giving rise to such liability.

16.8 <u>Apportionment of Costs</u>. The Parties recognize and acknowledge that third-party claims may be made as part of an action, suit, investigation or proceeding that may give rise to the indemnification obligations of more than one (1) Party as set forth in <u>Section 16.1</u> (Indemnification of Bank by EFS) and <u>16.2</u> (Indemnification of EFS by Bank), or that may include allegations that are not subject to indemnification, and the Parties agree that they shall cooperate

in good faith to fairly apportion the Indemnified Losses relating to such third-party claims. Indemnified Losses incurred in defending third-party claims shall be apportioned to the respective Party that has responsibility for each specific third-party claim as set forth in <u>Section 16.1</u> (Indemnification of Bank by EFS) and <u>16.2</u> (Indemnification of EFS by Bank), but only to the extent that those Indemnified Losses directly arise from such third-party claim.

## 16.9 Limitation of Liability.

(a) In no event shall an Indemnifying Party be liable under this Agreement or any Product Schedule for any indirect, consequential, incidental, special, punitive or exemplary damages or lost profits, to an Indemnified Party, whether in contract, tort (whether in negligence or strict liability) or other legal or equitable theory, regardless of whether such Indemnifying Party knew or should have known of the possibility of such damages.

(b) Notwithstanding Section 16.9(a), an Indemnifying Party may be liable for damages suffered by all Indemnified Parties relating to lost profits or revenue of up to, but not in excess of, \$10,000,000, in the aggregate, for all Claims over the life of the Program, whether in contract, tort (whether in negligence or strict liability) or other legal or equitable theory, regardless of whether such Indemnifying Party knew or should have known of the possibility of such damages.

## ARTICLE 17 MISCELLANEOUS

17.1 <u>Assignment</u>. Any assignment by either Party of any of that Party's rights and obligations under this Agreement or any Product Schedule shall require the prior consent of the other Party to this Agreement, which consent shall not be unreasonably withheld.

17.2 Entire Agreement; Amendments. Except as it relates to the Purchase Agreement, this Agreement, together with all Exhibits and Schedules (including all Product Schedules) hereto, which are expressly incorporated by reference herein and made a part hereof, and the other agreements pertaining to the transactions contemplated hereunder and executed and delivered by a Party hereto to the others contemporaneous with the execution and delivery of this Agreement, are the entire agreement of the Parties with respect to the subject matter hereof and thereof, and supersede all other prior understandings and agreement of the Parties with respect to the subject matter hereof and thereof, whether written or oral. It is the intent and agreement of the Parties that this Agreement be strictly interpreted according to its express terms. This Agreement may not be amended except by a written instrument signed by EFS and Bank.

17.3 <u>No Third-Party Beneficiaries</u>. Except as set forth in Article 16 (Indemnification; Limitation of Liability), nothing in this Agreement is intended or shall be deemed to confer any rights or benefits upon any Person other than EFS and Bank and to make or render any such other Person a third-party beneficiary of this Agreement.

17.4 <u>Non-Waiver of Default</u>. The failure of either Party to insist, in any one or more instances, on the performance of any terms or conditions of this Agreement shall not be construed as a waiver or relinquishment of any rights granted hereunder or of the future performance of any such term or condition, and the obligations of the non-performing Party with respect thereto shall continue in full force and effect.

17.5 <u>Severability</u>. If any provision of this Agreement or any Product Schedule is held to be invalid, void or unenforceable, the Parties shall work in good faith to reform such provision and all other provisions shall remain valid and be enforced and construed to the extent permitted by law. The Parties intend all provisions of this Agreement to be enforced to the fullest extent permitted by Applicable Law. Accordingly, should a court of competent jurisdiction determine that the scope of any provision hereof is too broad to be enforced as written, the Parties intend that the court should reform the provision to such narrower scope as it determines to be enforceable.

17.6 <u>Further Assurances</u>. Each Party agrees to execute all such further documents and instruments and to do all such further things as the other Party may reasonably request in order to give effect and to consummate the transactions contemplated hereby, and to provide reasonable access to the other Party and the Regulatory Authority asserting jurisdiction over the other Party to the extent necessary for the other Party to comply with Applicable Law.

17.7 <u>Notices</u>. All notices, demands and other communications hereunder shall be in writing and shall be sent by electronic mail, certified mail return receipt requested, by hand, by facsimile with verbal confirmation of receipt, or by nationally recognized overnight courier service addressed to the Party to whom such notice or other communication is to be given or made at such Party's address as set forth below, or to such other address as such Party may designate in writing to the other Party from time to time in accordance with the provisions hereof as follows:

## If to EFS, to:

Emerald Financial Services, LLC One H&R Block Way Kansas City, MO 64105 Attn: Greg Quarles Email: greg.quarles@hrblock.com Telephone: 816-854-5709 Facsimile: 816-854-8052

Attn: David Zarski Email: david.zarski@hrblock.com Telephone: 816-854-4502 Facsimile: 816-802-1013

Attn: Walter Pirnot Email: wpirnot@hrblock.com Telephone: 816-854-5757 Facsimile: 816-802-1065

## with copies to:

H&R Block, Inc. One H&R Block Way Kansas City, Missouri 64105 Attn: Tom Gerke, Chief Legal Officer Email: tom.gerke@hrblock.com Telephone: 816-854-6060 Facsimile: 816-854-8500

## If to Bank, to:

Bofl Federal Bank 4350 La Jolla Village Drive, Suite 140 San Diego, California 92122 Attn: Gregory Garrabrants Email: ggarrabrants@bofifederalbank.com Telephone: 858-350-6203 Facsimile: 858-764-6561

#### with copies to:

BofI Federal Bank 4350 La Jolla Village Drive, Suite 140 San Diego, California 92122 Attn: Eshel Bar-Adon Email: ebaradon@bofifederalbank.com Telephone: 858-764-2905 Facsimile: 858-764-6561 Stinson Leonard Street LLP Attn: Mike Lochmann 1201 Walnut Street, Suite 2900 Kansas City, Missouri 64106 Email: mike.lochmann@stinsonleonard.com Telephone: 816-691-3208 Facsimile: 816-412-1249

Sheppard Mullin Richter & Hampton Attn: Lawrence M. Braun 333 South Hope Street, 43<sup>rd</sup> Floor Los Angeles, CA 90071-1422 Email: LBraun@sheppardmullin.com Telephone: 213-617-4184 Facsimile: 213-443-2814

or to such other Person or address as either Party shall have previously designated to the other by written notice given in the manner set forth in this <u>Section 17.7</u>. Notice shall be deemed to have been given on the day of delivery if (x) sent by certified mail; (y) sent by a nationally recognized overnight courier service; or (z) delivered by hand or via electronic mail. Any notice provided pursuant to this <u>Section 17.7</u> shall be effective on the day of completion of all delivery requirements.

17.8 Force Majeure.

(a) If the performance by a Party of its respective non-monetary obligations under this Agreement is delayed or prevented (in whole or in part) by acts of God, third-party cyber, IT or network attacks, fire, floods, storms, explosions, accidents, epidemics, war, civil disorder, strikes, terrorism, nuclear or biological disaster, riot or any other similar event or cause not reasonably within such Party's control, whether or not specifically mentioned herein (any such event, a "Force Majeure Event"), such Party shall be excused, discharged and released of performance to the extent such performance or obligation is so delayed or prevented by the Force

Majeure Event without liability of any kind; provided, however, that (i) the occurrence of a Force Majeure Event shall not excuse a Party from following the procedures set forth in its disaster recovery plan; and (ii) the Parties shall have the termination rights set forth in <u>Section 14.2(c)</u> (Force Majeure Event).

(b) The Party subject to a delay or prevention as contemplated herein shall, as soon as practicable and in all events within three (3) Business Days following the occurrence of a Force Majeure Event, notify the other Party of such Force Majeure Event, which notice shall set forth: (i) the nature of the Force Majeure Event; (ii) its expected effect(s) and duration; (iii) any expected development which may further affect performance hereunder; and (iv) the efforts undertaken or to be undertaken to cure such Force Majeure Event or provide substitute performance; provided, however, that failure of a Party to provide the notice required in this <u>Section 17.8(b)</u> shall not affect the operation of <u>Section 17.8(a)</u> to excuse, discharge and release a Party from performance as provided in such Section.

17.9 <u>Interpretation</u>. Each Party acknowledges that its legal counsel participated in the drafting of this Agreement and that this Agreement has been fully reviewed and negotiated by the Parties and their respective counsel. Accordingly, in interpreting this Agreement, no weight shall be placed upon which Party or its counsel drafted the provision being interpreted.

17.10 <u>Exhibits and Schedules</u>. Unless any provision hereof is specifically excluded or modified in a particular Exhibit or Schedule, including any Product Schedule, each such Exhibit or Schedule, including any Product Schedule, shall be deemed to incorporate therein all the terms and conditions of this Agreement and may contain such additional terms and conditions as the Parties may mutually agree in writing.

17.11 <u>No Agency</u>. Except as expressly provided in this Agreement or mutually agreed in writing by the Parties, nothing contained in this Agreement shall authorize, empower or constitute EFS or Bank as agent of the other in any manner; authorize or empower EFS and Bank to assume or create an obligation or responsibility whatsoever, express or implied, on behalf of or in the name of the other Party; or authorize or empower EFS and Bank to bind the other Party in any manner or make any representation, warranty, covenant, agreement or commitment on behalf of the other Party or permit EFS and Bank to hold itself out as having the authority to do any of the foregoing.

17.12 <u>Relationship of Parties</u>. EFS and Bank hereby acknowledge that it is not their intention to create between themselves a partnership, joint venture, fiduciary or employment or agency relationship for purposes of this Agreement, or for any other purpose whatsoever, and nothing contained in this Agreement shall be construed to constitute Bank and EFS as partners, joint venturers, principal and agent, or employer and employee. Neither Party shall hold itself in a capacity contrary to the terms of this Agreement, and neither Party shall become liable by reason of any representations, acts or omissions of the other contrary to the provisions hereof.

17.13 <u>Governing Law</u>. This Agreement and all rights and obligations hereunder, including matters of construction, validity and performance, shall be governed by and construed in accordance with the laws of the State of New York without regard to its conflict of laws provisions.

17.14 <u>Consent to Jurisdiction</u>. Except as otherwise expressly provided in this Agreement, any suit, action or proceeding seeking to enforce any provision of, or based on any

matter arising out of or in connection with, this Agreement or the transactions contemplated hereby shall be in the United States District Court for the Eastern District of Missouri, located in the City of St. Louis, or as to those lawsuits to which the Federal Courts of the United States lack subject matter jurisdiction, before a court located in the State of Missouri in the City of St. Louis, and each of the parties hereby consents to the jurisdiction of such courts (and of the appropriate appellate courts therefrom) in any such suit, action or proceeding and irrevocably waives, to the fullest extent permitted by Law, any objection which it may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such court or that any such suit, action or proceeding which is brought in any such court has been brought in an inconvenient forum. The parties acknowledge and agree that this Agreement was executed and delivered in the State of Missouri. Process in any such suit, action or proceeding may be served on any party anywhere in the world, whether within or without the jurisdiction of any such court. For the purposes of such actions and proceedings, service of process on a Party hereto shall be deemed effective if it is dispatched by United States first class mail to such Party's address provided in Section 17.7 (Notices).

# 17.15 <u>Waiver of Trial by Jury</u>. EACH PARTY TO THIS AGREEMENT WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF AN ACTION, SUIT OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT.

17.16 <u>Cumulative Remedies</u>; <u>Waivers</u>. Except as otherwise expressly provided herein, all remedies provided for in this Agreement shall be cumulative and in addition to and not in lieu of any other remedies available to either Party, whether at law, in equity, or otherwise. No release, discharge or waiver of any provision hereof shall be enforceable against or binding upon either Party unless in writing and executed by a duly authorized officer of each of the Parties. Neither the failure to insist upon strict performance of any of the agreements, terms, covenants or conditions hereof, nor the acceptance of monies due hereunder with knowledge of a breach of this Agreement, shall be deemed a waiver of any rights or remedies that either Party may have or a waiver of any subsequent breach or default in any of such agreements, terms, covenants and conditions.

17.17 <u>Binding Agreement</u>. This Agreement is legally binding on the Parties hereto, and their respective successors and permitted assigns.

17.18 <u>Survival</u>. The following provisions shall survive the expiration or termination of this Agreement: <u>Article 1</u> (Definitions; Order of Precedence; Rules of Interpretation), <u>Article 8</u> (Audit Rights; Reporting), <u>Article 9</u> (Intellectual Property; License to Use Marks; Ownership Rights), <u>Section 10.3(b)</u> (Non-Solicitation), <u>Article 11</u> (Confidentiality), <u>Article 12</u> (Privacy and Data Security), <u>Article 15</u> (Rights Upon Termination), <u>Article 16</u> (Indemnification; Limitation of Liability), and <u>Article 17</u> (Miscellaneous), any other provision identified in the survival section of any Product Schedules; and any other provision stated by its term to survive. In addition, any payment obligations of a Party that accrue prior to termination or expiration of this Agreement or a Product Schedule or prior to the Final Wind-Down Date shall survive such termination or expiration.

17.19 <u>Multiple Counterparts and Facsimile Signatures</u>. This Agreement may be executed in any number of multiple counterparts, all of which shall constitute but one and the same original. Facsimile signatures to this Agreement shall be effective.

(signature page follows)

# IN WITNESS WHEREOF, the Parties have duly executed this Agreement as of the day and year first above written.

# EMERALD FINANCIAL SERVICES, LLC

By:

Name: Title:

## BofI FEDERAL BANK

By:

Name: Title: List of Schedules and Similar Attachments Omitted from Program Management Agreement

Schedule 2.1(a)	Financial Products By Territory
Schedule 2.4(f)	EFS Audit Plan
Schedule 3.1(a)(i)	Initial Designated Executives
Schedule 3.1(b)(i)	Initial Senior Program Managers
Schedule 7.1	List of Internationally Outsourced Service Providers
Schedule 7.2(a)	Material Third Party Service Providers
Schedule 15.3	Calculation of Fair Value of Emerald Advance Participation Interests
Schedule A	Prepaid Products Product Schedule
Schedule B	Refund Transfer Product Schedule
Schedule C	Emerald Advance Product Schedule
Schedule D	Credit Card Product Schedule
Schedule E	Deposit Products Schedule
Schedule F	Service Level Agreements

Upon request, the registrant agrees to furnish supplementally to the Securities and Exchange Commission a copy of any omitted schedule or similar attachment to the Program Management Agreement; provided, however, that the registrant may request confidential treatment of omitted items prior to any public disclosure.

## EMERALD ADVANCE

# RECEIVABLES PARTICIPATION AGREEMENT

# By and Among

# EMERALD FINANCIAL SERVICES, LLC

# BofI FEDERAL BANK

# HRB PARTICIPANT I, LLC

And

H&R BLOCK, INC.

Dated as of

\_\_\_\_\_, 2014

## **Table of Contents**

Page

- 1. Definitions 2
- 2. Purchase Date; Participation; Settlement Report; Fair Value 7
- 3. Conditions Precedent to Purchase and Sale of Participation 9
- 4. Participant's Obligations 10
- 5. Payments on Account 10
- 6. Control of Collateral 11
- 7. Standard of Care 11
- 8. Costs and Expenses 12
- 9. Representations 12
- 10. Notices 14
- 11. Payment Instructions 16
- 12. Interest on Overdue Payments 17
- 13. Transfers of Participations; Permitted Assignments 17
- 14. Charged-Off Accounts 18
- 15. Indemnification 18
- 16. Force Majeure 21
- 17. Binding Agreement 21
- 18. Amendments; Entire Agreement 21
- 19. Confidentiality; Records 22
- 20. Governing Law 23
- 21. Consent to Jurisdiction 23
- 22. Waiver of Jury Trial 23
- 23. Further Assurances 23
- 24. Termination 24
- 25. No Relationship 24
- 26. Counterpart Execution 24
- 27. Severability 24
- 28. Underwriting Standards 24
- 29. Clarification of Intent 24
- 30. Guarantee 25

## EMERALD ADVANCE RECEIVABLES PARTICIPATION AGREEMENT

THIS EMERALD ADVANCE RECEIVABLES PARTICIPATION AGREEMENT (this '<u>Agreement</u>'), dated as of \_\_\_\_\_\_\_, 2014, is made by and among **EMERALD FINANCIAL SERVICES**, **LLC**, a limited liability company organized under the laws of Delaware (together with its successors and assigns, "<u>Servicer</u>''), **BofI FEDERAL BANK**, a federal savings bank (together with its successors and permitted assigns, "<u>BofI</u>''), **HRB PARTICIPANT I, LLC**, a Delaware limited liability company (together with its successors and permitted assigns, "<u>Participant</u>'') and **H&R BLOCK, INC.**, a Missouri corporation (together with its successors and permitted assigns, "<u>Cuarantor</u>''). Servicer, BofI and Participant are at times hereinafter referred to as the "<u>Parties</u>" and each individually as a "<u>Party</u>."

#### **RECITALS:**

A. HRB Tax Group, Inc., a Missouri corporation ("<u>HRB Tax Group</u>"), and certain of its subsidiaries are in the business of providing (or making available through franchisees) tax preparation and related products and services primarily to consumer customers ( "<u>Company Customers</u>") throughout the Program Territory.

B. HRB Tax Group, the Servicer and the Participant are indirect subsidiaries of the Guarantor (the Guarantor, together with all its subsidiaries, the "<u>Company</u>").

C. Contemporaneously with the execution and delivery of this Agreement, Servicer and BofI have entered into that certain Program Management Agreement, dated as of the date hereof (the "<u>PMA</u>"), under which Servicer will facilitate and service BofI's offering and administration of certain financial products as described in the PMA.

D. BofI will advance funds to or on behalf of Accountholders (as defined below) as part of the Emerald Advance as contemplated under the PMA, and thereby originate Receivables (as defined below).

E. BofI desires to sell and transfer to Participant from time to time, and Participant desires to purchase and accept from BofI from time to time, a Participation (as defined below) in all Receivables from time to time outstanding, on the terms and conditions hereof.

F. The Parties desire for Servicer to administer and service the Accounts (as defined below) and the Receivables pursuant to the PMA, and in accordance with the terms of this Agreement.

G. The Parties desire for Guarantor to guarantee all obligations, including the payment and performance of the obligations, of Participant under this Agreement, and Guarantor will derive good and sufficient consideration in the form of indirect and actual benefits from the transactions contemplated hereunder and is willing to guarantee the payment and performance of all of Participant's obligations hereunder.

## AGREEMENT

ACCORDINGLY, in consideration of the mutual covenants and agreements of the Parties herein contained and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

## 1. <u>Definitions.</u>

(a) Capitalized terms use in this Agreement shall have the meaning set forth below:

"<u>Account</u>" means an H&R Block Emerald Advance account established by BofI under the Emerald Advance program pursuant to Account Documentation, excluding, however, any legacy Emerald Advance account that was previously participated under the Legacy Participation Agreement.

"<u>Account Documentation</u>" means, with respect to any Account or Accounts, the applications, Accountholder agreements, instruments, disclosures, privacy notices, change of terms notices, including any and all amendments or modifications thereto, however stored or kept, and any other written information relating to such Account's terms and conditions.

"Accountholder" means any Person who holds or has held an Account as a customer of BofI.

"<u>Accrued Pre-Purchase Date Interest</u>" means, with respect to any Advances to be Participated on a Purchase Date, all Interest that has accrued but has not been paid with respect to such Advances for the period until (but excluding) such Purchase Date.

"Act" has the meaning set forth in Section 9(d).

"<u>Administration Fee</u>" means, a fee payable to BofI by Participant on each Purchase Date with respect to any Participation sold on such Purchase Date, in an amount equal to the product of (i) the Administration Fee Rate <u>times</u> (ii) the sum of (x) the Outstanding Advance Participation Amount and (y) the Outstanding Fee Participation Amount.

"Administration Fee Rate" means 0.40% (40 basis points).

"<u>Advances</u>" means, with respect to an Account, the principal amount of any funds advanced by BofI to or on behalf of the Accountholder under such Account, from time to time, to the extent not repaid.

"<u>Affiliate</u>" means any Person that, directly or indirectly, through one or more intermediaries, (a) owns or controls another Person, (b) is owned or controlled by another Person, or (c) is under common control or ownership with another Person, and "ownership" means the direct or indirect beneficial ownership of more than 30% of the equity securities of a Person, or, in the case of a Person that is not a corporation, more than 30% of the voting and/or equity interest of such Person.

"Agreement" means this Emerald Advance Receivables Participation Agreement.

"<u>Applicable Law</u>" means any and all laws, treaties, rules, regulations, regulatory guidance and determinations of a Regulatory Authority, mandatory written direction from a Regulatory Authority, and orders opinions and interpretations of any Regulatory Authority, including under the Bank Secrecy Act, laws relating to anti-money laundering, identity theft, fraud schemes, and predatory, unfair or deceptive acts, any and all sanctions or regulations enforced by OFAC, and statutes or regulations of any state relating to gift cards, money transmission or unclaimed property, that are applicable to the Program, or otherwise applicable to any of the Parties.

"<u>Assumed Obligations</u>" means, with respect to an Account and any related Receivables, all obligations of BofI with respect to or in connection with such Account and Receivables, including any obligation or liability owed to the related Accountholder in connection therewith.

"BofI" has the meaning set forth in the preamble.

"BofI Indemnified Party" has the meaning set forth in Section 15(a).

"BofI Percentage Interest" with respect to Accounts and Receivables participated under this Agreement, means 10%.

"Business Day" means any day, except Saturday, Sunday or federal legal holiday.

"<u>Charged-Off Accounts</u>" means any and all Accounts with respect to which Receivables have been charged off by BofI in accordance with its policies and procedures regarding charge-offs, as such policies and procedures may be established or amended, and are in effect from time to time.

"<u>Claim</u>" means any claim, demand, suit, legal action, regulatory action, administrative action, arbitration or proceeding, including those brought in connection with allegations of misrepresentations, breach of warranty, breach of contract, violation of Applicable Law, unfair or deceptive acts or practices, or otherwise seeking to recover Indemnified Losses.

"<u>Collateral</u>" means any property, whether real or personal, tangible or intangible, of whatever kind and wherever located, whether now owned or hereafter acquired or created, in or over which an any mortgage, pledge, lien, security interest, charge, hypothecation, security agreement, security arrangement or encumbrance has been, or is purported to have been, granted to (or otherwise created for) or for the benefit of BofI under any Account Documentation.

"Company" has the meaning set forth in the recitals.

"Company Customers" has the meaning set forth in the recitals.

"Confidential Information" means (i) information that is provided by or on behalf of a Party to another Party, its Affiliates or to a Service Provider, in connection with the Program, or (ii) information about a Party, its Affiliates, or their respective businesses or employees, that is otherwise obtained by another Party in connection with the Program, in each case including: (A) information concerning Program marketing plans, marketing philosophies, objectives and financial results; (B) information regarding business systems, methods, processes, financing data, programs and products; (C) information unrelated to the Program obtained by another Party in connection with this Agreement, including by accessing or being present at the business location of another Party; (D) proprietary technical information, including source and object codes; (E) competitive advantages and disadvantages, technological development, sales volumes, merchandise mix, business relationships and methods of transacting business, product design, product features and functionalities, operational and data processing capabilities, and systems software and hardware and the documentation thereof; (F) other information regarding the business or affairs of a Party or its Affiliates or the transactions contemplated by this Agreement that such other Party or its Affiliates reasonably considers confidential or proprietary; and (G) any copies, excerpts, summaries, analyses or notes of the foregoing. Subject to any disclosure required by Applicable Law, the Parties agree that the terms of this Agreement and the PMA shall be Confidential Information of the Parties.

"Disclosing Party" has the meaning set forth in Section 19(b).

"<u>Distribution</u>" means any collection or other recovery, whether by setoff or otherwise, of cash (including Interest and Fees) or other property (including Collateral) or proceeds under or in respect of any Account or Participated Receivable.

"Distributions Account" has the meaning set forth in Section 5(a).

"<u>Emerald Advance</u>" means an open-end line of credit offered by BofI under the Program whereby Company Customers may obtain credit, as further described in the Emerald Advance Product Schedule to the PMA.

"Emerald Advance Product Schedule" means Schedule C attached to the PMA.

"<u>Federal Funds Target Rate</u>" means, for any date that is a Business Day, the weighted average (rounded upward, if necessary, to the next 1/100 of 1%) of the rates set by the Federal Reserve Bank of New York on overnight federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers, as published on the next succeeding Business Day in *The Wall Street Journal* (Eastern Edition), or, if such rate is not so published for such Business Day, the average (rounded upward, if necessary, to the next 1/100 of 1%) of the quotations for such day for such transactions received by the Parties from three federal funds brokers of recognized standing selected by the Parties. For a day that is not a Business Day, the Federal Funds Target Rate shall be the rate applicable to federal funds transactions on the immediately preceding Business Day for which such rate is reported.

"<u>Fees</u>" means, with respect to an Account, the annual fee and any other fees (including any late payment fee and other similar fees) payable by the Accountholder under the applicable Account Documentation.

"Guarantee" has the meaning set forth in Section 30(a).

"Guaranteed Obligations" has the meaning set forth in Section 30(a).

"Guarantor" has the meaning set forth in the preamble.

"Indemnified Party" has the meaning set forth in Section 15(c).

"Indemnified Losses" means any and all losses, liabilities, costs and expenses of any kind, nature or description imposed or incurred in connection with this Agreement (including reasonable attorneys' fees and expenses, reasonable out-of-pocket costs, interest and penalties), settlements, equitable relief, judgments, damages (including liquidated damages), claims (including counter and crossclaims, and allegations whether or not proven) demands, offsets, defenses, actions, investigations or proceedings by whomsoever asserted (including Regulatory Authorities).

"Indemnifying Party" has the meaning set forth in Section 15(c).

"Information" has the meaning set forth in Section 9(a)(iv).

"<u>Initial Purchase Date</u>" means the date that is the earlier of November [\_\_\_], 2014 and one Business Day following the launch of the Promotional Period.

"<u>Interest</u>" means, with respect to an Account, all finance charges (excluding Fees) that are paid or payable by the Accountholder of such Account (including under the applicable Account Documentation).

"Legacy Participation Agreement" means the Amended and Restated Legacy Emerald Advance Participation Agreement, dated , 2014, by and among Servicer, BofI, Guarantor and HRB Products, LLC.

"Loan Participation Cash Flow" means, with respect to any calendar month and all Accounts and Participated Receivables under this Agreement, an amount, but not less than zero, equal to the aggregate amount of Distributions received by BofI during such month.

"Loan Participation Fair Value Percentage" means 100%.

"Monthly Statement" has the meaning set forth in Section 5(a).

"<u>Outstanding Advance Participation Amount</u>" means, with a respect to an Account subject to a Participation to be sold on a Purchase Date, the product of the (i) the sum of the aggregate amount of all outstanding Advances relating to such Account for which a Participation has not previously been sold to Participant, <u>plus</u> all applicable Accrued Pre-Purchase Date Interest relating to such Account, <u>times</u> (ii) the Participant Percentage Interest.

"Outstanding Fee Participation Amount" means the product of (i) the Fees with respect to Accounts for which a Participation has been sold by BofI to Participant pursuant to Section 2 and (ii) Participant Percentage Interest.

"Participant" has the meaning set forth in the preamble.

"Participant Indemnified Party" has the meaning set forth in Section 15(b).

"Participant Payment Breach" has the meaning set forth in Section 30.

"Participant Percentage Interest" with respect to Accounts and Receivables participated under this Agreement, means 90%.

"<u>Participated Receivables</u>" means the interest of the Participant in Receivables that have been participated pursuant to this Agreement.

"Participation" has the meaning set forth in Section 2(b).

"Participation Payment" has the meaning set forth in Section 2(a).

"<u>Person</u>" means and includes any individual, partnership, joint venture, corporation, company, bank, trust, unincorporated organization, government or any department, agency or instrumentality thereof.

"Potential Transferee" has the meaning set forth in Section 19(d).

"<u>Prior Payment</u>" has the meaning set forth in Section 5(c).

"Proceeds Subject to Return" has the meaning set forth in Section 5(c).

"Program" has the meaning set forth in the PMA.

"Program Management Agreement" or "PMA" has the meaning set forth in the recitals.

"Program Year" has the meaning ascribed thereto in the PMA.

"Promotional Period" means, for each Program Year, the period beginning early November and ending at the end of January.

"<u>Purchase Agreement</u>" means the Purchase and Assumption Agreement, dated \_\_\_\_\_\_, 2014, by and among BofI, H&R Block Bank and Block Financial, LLC.

"<u>Purchase Date</u>" means (i) the Initial Purchase Date, (ii) each Business Day (subsequent to the Initial Purchase Date) during the Promotional Period of each Program Year, (iii) the last Business Day of any month that is not a month in the Promotional Period of each Program Year; and (iv) any other Business Day during the term of this Agreement for which Participant delivers to BofI (in response to a Purchase Price Notice delivered by BofI to Participant) a notice indicating that Participant desires to acquire a Participation or Participations on such Business Day, as long as Participant delivers such notice at least one Business Day prior to the Business Day on which Participant wishes to acquire a Participation or Participations, as applicable.

"<u>Purchase Price</u>" means, with respect to a Participation sold on a Purchase Date, the product of (i) the Outstanding Advance Participation Amount, <u>times</u> (ii) the Loan Participation Fair Value Percentage.

"<u>Purchase Price Notice</u>" has the meaning set forth in Section 2(a).

"<u>Receivable</u>" means all past, present and future obligations of an Accountholder to pay money to BofI under the Account Documentation, including obligations for (i) repayment of Advances and (ii) payment of Interest and Fees. The Receivable balance recorded on the general ledger identifies the component Advances and Fees so as to distinguish between the two in the calculation of Purchase Price.

"<u>Receiving Party</u>" has the meaning set forth in Section 19(b).

"<u>Regulatory Authority</u>" means any federal, state or local, domestic, foreign or supranational governmental, regulatory or selfregulatory authority, agency, court, tribunal, commission or other governmental, regulatory or self-regulatory entity with jurisdiction over a Party.

"Regulatory Request" has the meaning set forth in Section 19(b).

"<u>Retained Receivable</u>" means all of BofI's right, title and interest in and to any (i) Receivable to the extent not yet Participated hereunder and (ii) any Participated Receivable solely to the extent of the BofI Percentage Interest in such Participated Receivable.

"<u>Return Notice</u>" has the meaning set forth in Section 5(c).

"<u>Service Provider</u>" means, with respect to a Party, a third-party service provider (including an Affiliate) (other than, in the case of BofI, the Servicer) used by such Party in connection with the performance of its obligations hereunder or with respect to Emerald Advance.

"Servicer" has the meaning set forth in the preamble.

"Servicing Fees" means all of the fees payable by BofI to Servicer for servicing the Accounts, as set forth in the Emerald Advance Product Schedule attached to the PMA.

"<u>Settlement Report</u>" has the meaning set forth in Section 2(e).

"Transfer" and "Transferred" have the meaning set forth in Section 13(a).

"<u>Underwriting Standards</u>" means the standards established by BofI from time to time for entry into Account Documentation and the advance of funds by BofI to Accountholders, as they may be determined and modified from time to time in accordance with Section 28.

(b) <u>Rules of Interpretation</u>. Unless otherwise expressly provided in this Agreement or the context otherwise requires, the following rules apply hereto:

(i) the singular includes the plural and the plural includes the singular;

(ii) all references to the masculine gender include the feminine gender (and vice versa);

(iii) "include," "includes" and "including" are not limiting and are deemed to be followed by the words "without limitation";

(iv) references to a particular agreement, instrument or document also refer to and include all renewals, extensions, modifications, amendments and restatements of such agreement, instrument or document;

(v) a reference in this Agreement to an Article, Section, Schedule or Exhibit is to the Article, Section, Schedule or Exhibit of or to this Agreement;

(vi) a reference to an Article or Section in this Agreement refers to all sub-parts or sub-components of any such article or section;

(vii) words such as "hereunder," "hereto," "hereof," and "herein," and other words of like import refer to the whole of this Agreement and not to any particular section, subsection or clause hereof;

(viii) a reference in this Agreement to a "party" or "parties" includes any Person;

(ix) the headings and subheadings of the sections of this Agreement are inserted for convenience of reference only and do not control or affect the meaning or construction of any of the agreements, terms, covenants and conditions of this Agreement in any manner;

(x) a reference to "unreasonably withheld" means "unreasonably withheld, delayed or conditioned;"

(xi) any approval, consent or notice required hereunder means "written approval," "written consent" or "written notice," as applicable; and

(xii) any reference made in this Agreement to Applicable Law means such Applicable Law as may be amended from time to time, and any successor Applicable Law relating to the same subject.

2. <u>Purchase Date; Participation; Settlement Report; Fair Value.</u>

(a) On each Purchase Date, BofI, or Servicer acting on behalf of BofI, shall provide Participant with a notice listing: (i) the Purchase Price for the aggregate Outstanding Advance Participation Amount available to be sold on such Purchase Date; (ii) the sum of (A) the aggregate Purchase Price for all Participations sold on such Purchase Date <u>plus</u> (B) the applicable Administration Fee (such sum, the "<u>Participation Payment</u>"); (iii) the related Accounts (identified by the identifying number assigned to such Account); (iv) the outstanding amount of each Advance to be Participated on such Purchase Date; (v) the amount of Accrued Pre-Purchase Date Interest related to each Advance to be Participated on such Purchase; and (vi) the outstanding amount of

each Fee to be Participated on such Purchase Date (the "<u>Purchase Price Notice</u>"). Each Purchase Price Notice shall be delivered by electronic mail or facsimile in accordance with Section 10 by no later than 10:00 a.m. (Central Time). For the avoidance of doubt, an example of the calculation of the Participation Payment is set forth on <u>Exhibit A</u>. Each Purchase Price Notice shall be deemed to have been accepted by, and shall become finally binding upon, Participant (i) absent demonstrable error attributable solely to BofI and not Servicer or Participant, and (ii) in any event, by 2:00 p.m. (Central Time) on the date of BofI's (or Servicer's) delivery of such Purchase Price Notice, unless Participant furnishes BofI with a written notice objecting to any of the information contained therein by no later than 2:00 p.m. (Central Time) on such date. No such objection to, or error in such Purchase Price Notice, shall excuse or release Participant of its obligations to Purchase Participations hereunder, but rather, if timely raised, may only result in a pricing adjustment to correct the error.

(b) On each Purchase Date and subject to (i) payment of the aggregate Participation Payment by Participant in accordance with Section 2(c); and (ii) (A) solely with respect to the Initial Purchase Date, the satisfaction or waiver of the conditions in Section 3(c), and (B) solely with respect to any Purchase Date other than the Initial Purchase Date, the satisfaction or waiver of the conditions in Section 3(d), Participant shall irrevocably purchase and accept, and BofI shall irrevocably sell, transfer and convey to Participant, an undivided participation interest equal to the Participant Percentage Interest in and to all Outstanding Advance Participation Amounts and Outstanding Fee Participation Amounts originated since (x) the date hereof, if such Purchase Date is the Initial Purchase Date, or (y) the immediately preceding Purchase Date, if such Purchase Date is a subsequent Purchase Date, as applicable (each such undivided participation interest, a "<u>Participation</u>"). Without limiting the foregoing, the Participations sold on any Purchase Date shall be the Participations identified on the Purchase Price Notice delivered with respect to such Purchase Date. Participations shall have the characteristics set forth in FASB Accounting Standards Codification (ASC) 860-10-40-6A (formerly FAS No. 166 paragraphs 8B(a) through (d)).

(c) On each Purchase Date and subject to (x) solely with respect to the Initial Purchase Date, the satisfaction or waiver of the conditions set forth in Section 3(a), and (y) solely with respect to any Purchase Date other than the Initial Purchase Date, the satisfaction or waiver of the conditions set forth in Section 3(b), Participant shall with respect to each Participation to be acquired by Participant on such Purchase Date (i) pay BofI by wire transfer of immediately available funds in accordance with Section 11 hereof without setoff, counterclaim or deduction of any kind, an amount equal to the Participation Payment, which wire transfer of funds shall be initiated by Participant not later than 3:00 p.m. (Central Time) on such Purchase Date, and (ii) in consideration for the payment of such Participation Payment, irrevocably purchase from BofI the Participation and irrevocably assume the Assumed Obligations with respect thereto.

(d) The Parties acknowledge and agree that, upon payment of the applicable Participation Payment to BofI and satisfaction of the conditions precedent referenced in Sections 2(b) and 2(c), BofI irrevocably sells, transfers and conveys to Participant, and Participant irrevocably acquires and accepts from BofI, a Participation (and the related Assumed Obligations) effective as of the time at which the applicable Participation Payment is received by BofI. As of such effective time, all risk of loss with respect to the applicable Participated Receivables is irrevocably transferred and conveyed to, and accepted and assumed by, Participant.

(e) Servicer shall deliver to Participant and BofI a report (the "<u>Settlement Report</u>") by electronic mail or facsimile in accordance with Section 10 not later than 3:00 p.m. (Central Time) on each Purchase Date containing the information called for in <u>Exhibit B</u>, including the Participation Payments to be made on the relevant Purchase Date. Servicer represents and warrants to Participant and BofI that the information contained in such Settlement Report shall be complete and correct in all material respects as of the applicable Purchase Date. Upon delivery, each Settlement Report shall be deemed to have been accepted by, and shall become finally binding upon, Participant (i) absent demonstrable error, and (ii) in any event, by 10:00 a.m. (Central Time) on the first Business Day after the date of BofI's (or Servicer's) delivery of such Settlement Report, unless Participant furnishes BofI with a written notice objecting to any of the information contained therein. No such objection to, or error in, such Settlement Report shall excuse or release Participant from its obligations hereunder to purchase Participations.

(f) The applicable Settlement Reports and Purchase Price Notices, together with this Agreement, shall be complete evidence of the purchase and sale of the Participations.

(g) Prior to the commencement of each Program Year, BofI and Participant shall negotiate in good faith to mutually confirm that the Purchase Price of Participations sold during such forthcoming Program Year will be equal to the fair value of such Participations.

(h) Nothing in this Agreement shall require BofI to establish any Account, make any Advance or originate any Receivable.

## 3. <u>Conditions Precedent to Purchase and Sale of Participation.</u>

(a) Solely with respect to the Participations to be acquired on the Initial Purchase Date, Participant's obligation to pay the Participation Payment in respect of such Participations to BofI and to acquire such Participations (and assume the related Assumed Obligations) on the Initial Purchase Date shall be subject to the conditions that: (i) Servicer has provided the applicable Purchase Price Notice to Participant in accordance with Section 2(a); and (ii) the PMA has been executed and delivered and is in full force and effect with respect to Emerald Advance product.

(b) With respect to any Participation to be acquired on a Purchase Date other than the Initial Purchase Date, Participant's obligation to pay the Participation Payment in respect of such Participation to BofI and to acquire such Participation (and assume the related Assumed Obligations) on such Purchase Date shall be subject to the conditions that: (i) Servicer has provided the applicable Purchase Price Notice to Participant in accordance with Section 2(a); and (ii) the PMA remains in full force and effect with respect to Emerald Advance product.

(c) Solely with respect to the Participations to be sold, transferred and conveyed on the Initial Purchase Date, BofI's obligations to sell, transfer and convey such Participations on the Initial Purchase Date shall be subject to the conditions that the PMA is in full force and effect with respect to Emerald Advance.

(d) With respect to any Participation to be sold, transferred and conveyed on a Purchase Date other than on the Initial Purchase Date, BofI's obligation to sell, transfer and convey such Participation on such Purchase Date shall be subject to the conditions that the PMA remains in full force and effect with respect to Emerald Advance.

## 4. <u>Participant's Obligations.</u>

From and after the applicable Purchase Date, Participant shall accept, assume, perform, and be obligated to pay to BofI, without setoff, counterclaim or deduction of any kind, any amounts required to be paid by BofI in respect of the Assumed Obligations. BofI shall remain the owner of the Retained Receivables.

## 5. <u>Payments on Account.</u>

(a) BofI shall establish a segregated deposit account in the name of "Emerald Financial Services, LLC, as Servicer" (the "<u>Distributions Account</u>") into which it will credit, on a daily basis, all funds received on the Accounts, including all Loan Participation Cash Flows. Servicer shall have the right to make deposits to and withdrawals from the Distribution Account, for the benefit of BofI and Participant, and manage all Advances and Distributions consistent with the terms of this Agreement. BofI shall charge no fees on the Distributions Account. BofI hereby waives all set off rights with respect to the Distributions Account. Servicer, on behalf of the BofI, shall account for and distribute all Distributions on each Business Day during the Promotional Period, and on a monthly basis during any month that is not during the Promotional Period, as follows:

(i) On each Business Day during the Promotional Period, Servicer, on behalf of BofI, shall prepare and send to BofI and Participant a statement showing (i) the Loan Participation Cash Flow received on (or as of) the previous Business Day and (ii) the service fees accrued with respect to the previous Business Day. On each Business Day during the Promotional Period, BofI shall cause the Servicer to distribute to Participant an amount equal to the product of (x) the Loan Participation Cash Flow for the previous Business Day, times (y) the Participant Percentage Interest; and

(ii) With respect to each month that is not in the Promotional Period, within ten (10) Business Days following the completion of such month, Servicer, on behalf of BofI, shall prepare and send to BofI and Participant a statement (the "<u>Monthly Statement</u>") showing (i) the Loan Participation Cash Flow received during such month and (ii) the service fees accrued with respect to such month. Within one Business Day of providing the Monthly Statement to Participant, BofI shall cause the Servicer to distribute to Participant an amount equal to the product of (x) the Loan Participation Cash Flow for such month, times (y) the Participant Percentage Interest.

For the avoidance of doubt, an example of the calculation of the payments to the Participant described in the previous sentence is set forth on Exhibit C.

(b) BofI, or Servicer acting on behalf of the BofI, shall hold all property to be distributed to Participant under Section 5(a) for benefit of Participant in a non-fiduciary capacity, and shall pay or deliver any amounts to be paid to Participant (according to the Participant Percentage Interest), without setoff, counterclaim or deduction of any kind except as required by Applicable Law or expressly provided hereunder. BofI shall have no legal, equitable or beneficial interest in such property.

(c) If BofI distributes any property to Participant pursuant to Section 5(a) (a "<u>Prior Payment</u>") and BofI is required to return to any Person all or any portion of any property in respect of which such Prior Payment was made (the "<u>Proceeds Subject to Return</u>"), Participant shall, upon

the written request (each such notice, a "<u>Return Notice</u>") of Bofl, or Servicer acting on behalf of Bofl, promptly return to Bofl such Proceeds Subject to Return. Each Return Notice shall be deemed to have been accepted by, and shall become finally binding upon, Participant (i) absent demonstrable error, and (ii) in any event, within 10 days of Bofl's (or Servicer's) delivery of such Return Notice, unless Participant furnishes Bofl with a written notice objecting to any of the information contained therein within such 10-day period.

## 6. <u>Control of Collateral.</u>

(a) BofI and Participant acknowledge that BofI is the depository institution with respect to Accounts and deposits serving as Collateral. Should any Accountholder default occur, unless otherwise required by Applicable Law or the Account Documentation, BofI will take instruction only from Servicer, and not from the Accountholder, concerning withdrawals from such Accountholder's Account.

(b) BofI hereby subordinates any security interest and right of set off that it may have in the Accounts and Participated Receivables, except to the interests of itself and Participant under this Agreement.

## 7. <u>Standard of Care.</u>

(a) Each of BofI and Servicer shall act in good faith and exercise the same care as it would exercise if it was the sole beneficial owner of the Accounts and Participated Receivables.

(b) Servicer, Participant and Guarantor acknowledge that the Participations are sold to Participant without recourse, and Participant expressly assumes all risk of loss in connection with such Participations, as if Participant had made loans in amounts equal to the Participant Percentage Interest in the Participated Receivables directly to the Accountholders. Without limiting the foregoing and notwithstanding any provision hereunder, BofI makes no representation or warranty as to, and shall not be responsible for: the enforceability, genuineness or collectability of any Account Documentation; any failure by any Accountholder or other obligor to perform its obligations thereunder; the collectability of any Receivable (including any Advance, Interest or Fees); any Accountholder's or other obligor's use of the proceeds therefrom; or the pres ervation of any Collateral or the loss, depreciation or release thereof; <u>provided, however</u>, that nothing contained in this Section 7(b) shall relieve BofI of its obligation to comply with the express terms and conditions of this Agreement or the PMA, or from any liability for any breach of its express representations, warranties, covenants or agreements contained herein, or from any liability arising out of its grossly negligent, willful or fraudulent acts or omissions as provided in Section 15. Servicer, Participant and Guarantor further acknowledge that subject to BofI's oversight and control, Servicer shall have operational responsibility for all aspects of servicing and collecting the Accounts, the Receivables and the transactions contemplated by this Agreement.

(c) BofI shall not be, or be deemed to be, a trustee, agent or fiduciary for Participant in connection with any extension of credit pursuant to any Account Documentation. Subject to the PMA, BofI or any of its Affiliates, may accept deposits from, make loans or otherwise extend credit to and generally engage in any kind of banking or other business with any Person that is an Accountholder, and receive payments on such extensions of credit and otherwise act with respect thereto freely and without accountability to Participant or any other Participant Indemnified Party,

in the same manner as if the Participations did not exist (and, in each case, without any additional notice to Participant).

(d) Except as permitted under the PMA or as required by Applicable Law, Bofl and Servicer shall not cause or permit (i) the amendment or modification of any Account Documentation or (ii) the waiver of any right thereunder, in each case, relating to any Participated Receivable, without the prior written consent of Participant, which consent shall not be unreasonably withheld.

## 8. <u>Costs and Expenses.</u>

(a) Each of BofI, Participant and Guarantor shall pay all of its own out-of-pocket fees and expenses, including attorneys' fees, incurred in connection with the preparation, negotiation and consummation of this Agreement and, except as otherwise provided for by this Agreement, the transactions contemplated hereby.

(b) BofI shall pay to Servicer the Servicing Fees in such amounts and at such times as set forth in the Emerald Advance Product Schedule attached to the PMA. No Servicing Fees are payable by Participant under this Agreement.

- (c) Servicer shall be responsible for payment of all fees payable to its Service Providers.
- 9. <u>Representations.</u>

(a) Each of BofI, Participant and Servicer (and Guarantor, solely with respect to Sections 9(a)(i) through (iii)) represents, warrants, acknowledges and agrees to and with the others, as of the date hereof, and as of each Purchase Date, that:

(i) It is duly organized and validly existing under the laws of its jurisdiction of organization or incorporation, has the banking, corporate or company power and authority, and has taken all action necessary, to execute, deliver and perform this Agreement and all documents required to be executed and delivered by it in connection herewith, to fulfill its obligations hereunder and thereunder, and to consummate the transactions contemplated hereby and thereby.

(ii) The making and performance by it of this Agreement and all documents required to be executed and delivered by it in connection herewith, and fulfillment of its obligations hereunder and thereunder, do not violate any Applicable Law or any writ, order, judgment, injunction, decree or determination applicable to it, or constitute a breach or default of any material agreement to which it is a party or by which it is bound, or contravene any provision of any document under which it was organized, and all authorizations, consents, orders, approvals, registrations or declarations required to be obtained from any Regulatory Authority in connection with the making and performance by it of this Agreement and all documents required to be executed and delivered by it in connection herewith, and fulfillment of its obligations hereunder and thereunder, have been obtained.

(iii) This Agreement and all documents required to be executed hereunder on or before the date hereof or such Purchase Date, as applicable, have been duly and validly authorized, executed and delivered by it and constitute its legal, valid and binding obligations, enforceable (subject to any bankruptcy, insolvency, reorganization, restructuring, moratorium or similar laws affecting creditors' rights generally and to a court's

discretion in relation to equitable remedies) against it in accordance with the respective terms hereof and thereof.

(iv) Another Party or such other Party's Affiliates may be in possession of information concerning one or more Accounts or Accountholders (the "<u>Information</u>") which may not be known to it, and the other Party shall have no liability whatsoever to it based on such other Party's or any of such other Party's Affiliate's knowledge, use, possession or nondisclosure of the Information, and it waives and releases any claims that it might have against the other Party and such other Party's Affiliates with respect to the nondisclosure of the Information in connection with the purchase and sale of Participations hereunder and the transactions described herein; <u>provided</u>, <u>however</u>, that the foregoing shall not in any way impair or limit the express representations and warranties (if any) made by the other Party or such other Party's Affiliates hereunder or under the PMA or the obligations (if any) of the other Party or such other Party's Affiliates hereunder.

(v) It (i) is a sophisticated Person with respect to the purchase or sale of the Participations (as applicable) and the retention of the Retained Receivables or assumption of the Assumed Obligations (as applicable), (ii) has adequate information concerning the financial condition of the Accountholders to make an informed decision regarding the purchase or sale of the Participations (as applicable) and the retention of the Retained Receivables or the assumption of the Assumed Obligations (as applicable) and the retention of the Retained Receivables or the assumption of the Assumed Obligations (as applicable) and (iii) has independently and without reliance upon the other Party or such other Party's Affiliates, and based on such information as it has deemed appropriate, made its own analysis and decision to enter into this Agreement, except that it has relied upon the express representations, warranties, covenants and indemnities of the other Parties set forth in this Agreement and the PMA.

(vi) No Party or Affiliate of such Party has given to any other Party any investment, tax or accounting advice or opinion on whether the purchase or sale of any Participation or the retention of the Retained Receivables or assumption of the Assumed Obligations (as applicable) is prudent.

(b) BofI further represents, warrants and acknowledges and agrees to and with Participant as of each Purchase Date that immediately prior to the sale of Participations to be sold on such Purchase Date, BofI is the legal and beneficial owner of and has good title to the Accounts and Receivables, free and clear of any mortgage, pledge, lien, security interest, charge, hypothecation, security agreement, security arrangement or encumbrance or other adverse claim against title of any kind; any purchase, option, call or put agreement or arrangement(except this Agreement); any prior sale, transfer, assignment or participation by BofI; or any agreement to create or effect any of the foregoing (other than this Agreement).

(c) Participant further represents, warrants, acknowledges and agrees to and with BofI as of each Purchase Date, and as of each date on which any amount of Loan Participation Cash Flow is to be paid or delivered to Participant, that Participant is entitled to receive any payments and distributions to be made to it hereunder without the withholding of any tax.

(d) Without characterizing any Participation as a "security" within the meaning of Securities Act of 1933 (the "<u>Act</u>") or any state "blue sky" law, Participant represents to BofI that



it is acquiring each Participation solely for its own account for investment purposes only and with no present intention of selling or distributing the same publicly or making any further distribution thereof in violation of the Act; <u>provided</u>, <u>however</u>, the Participant may Transfer (as defined below) any Participation if such Transfer is in accordance with the Act, such laws and the provisions of this Agreement.

(e) Except as expressly provided in this Agreement, none of the Parties make any representations or warranties, express or implied, with respect to the transactions contemplated herein. Bofl and Participant acknowledge that: (i) the sale of any Participation pursuant to this Agreement (A) is irrevocable, (B) is intended by Bofl and Participant to be a true sale of Bofl's right, title and interest in and to the Participated Receivables to the extent of such Participation sold hereunder and shall be treated as such for accounting purposes by Bofl and Participant, (C) is not intended to be a lending transaction or extension of credit from Participant to Bofl and (D) is intended to transfer and convey risk of loss to the extent of each Participation from Bofl to Participant (other than as expressly provided herein); (ii) the amount of the Purchase Price for any Participation is equal to or greater than the fair value of such Participation as of the applicable Purchase Date; and (iii) neither Bofl nor Participant shall have any recourse to one another (or in Bofl's case, to Guarantor) with respect to this Agreement or any Participation sold pursuant thereto, except as expressly provided herein. Without limiting the foregoing, Bofl shall have no liability for, and Participant will have no recourse (credit or otherwise) to Bofl with respect to, the performance of any Participation except as expressly provided herein.

#### 10. <u>Notices.</u>

All notices required or permitted under this Agreement must be given in writing, reference this Agreement and will be deemed delivered and given (i) upon personal delivery to the Guarantor or the party to be notified; (ii) on the date sent by facsimile or email (with confirmation of transmission); (iii) on the date received if sent by registered or certified U.S. mail, return receipt requested, postage and charges prepaid; (iii) one Business Day after deposit with a nationally-recognized commercial overnight courier, specifying next day delivery, with written verification of receipt. All communications shall be sent to the following addresses (or at such other address as shall be specified by like notice):

## If to Participant, to:

HRB Participant I, LLC One H&R Block Way Kansas City, MO 64105 Attn: [•] Email: [•] Telephone: [•] Facsimile: [•]

Attn: Walter Pirnot Email: wpirnot@hrblock.com Telephone: 816-854-5757 Facsimile: 816-802-1065

#### with copies to:

H&R Block, Inc. One H&R Block Way Kansas City, Missouri 64105 Attn: Tom Gerke, Chief Legal Officer Email: tom.gerke@hrblock.com Telephone: 816-854-6060 Facsimile: 816-854-8500

If to BofI, to:

BofI Federal Bank 4350 La Jolla Village Drive, Suite 140 San Diego, California 92122 Attn: Gregory Garrabrants Email: ggarrabrants@bofifederalbank.com Telephone: 858-350-6203 Facsimile: 858-345-0460

#### With copies to:

Boff Federal BankSheppard Mullin4350 La Jolla Village Drive, Suite 140Attn: Lawrence M. BraunSan Diego, California 92122333 South Hope StreetAttn: Eshel Bar-AdonForty-Third FloorEmail: EBar-Adon@boffederalbank.comLos Angeles, CA 90071Telephone: 858-764-2905Telephone: 213-617-4184Facsimile: 858-740-1675Facsimile: 213-443-2814

## If to Servicer, to:

Emerald Financial Services, LLC One H&R Block Way Kansas City, Missouri 64105 Attn: Greg Quarles Email: greg.quarles@hrblock.com Telephone: 816-854-5709 Facsimile: 816-854-8052

Attn: Walter Pirnot Email: wpirnot@hrblock.com Telephone: 816-854-5757 Facsimile: 816-802-1065 Stinson Leonard Street LLP Attn: Mike Lochmann 1201 Walnut Street, Suite 2900 Kansas City, Missouri 64106 Email: mike.lochmann@stinsonleonard.com Telephone: 816-691-3208 Facsimile: 816-412-1249

#### with copies to:

H&R Block, Inc. One H&R Block Way Kansas City, Missouri 64105 Attn: Tom Gerke, Chief Legal Officer Email: tom.gerke@hrblock.com Telephone: 816-854-6060 Facsimile: 816-854-8500 Stinson Leonard Street LLP Attn: Mike Lochmann 1201 Walnut Street, Suite 2900 Kansas City, Missouri 64106 Email: mike.lochmann@stinsonleonard.com Telephone: 816-691-3208 Facsimile: 816-412-1249

## If to Guarantor, to:

H&R Block, Inc. One H&R Block Way Kansas City, Missouri 64105

Attn: Tom Gerke, Chief Legal Officer Email: tom.gerke@hrblock.com Telephone: 816-854-6060 Facsimile: 816-854-8500

Attn: Walter Pirnot Email: wpirnot@hrblock.com Telephone: 816-854-5757 Facsimile: 816-802-1065

#### with a copy to:

H&R Block, Inc. One H&R Block Way Kansas City, Missouri 64105 Attn: Tom Gerke, Chief Legal Officer Email: tom.gerke@hrblock.com Telephone: 816-854-6060 Facsimile: 816-854-8500 Stinson Leonard Street LLP Attn: Mike Lochmann 1201 Walnut Street, Suite 2900 Kansas City, Missouri 64106 Email: mike.lochmann@stinsonleonard.com Telephone: 816-691-3208 Facsimile: 816-412-1249

provided, however, that the providing of notice to counsel shall not, of itself, be deemed the providing of notice to the Guarantor or a party hereto. Any notice provided pursuant to this Section 10 shall be effective on the date of completion of all delivery requirements.

11. <u>Payment Instructions.</u>

All payments from one Party to another in connection with this Agreement shall be made in accordance with the payment instructions set forth below:

## BofI:

ABA Number: [•] Account Number: [•] Reference: [•] Attention: [•]

## **Participant:**

ABA Number: [•] Account Number: [•] Reference: [•] Attention: [•]

## Servicer:

ABA Number: [•] Account Number: [•] Reference: [•] Attention: [•]

## **Guarantor:**

ABA Number: [•] Account Number: [•] Reference: [•] Attention: [•]

## 12. Interest on Overdue Payments.

If any payment hereunder is not paid by any Party to another Party when due hereunder, then interest shall accrue, and be payable immediately on demand, on all such amounts at a per-annum rate equal to (a) the Federal Funds Target Rate during the first five days following the date on which payment is due and (b) the Federal Funds Target Rate plus 2% per annum during all times thereafter until payment is made, in each case, accruing from and including the due date to but excluding the date the payment is made; <u>provided</u>, <u>however</u>, that no such interest shall accrue if any such payment is made to the Party to which it is due within one (1) Business Day from the due date thereof.

## 13. <u>Transfers of Participations; Permitted Assignments.</u>

(a) Participant may sell, assign, pledge, subparticipate or otherwise transfer (each, a "<u>Transfer</u>") to any Person any Participation, or any part thereof or interest therein; <u>provided</u>, <u>however</u>, that the transferring Participant provides to BofI and Servicer a copy of the executed assignment agreement that identifies the transfere participant, its address for notices, its bank account and wire transfer information, its EIN and other relevant information reasonably requested by BofI or Servicer, and the transferee Participant agrees to the applicable terms of this Agreement (including the confidentiality provisions of Section 19).

(b) Except as provided in Section 13(c), Participant may not assign or delegate (x) its obligations hereunder (including its purchase obligations set forth in Section 2 hereof) or (y) its rights to purchase Participations under Section 2 hereof, in either case, without prior written consent by BofI. In the event of any such assignment made with the prior written consent of the BofI, the

obligations of the assignee Participant will not be Guaranteed Obligations covered by the Guarantee in Section 30.

(c) Notwithstanding Section 13(b), Participant may assign or delegate (x) its obligations hereunder (including its purchase obligation set forth in Section 2 hereof) or (y) its rights to purchase Participations under Section 2 hereof, in either case, without the prior written consent of the BofI, if such assignment or delegation is made to any direct or indirect subsidiary of the Guarantor.

(d) In the event of any assignment or delegation, Participant shall remain directly liable to BofI under this Agreement until (1) the assigning or delegating Participant provides to BofI and Servicer a copy of the executed assignment agreement that identifies the assignee Participant, its address for notices, its bank account and wire transfer information, its EIN and other relevant information reasonably requested by BofI or Servicer, and the assignee Participant agrees to the applicable terms of this Agreement (including the confidentiality provision of Section 19, and (2) if required by this Section 13, the BofI consents to such assignment.

(e) BofI may Transfer to any Person a participation interest in any Retained Receivable; provided, however, that BofI shall remain subject to its obligations to Participant under this Agreement with respect to the Participations and the Participated Receivables. Subject to the preceding sentence, BofI may not Transfer to any Person any of its rights under this Agreement or any Account Documentation or with respect to any Participated Accounts without the prior written consent of Participant, which shall not be unreasonably withheld.

(f) BofI and Servicer agree to cooperate with and provide further assistance to Participant with respect to any Transfer of a Participation that is permitted under Section 13(a) of this Agreement. From time to time at the request of Participant, BofI and Servicer, without further consideration, shall (i) execute, acknowledge, and deliver, or cause to be executed, acknowledged and delivered, all such instruments and take such actions as may be reasonably necessary and appropriate, or reasonably requested by Participant, in order to effect such permitted Transfer to the transferee participant equitable title to the Transferred Participation, and put the transferee participant in possession of the Transferred Participation, and (ii) use communally reasonable efforts to otherwise assist Participant in the orderly Transfer of any such Participation.

### 14. <u>Charged-Off Accounts.</u>

Bofl, Servicer, and Participant acknowledge and agree that: (a) Receivables arising from Charged-Off Accounts may require a material period of time to collect and/or be uncollectable; (b) Bofl and Servicer shall continue to administer the Charged-Off Accounts until the earliest of (1) the collection of the Receivables relating to such Charged-Off Account, or (2) the fulfillment of their obligations under the Account Documentation with respect to such Charged-Off Account following the termination of this Agreement; and (c) Servicer may continue to use Bofl's name in connection with its collections activity with respect to each Charged-Off Account until collection of the Receivables relating to such Charged-Off Account.

15. <u>Indemnification.</u>

(a) Participant agrees to protect, indemnify, defend and hold harmless BofI, its Affiliates, its Service Providers and their respective shareholders, directors, officers, employees, agents, representatives and permitted assigns (each, a " <u>BofI Indemnified</u> <u>Party</u>"), from and against any and

all Indemnified Losses to the extent such Indemnified Losses arise out of or result from (i) any breach by Participant of any representation, warranty, covenant or other provision contained in this Agreement; (ii) the negligent, willful or fraudulent acts or omissions of Participant in connection with matters contemplated by this Agreement or (iii) the failure of Guarantor to perform its obligations under the Guarantee in Section 30; provided, however, that in no event shall Participant be obligated to indemnify any BofI Indemnified Party under this Section 15(b) against any Indemnified Losses to the extent such Indemnified Losses result from the grossly negligent, willful or fraudulent acts or omissions of BofI, its Affiliates or any of their respective Service Providers. In addition, should BofI be sued or threatened by suit by any receiver or trustee in connection with the bankruptcy of any Accountholder, or directly by any Accountholder as a debtor-in-possession, on account of any alleged preference, voidable transfer or fraudulent conveyance alleged to have been received as the result of any transaction in respect of which Participant has participated with Bank hereunder, or if any claim, suit or action shall be asserted against the Bank relating thereto, then in such event, Participant shall indemnify Bank for any money paid by Bank in satisfaction or compromise of such suit, action or demand, any money required to be returned by Bank to such Borrower or its estate, and any costs or fees associated therewith, in each case in proportion to the Participant Percentage Interest (provided that any subsequent payment received by Bank in such bankruptcy proceeding vis-a-vis the Account shall be a recovery and a Distribution with respect to which Participant shall receive its Participant Percentage Interest).

(b) BofI agrees to protect, indemnify, defend and hold harmless Participant and Guarantor, and their respective shareholders, directors, officers, employees, agents, representatives and permitted assigns (each, a "<u>Participant Indemnified Party</u>"), from and against any and all Indemnified Losses to the extent such Indemnified Losses arise out of or result from the grossly negligent, willful or fraudulent act or omission of BofI or its Affiliates in connection with matters contemplated by this Agreement; provided, however, that in no event shall BofI be obligated to indemnify any Participant Indemnified Party under this Section 15(a) against any Indemnified Losses to the extent such Indemnified Losses result from the grossly negligent, willful or fraudulent acts or omissions of Servicer, Participant, Guarantor, or any of their respective Affiliates or Service Providers.

(c) If a Participant Indemnified Party or a BofI Indemnified Party (in each case, the "<u>Indemnified Party</u>") receives notice of any third-party claim for which indemnification may be available under this Agreement, the Indemnified Party must promptly notify the indemnifying Party (the "<u>Indemnifying Party</u>") in writing of the third-party claim, including, to the extent readily available, an estimate of the amount of liability arising from it. The Indemnified Party shall use its commercially reasonable efforts to provide notice to the Indemnifying Party no later than 15 days after receipt of a complaint, summons or similar pleading by the Indemnified Party in the event a suit or action has commenced, or 30 days under all other circumstances; provided, however, that the failure to give such notice shall not relieve an Indemnifying Party of its obligation to indemnify under this Section 15, except to the extent the Indemnifying Party is materially prejudiced by such failure.

(d) Subject to Section 15(e), the Indemnifying Party shall have the right to defend any such third-party claim at its expense and in the name of the Indemnified Party and shall select the counsel for the defense of such third-party claim as approved by the Indemnified Party, such approval not to be unreasonably withheld, and the Indemnified Party shall reasonably cooperate with the

Indemnifying Party in the conduct of the defense against such third-party claim. The Indemnified Party may participate, at its own expense, in such defense and in any settlement discussions directly or through counsel of its choice on a monitoring, non-controlling basis, or at the Indemnifying Party's expense and with full control if the Indemnifying Party does not fulfill its obligations to appoint counsel to defend the Indemnified Party which is reasonably satisfactory to the Indemnified Party within a reasonable time after the Indemnifying Party has received written notice of such third-party claim from the Indemnified Party. Bofl and Participant agree to cooperate in good faith to coordinate the defense of any third-party claim that may give rise to indemnification obligations of more than one Indemnifying Party or that may include allegations that are not subject to indemnification.

(e) Notwithstanding Section 15(d), the Indemnifying Party shall not have the right to defend any such third-party claim on behalf of the Indemnified Party if: (i) it contests (in whole or in part) its indemnification obligations (but only as to the obligations specific to the Indemnifying Party's obligations to indemnify under this Section 15 in the event a third-party claim gives rise to indemnification obligations of more than one Indemnifying Party); (ii) it fails to employ counsel approved by the Indemnified Party (such approval not to be unreasonably withheld) to assume the defense of such third-party claim or refuses to replace such counsel upon the Indemnified Party's reasonable request; (iii) the Indemnified Party reasonably determines that there are issues which could raise possible conflicts of interest between the Indemnifying Party and the Indemnified Party or that the Indemnified Party has claims or defenses that are separate from or in addition to the claims or defenses of the Indemnified Party; or (iv) such third-party claim seeks an injunction, cease and desist order, or other equitable relief against the Indemnified Party. In each such case described in clauses (i) - (iv) above, the Indemnified Party shall have the right to direct the defense of the third-party claim and retain its own counsel, and the Indemnifying Party shall pay the cost of such defense, including reasonable attorneys' fees and expenses. If the Indemnifying Party does not assume the defense of any such third-party claim or refuses to replace such counsel upon the Indemnified Party's reasonable request as provided in this Section 15, the Indemnifying Party shall have the right to participate, at its own expense, in the defense of such third-party claim with counsel deemed satisfactory to it in its sole discretion.

(f) If the Indemnifying Party assumes the defense of any such third-party claim, the Indemnifying Party shall be subrogated to any counterclaims, claims in recoupment or similar rights of the Indemnified Party as against any other Persons to the extent of any amount which the Indemnifying Party is liable to pay in satisfaction or settlement of such third-party claim under this Section 15. The Indemnified Party shall reasonably cooperate with the Indemnifying Party, at the Indemnifying Party's expense, in the assertion by the Indemnifying Party of any such claim or right against such other Persons.

(g) The Indemnifying Party shall not be liable for any settlement of any such third-party claim effected without its written consent (which consent shall not be unreasonably withheld) but, if settled with such consent or if there is a judgment for the plaintiff, the Indemnifying Party shall indemnify the Indemnified Party from and against any Indemnified Loss by reason of such settlement or judgment to the extent required by Section 15(a) or 15(b), as applicable. If the Indemnifying Party assumes the defense of any such third-party claim, it shall be entitled to settle such third-party claim (i) with the consent of the Indemnified Party (which consent shall not be unreasonably withheld) or, (ii) if such settlement provides for an unconditional, irrevocable release of the

Indemnified Party in connection with all matters relating to the action or proceeding that have been asserted against such Indemnified Party in connection with such third-party claim by the other parties to such settlement, and such release does not include a statement as to or an admission of fault, culpability or a failure to act by or on behalf of such Indemnified Party, without the consent of such Indemnified Party. The amount paid or payable by an Indemnified Party as a result of the Indemnified Losses shall be deemed to include, subject to the limitations set forth above, any legal or other expenses reasonably incurred by such Indemnified Party in connection with investigating or defending any such third-party claim, except where the Indemnified Party is required to bear such expenses pursuant to this Section 15, which expenses the Indemnifying Party shall pay as and when incurred, at the request of the Indemnified Party, to the extent it is reasonable to believe that the Indemnifying Party will be ultimately obligated to pay such expenses. If any expenses so paid by the Indemnifying Party are subsequently determined to not be required to be borne by the Indemnifying Party under this Section 15, the Indemnifying Party that received the benefit of such payment shall promptly refund the amount so paid to the Indemnifying Party.

(h) BofI and Participant recognize and acknowledge that third-party claims may be made as part of an action, suit, investigation or proceeding that may give rise to the indemnification obligations of more than one Indemnifying Party or that may include allegations that are not subject to indemnification, and the Indemnifying Parties agree that they shall cooperate in good faith to fairly apportion the Indemnified Losses relating to such third-party claims. Indemnified Losses incurred in defending third-party claims shall be apportioned to the respective Indemnifying Party that has responsibility under this Section 15 for indemnification as to each specific third-party claim, but only to the extent that those Indemnified Losses directly arise from such third-party claim.

# 16. <u>Force Majeure.</u>

If either of BofI, on the one hand, or any of Participant, Servicer or Guarantor, on the other hand, is unable to carry out the whole or any part of its obligations under this Agreement for any reason beyond its respective control, including, but not limited to, acts of God, acts of governmental authorities, strikes, war, riot, terrorism and any other causes of such nature, then its respective performance of the obligations under this Agreement as it is affected by such cause is to be excused during the continuance of the inability so caused.

# 17. <u>Binding Agreement.</u>

This Agreement, including the representations, warranties, covenants and indemnities contained herein, shall be binding upon and inure to the benefit of the Parties and their successors and permitted assigns. All representations, warranties and indemnities made in this Agreement shall survive the execution, delivery and performance of this Agreement.

18. <u>Amendments; Entire Agreement.</u>

This Agreement, together with the PMA and all Exhibits and Schedules thereto, is the entire agreement of the Parties with respect to the subject matter hereof and supersedes all other prior understandings and agreements among the Parties and the Guarantor with respect to the subject matter hereof, whether written or oral. This Agreement may not be amended except by written instrument signed by the Parties and with respect of Section 30, the Guarantor.

## 19. <u>Confidentiality; Records.</u>

(a) Each Party shall cause its respective directors, officers, employees, representatives, Affiliates and Service Providers, to comply with the provisions of this Section 19. Notwithstanding anything to the contrary contained herein, for purposes of this Section 19, Participant, Guarantor and Servicer shall be treated as a single Party.

(b) If a Party receives Confidential Information (the "<u>Receiving Party</u>") of the other Party (the "<u>Disclosing Party</u>"), the Receiving Party shall not use or disclose Confidential Information of the applicable Disclosing Party except: (i) to perform its obligations or enforce its rights with respect to Accounts, the Account Documentation, this Agreement or the PMA; (ii) as expressly permitted by this Agreement or the PMA; (iii) with the prior consent of the Disclosing Party; (iv) pursuant to a subpoena, summons or other order requesting information that is issued to the Receiving Party by any Regulatory Authority (including routine and supervisory examinations by Regulatory Authorities) ("<u>Regulatory Request</u>"); (v) based on advice of legal counsel, to the extent the Receiving Party is required by Applicable Law or valid court order or Regulatory Authority order to disclose, or (vi) as otherwise required by Applicable Law; <u>provided however</u>, that in the case of clauses (iv) - (vi) (other than routine and supervisory examinations by Regulatory Authorities), to the extent permitted or required by Applicable Law, the Receiving Party shall give prompt notice to the Disclosing Party to allow it an opportunity to seek a protective order.

(c) A Receiving Party shall: (i) limit access to the Disclosing Party's Confidential Information to (x) those employees, professionals, Affiliates and Service Providers, that have a reasonable need to access such Confidential Information in connection with Accounts or other purposes permitted by this Agreement or the PMA and (y) to Potential Transferees as provided under Section 19(d) and (ii) obtain contractual or other enforceable confidentiality commitments substantially similar to those set forth in this Section 19 from those of its Affiliates, Service Providers and Potential Transferees, as applicable, to which such Receiving Party provides access to the Disclosing Party's Confidential Information.

(d) Notwithstanding anything else contained in this Section 19, (i) Participant may disclose the contents of this Agreement to any proposed assignee, participant, subparticipant or other transferee of, or to any proposed counterparty in a loan or securitization transaction with respect to, all or any part of its interest in a Participation and (ii) BofI may disclose the contents of this Agreement to any proposed purchaser of a participation interest in the Retained Receivables or any proposed purchaser of an Account (in each case, a "Potential Transferee"), in each case, in connection with a Transfer permitted under Section 13. Such permitted disclosure may also be made to professionals employed or engaged by such entities; provided, however, that each such Person or entity agrees to comply with the requirements of the applicable Account Documentation regarding confidentiality. At the request of Participant, BofI (or Servicer acting on behalf of BofI) shall, subject to Applicable Law and any applicable confidentiality restrictions contained in the Account Documentation, use commercially reasonable efforts to deliver or otherwise make available to Participant copies of all written information and documents received by BofI in its capacity as a lender under the Accounts, or Servicer acting on behalf of BofI, from time to time with respect to the Account Documentation and the Receivables as soon as practicable after the same are received by BofI or Servicer, as applicable.

(e) Except as otherwise permitted by this Section 19 and any disclosure required by Applicable Law, the Parties agree that the terms of this Agreement shall be Confidential Information of each Party.

(f) Notwithstanding anything else contained in this Agreement, a Party will not be obligated to take any action with respect to the collection, use or disclosure of Confidential Information in connection with the Accounts, the Account Documentation or this Agreement that such Party reasonably believes in good faith would cause, or is reasonably likely to cause, such Party to violate any Applicable Law.

## 20. <u>Governing Law.</u>

This Agreement and all rights and obligations hereunder, including matters of construction, validity and performance, shall be governed by and construed in accordance with the laws of the State of New York without regard to its conflict of laws provisions.

21. <u>Consent to Jurisdiction.</u>

(a) Each party hereto agrees that all actions, proceedings or counterclaims arising out of or relating to this Agreement or any of the transactions contemplated hereby shall be brought in the United States District Court for the Eastern District of Missouri located in the City of St. Louis, or as to those lawsuits to which the Federal Courts of the United States lack subject matter jurisdiction, before a court located in the State of Missouri in the City of St. Louis, and each of the parties irrevocably submits to the exclusive jurisdiction of each such court in any proceeding, waives any objection it may now or hereafter have to venue or to convenience of forum, agrees that all claims in respect of such proceeding shall be heard and determined only in any such court, and agrees not to bring any proceeding arising out of or relating to this Agreement or any of the transactions contemplated hereby in any other court. For the purposes of such actions, proceedings, or counterclaims, service of process on a Party hereto shall be deemed effective if it is dispatched by United States first class mail to such Party's address provided in Section 10 (Notices). The Parties and the Guarantor acknowledge and agree that this Agreement was drafted, negotiated and executed in the State of Missouri.

## 22. <u>Waiver of Jury Trial.</u>

THE GUARANTOR AND EACH PARTY TO THIS AGREEMENT WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY ACTION SUIT OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT.

# 23. <u>Further Assurances.</u>

Each Party agrees to execute all such further documents and instruments and to do all such further things as any other Party may reasonably request in order to give effect to and to consummate the transactions contemplated hereby, and to provide access to any other Party or Regulatory Authority asserting jurisdiction over such other Party to the extent necessary for such other Party to comply with Applicable Law.

## 24. <u>Termination.</u>

The obligations of the Parties hereunder shall terminate automatically upon the effectiveness of termination of the PMA or, if earlier terminated, the Emerald Advance program. Upon termination of this Agreement, the Parties shall have such rights and obligations with respect to the Accounts and Receivables, including Participated Receivables and Retained Receivables, as may exist pursuant to the terms of this Agreement as of the termination date, or as may be otherwise agreed upon in writing by the Parties.

## 25. <u>No Relationship.</u>

Nothing contained in this Agreement shall establish any fiduciary, partnership, joint venture or similar relationship between any of the Parties.

# 26. <u>Counterpart Execution.</u>

This Agreement may be signed in any number of counterparts, each of which shall be deemed an original, and which taken together shall constitute one and the same instrument. Delivery of an executed counterpart of a signature page of this Agreement in Portable Document Format (PDF) or by facsimile transmission shall be as effective as delivery of a manually executed original counterpart of this Agreement.

# 27. <u>Severability.</u>

The illegality, invalidity or unenforceability of any provision of this Agreement under the law of any jurisdiction shall not affect its legality, validity or enforceability under the law of any other jurisdiction nor the legality, validity or enforceability of any other provision.

# 28. <u>Underwriting Standards.</u>

Pursuant to the PMA, BofI shall establish the Underwriting Standards. BofI may change the Underwriting Standards as permitted by the PMA, and in consultation with and upon prior written notice to Participant; <u>provided</u>, <u>however</u>, that, except as required by Applicable Law, no change in the Underwriting Standards that could reasonably be expected to have a material adverse effect on the Participant shall be made without the prior written consent of Participant.

# 29. <u>Clarification of Intent.</u>

BofI and Participant intend that each sale and purchase of a Participation pursuant to this Agreement shall constitute a true sale of BofI's right, title and interest in and to the related Participated Receivables to the extent of the Participant Percentage Interest in such Participated Receivables and not a pledge of security for a loan from Participant. BofI and Participant intend to account for the sale and purchase of each Participation in accordance with the guidelines set forth under FASB ASC 860-10-40-6A (as amended, from time to time). Should this treatment be challenged, the parties agree to modify and reform this Agreement, so that all Participations conform to the guidelines set forth under FASB ASC 860-10-40-6A and are treated as a "sale" for accounting purposes. If, notwithstanding such intent and any such reform, any sale and purchase of a Participation is deemed to be a pledge of security for a loan, BofI and Participant intend that the rights and obligations of the Parties with respect to such loan shall be established pursuant to the terms of this Agreement, and BofI hereby grants to Participant a perfected, first priority security interest in all of BofI's right, title and interest in and to the related Participated Receivables and the

related Distributions, in each case, to the extent of the Participant Percentage Interest in such Participated Receivables, to secure all of BofI's obligations under this Agreement, the Purchase Price Notices and the Settlement Reports, and in such case, this Agreement shall constitute a security agreement between BofI and Participant under Applicable Law. BofI authorizes Participant to file UCC-1 financing statements, any continuation statements with respect thereto and any amendments thereto, to declare the intent of the Parties and as necessary to perfect and protect the interests of Participant in such Participations and Distributions.

### 30. <u>Guarantee.</u>

(a) In consideration of BofI entering into this Agreement and the transactions contemplated hereunder, Guarantor hereby absolutely, unconditionally and irrevocably guarantees (the "<u>Guarantee</u>") to BofI and BofI Indemnified Parties the full and prompt performance and payment when due of all liabilities, obligations and undertakings of Participant (and its successors and permitted assigns) under this Agreement, including under Sections 2, 4, 5(c), 12 and 15 (the "<u>Guaranteed Obligations</u>"), and all reasonable and documented legal fees, costs and expenses incurred by BofI in enforcing this Guarantee.

(b) This Guarantee is an absolute, unconditional and continuing guarantee of the full and punctual payment of the Guaranteed Obligations. This Guarantee is in no way conditioned upon any requirement that Bofl first attempt to collect the Guaranteed Obligations from Participant or resort to any security or other means of collecting payment. Claims under this Guarantee may be made on one or more occasions. If any payment in respect of any Guaranteed Obligations is rescinded or must otherwise be returned for any reason whatsoever, Guarantor shall remain liable hereunder with respect to such Guaranteed Obligations will be paid strictly in accordance with the terms of the Agreement. This Guarantee is a guarantee of performance and payment and not merely of collection.

(c) If Participant defaults in the payment of any portion of the Guaranteed Obligations or otherwise is unable for any reason to pay any Guaranteed Obligations as and when due, or if BofI is unable to bring a claim for the Guaranteed Obligations against Participant for any reason (including by reason of any bankruptcy or similar proceeding, automatic stay or otherwise) Guarantor shall make the payment required hereunder or otherwise cause such payment to be made within five (5) Business Days after the receipt by Guarantor of written notice from BofI or Servicer of such payment default by Participant. All sums payable by Guarantor hereunder shall be made in immediately available funds.

(d) Notwithstanding anything to the contrary contained or implied herein, Guarantor reserves the right to assert defenses that Participant may have to payment or performance of the Guaranteed Obligations (other than defenses arising from the bankruptcy, insolvency or similar rights of Participant), or defenses related to Participant's capacity or authority to enter into the Agreement.

(e) For avoidance of doubt, the Guarantee set forth in this Section 30 covers all obligations and rights of Participant that are assigned or delegated to another subsidiary of the Guarantor pursuant to Section 13(c) hereof.

### (signature page follows)

IN WITNESS WHEREOF, the Parties have duly executed this Agreement as of the date first above written.

# BofI FEDERAL BANK

By:\_\_\_\_\_ Name: Title:

# EMERALD FINANCIAL SERVICES, LLC

By:\_\_\_\_\_

Name:

Title:

# HRB PARTICIPANT I, LLC

By:\_\_\_\_\_

Name:

Title:

H&R Block, Inc.

By:\_\_\_\_\_

Name:

Title:

List of Schedules and Similar Attachments Omitted from Receivables Participation Agreement

Exhibit ACalculation of Purchase Price of Participation InterestExhibit BSettlement ReportExhibit CCalculation of Distribution of Cash Flow from Receivables

Upon request, the registrant agrees to furnish supplementally to the Securities and Exchange Commission a copy of any omitted schedule or similar attachment to the Receivables Participation Agreement; provided, however, that the registrant may request confidential treatment of omitted items prior to any public disclosure.

### **GUARANTY AGREEMENT**

THIS GUARANTY AGREEMENT (this "<u>Guaranty</u>"), dated as of \_\_\_\_\_\_, 2014, is made by and between **H&R** BLOCK, INC., a Missouri corporation (the "<u>Guarantor</u>"), and BofI FEDERAL BANK, a federal savings bank ("<u>Bank</u>").

## **RECITALS:**

A. As of the date hereof, Bank and Emerald Financial Services, LLC, a Delaware limited liability company (" <u>EFS</u>"), have entered into a Program Management Agreement and related documents and agreements in connection with the transactions contemplated thereunder (as amended, modified, supplemented and restated from time to time, collectively, the "<u>Transaction</u> <u>Documents</u>"), pursuant to which EFS has agreed to facilitate and service the offering and administration of certain financial products and related accounts by Bank.

B. Guarantor has agreed to guarantee the obligations of EFS to Bank as set forth herein.

C. EFS is an indirect subsidiary of Guarantor, and Guarantor will derive a tangible and substantial benefit from the transactions contemplated by the Transaction Documents.

D. Capitalized terms used and not otherwise defined herein shall have the meanings set forth in the Transaction Documents.

## AGREEMENT

ACCORDINGLY, in consideration of the foregoing premises, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, it is agreed as follows:

Section 1. Obligation of the Guarantor.

(a) Guarantor hereby absolutely, unconditionally and irrevocably guarantees to Bank, its Affiliates, the officers and directors of Bank and its Affiliates, and Bank's successors and permitted assigns (collectively, the "<u>Bank Indemnified Parties</u>"), the full and prompt payment and performance of each and every covenant, agreement, undertaking and obligation of EFS to Bank under or in connection with the Transaction Documents or the transactions contemplated thereby (collectively, the "<u>Obligations</u>"). Guarantor expressly waives diligence on the part of Bank in the collection or enforcement of the Obligations, whether fixed or contingent, and waives presentment, protest, notice of protest, dishonor, notice of acceptance of this Guaranty, demands for performance, and approval of any modifications, renewals or extensions of the Obligations that may be granted to EFS. Bank shall be under no obligation to notify Guarantor of its acceptance of this Guaranty, nor to

use diligence in preserving the liability of any entity or person on said Obligations whether fixed or contingent, nor in bringing suit to enforce collection or enforcement of the Obligations secured by this Guaranty, nor of notice of any instruments now or hereafter executed in favor of Bank evidencing or securing said indebtedness.

(b) Guarantor shall, upon EFS failing to satisfy any Obligation (such failure, an "<u>EFS Breach</u>"), pay or perform, as applicable, such Obligation within thirty (30) days after Bank provides Guarantor with written notice of such EFS Breach.

Section 2. <u>Reservation of Rights</u>. Notwithstanding anything to the contrary contained or implied herein, Guarantor reserves the right to assert any defenses that EFS may have to payment or performance of an Obligation, in whole or in part, under the Transaction Documents or otherwise (other than defenses arising from the bankruptcy, insolvency or similar rights of EFS, or defenses related to the capacity or authority of EFS to enter into the Transaction Documents).

Section 3. <u>Continuing Guaranty: Reinstatement</u>. Guarantor shall not be relieved from its obligations hereunder until such time as all Obligations have been indefeasibly satisfied and paid in full and the time period for assertion of claims by Bank Indemnified Parties under the Transaction Documents shall have expired. This Guaranty shall continue to be effective or be reinstated, as the case may be, if at any time any payment on account of the Obligations shall be rescinded or must otherwise be restored upon the bankruptcy or reorganization of EFS or any other party or otherwise.

Section 4. <u>Guaranty of Payment</u>. It is the intent of the parties that this Guaranty is a guaranty of payment and not of collection.

Section 5. <u>Representations and Warranties of Guarantor</u>. Guarantor hereby represents and warrants for the benefit of Bank that:

(a) Guarantor has the requisite corporate power and authority to execute, deliver and perform this Guaranty;

(b) as of the date hereof, there is no undisclosed action, suit or proceeding pending or, to Guarantor's knowledge, threatened against Guarantor before any Governmental Authority in which there is a reasonable possibility of an adverse decision which could affect, in a materially adverse manner, the ability of Guarantor to perform any of its obligations under, or which in any manner questions the validity of, this Guaranty;

(c) the execution, delivery and performance of this Guaranty by Guarantor does not contravene or constitute a default under (i) any statute, regulation or rule of any Governmental Authority, (ii) any provision of the certificate of incorporation or by-laws of Guarantor or (iii) any contractual restriction binding on Guarantor; and

(d) this Guaranty constitutes the legal, valid and binding obligation of Guarantor enforceable in accordance with its terms, subject to the effect of any bankruptcy, insolvency, reorganization, moratorium or similar law affecting creditors' rights, and to general

principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).

Section 6. <u>Notices</u>. All notices, demands and other communications hereunder shall be in writing and shall be sent by electronic mail, certified mail return receipt requested, by hand, by facsimile with verbal confirmation of receipt, or by nationally recognized overnight courier addressed to the party to whom such notice or other communication is to be given as follows:

## If to Guarantor:

H&R Block, Inc. One H&R Block Way Kansas City, Missouri 64105 Attn: Tom Gerke, Chief Legal Officer Email: tom.gerke@hrblock.com Telephone: 816-854-6060 Facsimile: 816-854-8500

Attn: David Zarski Email: david.zarski@hrblock.com Telephone: 816-854-4502 Facsimile: 816-802-1013

Attn: Walter Pirnot Email: wpirnot@hrblock.com Telephone: 816-854-5757 Facsimile: 816-802-1065

# with copy to:

Stinson Leonard Street LLP Attn: Mike Lochmann 1201 Walnut Street, Suite 2900 Kansas City, Missouri 64106 Email: mike.lochmann@stinsonleonard.com Telephone: 816-691-3208 Facsimile: 816-412-1249



#### If to Bank:

BofI Federal Bank 4350 La Jolla Village Drive, Suite 140 San Diego, California 92122 Attn: Gregory Garrabrants Email: ggarrabrants@bofifederalbank.com Telephone: 858-350-6203 Facsimile: 858-764-6561

### with copies to:

Bofl Federal Bank 4350 La Jolla Village Drive, Suite 140 San Diego, California 92122 Attn: Eshel Bar-Adon Email: ebaradon@bofifederalbank.com Telephone: 858-764-2905 Facsimile: 858-764-6561

Sheppard Mullin Richter & Hampton, LLP 333 South Hope Street, 43<sup>rd</sup> Floor Los Angeles, California 90071-1422 Attn: Lawrence M. Braun Email: LBraun@sheppardmullin.com Telephone: 213-617-4184 Facsimile: 213-443-2814

or to such other person or address as either party shall have previously designated to the other by written notice given in the manner set forth in this Section 6. Any notice provided pursuant to this Section 6 shall be effective on the day of delivery if (x) sent by certified mail; (y) sent by a nationally recognized overnight courier service; and (z) delivered by hand or via electronic mail.

Section 7. <u>Severability</u>. Should any provision of this Guaranty be declared invalid for any reason or to have ceased to be binding on the parties hereto, such provision shall be severed, and all other provisions herein shall continue to be effective and binding.

Section 8. <u>Assignment</u>. Any assignment by a party of its rights or obligations hereunder shall require the prior consent of the other party hereto, which consent shall not be unreasonably withheld.

Section 9. <u>Governing Law</u>. This Guaranty and all rights and obligations hereunder, including matters of construction, validity and performance, shall be governed by and construed in accordance with the laws of the State of New York without regard to its conflict of laws provisions.

### Section 10. Consent to Jurisdiction.

(a) Each party hereto agrees that all actions, proceedings or counterclaims arising out of or relating to this Guaranty or any of the transactions contemplated hereby shall be brought in the United States District Court for the Eastern District of Missouri located in the City of St. Louis, or as to those lawsuits to which the Federal Courts of the United States lack subject matter jurisdiction, before a court located in the State of Missouri in the City of St. Louis, and each of the parties irrevocably submits to the exclusive jurisdiction of each such court in any proceeding, waives any objection it may now or hereafter have to venue or to convenience of forum, agrees that all claims in respect of such proceeding shall be heard and determined only in any such court, and agrees not to bring any proceeding arising out of or relating to this Guaranty or any of the transactions contemplated hereby in any other court. For the purposes of such actions, proceedings, or counterclaims, service of process on a party hereto shall be deemed effective if it is dispatched by United States first class mail to such party's address provided in Section 6.

(b) The parties hereto acknowledge and agree that this Guaranty was executed and delivered in the State of Missouri.

Section 11. <u>Waivers</u>. To the extent permitted by law, and notwithstanding anything to the contrary in this Guaranty or in the Transaction Documents, Guarantor hereby waives and agrees not to assert or take advantage of:

(a) Subject to Guarantor's right to notice of an EFS Breach pursuant to Section 1(b) of this Guaranty, any right to require Bank to proceed against EFS, any member of EFS, or any other indemnitor or guarantor of the Obligations guaranteed hereby, or any other person, or to proceed against or exhaust any security held by Bank at any time or to pursue any other remedy in Bank's power or under any other agreement before proceeding against Guarantor hereunder;

(b) Any defense based upon an election of remedies by Bank, even though such election destroys or otherwise impairs the subrogation rights of Guarantor or the right of Guarantor (after payment of the obligations guaranteed by Guarantor under this Guaranty) to proceed against EFS for reimbursement, or both;

(c) Any right or claim or right to cause a marshaling of the assets of Guarantor;

(d) Any duty on the part of Bank to disclose to Guarantor any facts Bank may now or hereafter know about EFS, regardless of whether Bank has reason to believe that any such facts materially increase the risk beyond that which Guarantor intended to assume or has reason to believe that such facts are unknown to Guarantor or has a reasonable opportunity to communicate such facts to Guarantor, it being understood and agreed that Guarantor is fully responsible for being and keeping informed of the financial condition of EFS and of any and all circumstances bearing on the risk that liability may be incurred by Guarantor;

(e) Any lack of notice of disposition or of manner of disposition of any collateral;

(f) Any invalidity, irregularity or unenforceability, in whole or in part, of any one or more of the Transaction Documents;

(g) Any deficiencies in any collateral or any deficiency in the ability of Bank to collect or to obtain performance from any persons or entities now or hereafter liable for the payment and performance of any obligation hereby guaranteed;

(h) Any assertion or claim that the automatic stay provided by 11 U.S.C. §362 (arising upon the voluntary or involuntary bankruptcy proceeding of EFS) or any other stay provided under any other debtor relief law (whether statutory, common law, case law or otherwise) of any jurisdiction whatsoever, now or hereafter in effect, which may be or become applicable, shall operate or be interpreted to stay, interdict, condition, reduce or inhibit the ability of Bank to enforce any of its rights, whether now or hereafter required, which Bank may have against Guarantor or any collateral;

(i) Any modifications of the Transaction Documents or any obligation of EFS by operation of law or by action of any court, whether pursuant to the Bankruptcy Reform Act of 1978, as amended, or any other debtor relief law (whether statutory, common law, case law or otherwise) of any jurisdiction whatsoever, now or hereafter in effect, or otherwise;

(j) Any principles or provisions of law, statutory or otherwise, that are or might be in conflict with the terms of this Guaranty. By doing so, Guarantor agrees that Guarantor's obligations shall not be affected by any circumstances, whether or not referred to in this Guaranty, which might otherwise constitute a legal or equitable discharge of a surety or guarantor; and

(k) Any right of discharge under any and all statutes or other laws relating to guarantors or sureties, and any other rights of sureties and guarantors thereunder.

Section 12. <u>Attorney Fees</u>. If any action or other proceeding is brought for the enforcement or interpretation of any of the rights or provisions of this Guaranty, or because of an alleged dispute, breach, default or misrepresentation in connection with any of the provisions of this Guaranty, the successful or prevailing party will be entitled to recover reasonable attorneys' fees and all other costs and expenses incurred in that action or proceeding, in addition to any other relief to which it may be entitled.

Section 13. <u>Waiver of Jury Trial</u>. EACH PARTY TO THIS GUARANTY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY ACTION SUIT OR PROCEEDING ARISING OUT OF OR RELATING TO THIS GUARANTY.

Section 14. <u>Section Headings</u>. Section headings used herein are solely for the convenience of the parties hereto and shall not affect the interpretation or construction of this Guaranty.

Section 15. <u>Binding Agreement</u>. This Guaranty is legally binding on the parties hereto, and their respective successors and permitted assigns.

Section 16. <u>Counterparts and Facsimile Signatures</u>. This Guaranty may be executed in any number of multiple counterparts, all of which shall constitute but one and the same original. Facsimile signatures to this Guaranty shall be effective.

Section 17. <u>Entire Agreement</u>. This Guaranty constitutes the entire agreement between the parties hereto with respect to the subject matter hereof, and it shall not be amended, altered or changed except by a written agreement signed by each of the parties hereto.

IN WITNESS WHEREOF, the parties have executed this Guaranty as of the day and year first above written.

H&R BLOCK, INC.

By:\_\_\_\_\_ Name: Title:

BofI FEDERAL BANK

By:		 
Name:		
Title:		



**News Release** For Immediate Release: April 10, 2014

### H&R Block Reaches Definitive Agreement to Divest its Bank

- H&R Block Bank to sell certain assets and liabilities to BofI Federal Bank ("BofI")
- Transaction subject to regulatory approvals
- Net, ongoing financial impact of transaction expected to dilute H&R Block earnings by approximately \$0.07 to \$0.09 per share annually beginning in fiscal 2015, based on current fully diluted shares outstanding

KANSAS CITY, Mo. - H&R Block, Inc. (NYSE: HRB), the world's largest consumer tax services provider, today announced that H&R Block Bank has entered into a definitive purchase and assumption agreement with BofI Federal Bank ("BofI") to sell certain assets and transfer certain liabilities to BofI. The agreement is subject to regulatory approvals and other customary closing conditions. In addition, the parties have agreed to terms of a program management agreement under which BofI will act as the bank for H&R Block-branded financial services products: Emerald Prepaid MasterCard<sup>®</sup>, Refund Transfers and Emerald Advance<sup>®</sup> lines of credit through H&R Block's retail and digital channels. The program management agreement will be entered into upon closing of the definitive purchase and assumption agreement.

"We're very pleased to reach a definitive agreement with BofI that enables us to exit our bank while continuing to offer best-in-class financial services products to our clients," said Bill Cobb, H&R Block's president and chief executive officer. "This is an important step in ceasing to be regulated as a savings and loan holding company, which we believe is in the best strategic interests of our company and our shareholders."

Both H&R Block Bank and BofI are in the process of applying for required regulatory approvals with their respective regulators. Following the fulfillment of all closing conditions, including receipt of required regulatory approvals, H&R Block Bank will complete the transaction, merge with and into its parent, Block Financial LLC, and surrender its bank charter. Subject to meeting the aforementioned conditions, including the receipt of required regulatory approvals, H&R Block Bank will complete the transaction.

H&R Block expects the net, ongoing annual financial impact of the program management agreement to be dilutive by approximately \$0.07 to \$0.09 per share beginning in fiscal 2015, based on current fully diluted shares outstanding. Results will vary based upon the volume of financial services products sold and the actual closing date.

Contingent upon the timing of regulatory approval, H&R Block also expects to incur one-time charges for transaction and related costs of approximately \$0.01 per share in fiscal year 2014 and approximately \$0.02 to \$0.03 per share in fiscal year 2015, based on current fully diluted shares outstanding and the expected closing schedule. Additionally, the company will sell certain available for sale securities held by H&R Block Bank, and expects to recognize a loss of approximately \$0.03 per share in fiscal year 2014 in

connection with the planned sale, based on current market conditions. The company also expects the bank transaction will result in approximately \$200 to \$250 million of immediate excess capital.

H&R Block announced in October 2012 that it was seeking strategic alternatives for H&R Block Bank that would result in H&R Block, Inc. and certain subsidiaries no longer being regulated by the Board of Governors of the Federal Reserve System as savings and loan holding companies. This decision was prompted by new rules that will impose higher capital requirements on savings and loan holding companies such as H&R Block, Inc. The Federal Reserve promulgated the rules in order to implement changes required by the Dodd-Frank Act.

Goldman, Sachs & Co. and First Annapolis Consulting, Inc. are acting as advisors to H&R Block for the transaction, and Stinson Leonard Street LLP and Covington & Burling LLP are acting as H&R Block's legal counsel.

Additional information regarding the agreements with BofI, the transaction process, related contingencies, terms and conditions and the timing of the transaction is included in a Form 8-K filed today with the Securities and Exchange Commission. There can be no assurances regarding when or if the contemplated transactions will close, the ability to obtain all required regulatory and other approvals, or final terms and conditions of the various agreements.

### **Conference Call**

Discussion of the planned exit of H&R Block Bank, including the agreements and our current thinking regarding parameters of currently anticipated capital flexibility, will occur during a conference call for analysts, institutional investors, and shareholders. The call is scheduled for 9:00 a.m. Eastern time on April 11, 2014. To access the call, please dial the number below approximately 10 minutes prior to the scheduled starting time:

U.S./Canada (888) 895-5260 or International (443) 842-7595 Conference ID: 11933507

The call will also be webcast in a listen-only format for the media and public. The link to the webcast can be accessed directly at http://investors.hrblock.com.

A replay of the call will be available beginning at 12:00 p.m. Eastern time on April 11, 2014, and continuing until May 12, 2014, by dialing (855) 859-2056 (U.S./Canada) or (404) 537-3406 (International). The conference ID is 11933507. The webcast will be available for replay April 11, 2014 at http://investors.hrblock.com.

### About H&R Block

H&R Block, Inc. (NYSE: HRB) is the world's largest consumer tax services provider. More than 625 million tax returns have been prepared worldwide by and through H&R Block since 1955. In fiscal 2013, H&R Block had annual revenues of \$2.9 billion with 24.8 million tax returns prepared worldwide. Tax return preparation services are provided in over 11,000 company-owned and franchise retail tax offices worldwide by professional tax preparers, and through H&R Block digital products. H&R Block Bank provides affordable banking products and services. For more information, visit the H&R Block Newsroom.

### Forward-Looking Statements

This press release may contain forward-looking statements within the meaning of the securities laws. Forward-looking statements can be identified by the fact that they do not relate strictly to historical or

current facts. They often include words or variation of words such as "expects," "anticipates," "intends," "plans," "believes," "seeks," "estimates," "projects," "forecasts," "targets," "would," "will," "should," "could" or "may" or other similar expressions. Forward-looking statements provide management's current expectations or predictions of future conditions, events or results. All statements that address operating performance, events or developments that we expect or anticipate will occur in the future are forward-looking statements. They may include estimates of revenues, income, earnings per share, capital expenditures, dividends, liquidity, capital structure or other financial items, descriptions of management's plans or objectives for future operations, products or services, or descriptions of assumptions underlying any of the above. All forward-looking statements speak only as of the date they are made and reflect the company's good faith beliefs, assumptions and expectations, but they are not guarantees of future performance or events. Furthermore, the company disclaims any obligation to publicly update or revise any forwardlooking statement to reflect changes in underlying assumptions, factors, or expectations, new information, data or methods, future events or other changes, except as required by law. By their nature, forward-looking statements are subject to risks and uncertainties that could cause actual results to differ materially from those suggested by the forward-looking statements. Factors that might cause such differences include, but are not limited to, a variety of economic, competitive and regulatory factors, many of which are beyond the company's control and which are described in our Annual Report on Form 10-K for the fiscal year ended April 30, 2013 in the section entitled "Risk Factors," as well as additional factors we may describe from time to time in other filings with the Securities and Exchange Commission. In addition, there can be no assurances regarding the timing and ability to obtain all required regulatory and other approvals, the closing of the contemplated transactions, or the final terms and conditions of the various agreements. You should understand that it is not possible to predict or identify all such factors and, consequently, you should not consider any such list to be a complete set of all potential risks or uncertainties.

### **For Further Information**

Investor Relations: Colby Brown, (816) 854-4559, colby.brown@hrblock.com Media Relations: Gene King, (816) 854-4672, gene.king@hrblock.com